

# Operational Intervention Levels for Reactor Emergencies

and Methodology for Their Derivation

DATE EFFECTIVE: MARCH 2017



### IAEA SAFETY STANDARDS AND RELATED PUBLICATIONS

#### IAEA SAFETY STANDARDS

Under the terms of Article III of its Statute, the IAEA is authorized to establish or adopt standards of safety for protection of health and minimization of danger to life and property, and to provide for the application of these standards.

The publications by means of which the IAEA establishes standards are issued in the **IAEA Safety Standards Series**. This series covers nuclear safety, radiation safety, transport safety and waste safety. The publication categories in the series are **Safety Fundamentals**, **Safety Requirements** and **Safety Guides**.

Information on the IAEA's safety standards programme is available on the IAEA Internet site

#### http://www-ns.iaea.org/standards/

The site provides the texts in English of published and draft safety standards. The texts of safety standards issued in Arabic, Chinese, French, Russian and Spanish, the IAEA Safety Glossary and a status report for safety standards under development are also available. For further information, please contact the IAEA at: Vienna International Centre, PO Box 100, 1400 Vienna, Austria.

All users of IAEA safety standards are invited to inform the IAEA of experience in their use (e.g. as a basis for national regulations, for safety reviews and for training courses) for the purpose of ensuring that they continue to meet users' needs. Information may be provided via the IAEA Internet site or by post, as above, or by email to Official.Mail@iaea.org.

#### RELATED PUBLICATIONS

The IAEA provides for the application of the standards and, under the terms of Articles III and VIII.C of its Statute, makes available and fosters the exchange of information relating to peaceful nuclear activities and serves as an intermediary among its Member States for this purpose.

Reports on safety in nuclear activities are issued as **Safety Reports**, which provide practical examples and detailed methods that can be used in support of the safety standards.

Other safety related IAEA publications are issued as **Emergency Preparedness and Response** publications, **Radiological Assessment Reports**, the International Nuclear Safety Group's **INSAG Reports**, **Technical Reports** and **TECDOCs**. The IAEA also issues reports on radiological accidents, training manuals and practical manuals, and other special safety related publications.

Security related publications are issued in the IAEA Nuclear Security Series.

The **IAEA Nuclear Energy Series** comprises informational publications to encourage and assist research on, and the development and practical application of, nuclear energy for peaceful purposes. It includes reports and guides on the status of and advances in technology, and on experience, good practices and practical examples in the areas of nuclear power, the nuclear fuel cycle, radioactive waste management and decommissioning.

# OPERATIONAL INTERVENTION LEVELS FOR REACTOR EMERGENCIES

The following States are Members of the International Atomic Energy Agency:

AFGHANISTAN ALBANIA ALGERIA ANGOLA ANTIGUA AND BARBUDA ARGENTINA ARMENIA AUSTRALIA AUSTRIA AZERBAIJAN BAHAMAS BAHRAIN BANGLADESH BARBADOS BELARUS BELGIUM BELIZE BENIN BOLIVIA, PLURINATIONAL STATE OF BOSNIA AND HERZEGOVINA BOTSWANA BRAZIL BRUNEI DARUSSALAM BULGARIA BURKINA FASO BURUNDI CAMBODIA CAMEROON CANADA CENTRAL AFRICAN REPUBLIC CHAD CHILE CHINA COLOMBIA CONGO COSTA RICA CÔTE D'IVOIRE CROATIA CUBA CYPRUS CZECH REPUBLIC DEMOCRATIC REPUBLIC OF THE CONGO DENMARK DJIBOUTI DOMINICA DOMINICAN REPUBLIC ECUADOR EGYPT EL SALVADOR ERITREA **ESTONIA** ETHIOPIA FIJI FINLAND FRANCE GABON

GEORGIA GERMANY GHANA GREECE **GUATEMALA GUYANA** HAITI HOLY SEE HONDURAS HUNGARY **ICELAND** INDIA INDONESIA IRAN, ISLAMIC REPUBLIC OF IRAQ IRELAND ISRAEL ITALY JAMAICA JAPAN JORDAN **KAZAKHSTAN KENYA** KOREA, REPUBLIC OF KUWAIT **KYRGYZSTAN** LAO PEOPLE'S DEMOCRATIC REPUBLIC LATVIA LEBANON LESOTHO LIBERIA LIBYA LIECHTENSTEIN LITHUANIA LUXEMBOURG MADAGASCAR MALAWI MALAYSIA MALI MALTA MARSHALL ISLANDS MAURITANIA MAURITIUS MEXICO MONACO MONGOLIA MONTENEGRO MOROCCO MOZAMBIQUE MYANMAR NAMIBIA NEPAL **NETHERLANDS** NEW ZEALAND NICARAGUA NIGER NIGERIA NORWAY

OMAN PAKISTAN PALAU PANAMA PAPUA NEW GUINEA PARAGUAY PERU PHILIPPINES POLAND PORTUGAL QATAR REPUBLIC OF MOLDOVA ROMANIA RUSSIAN FEDERATION RWANDA SAN MARINO SAUDI ARABIA SENEGAL SERBIA SEYCHELLES SIERRA LEONE SINGAPORE SLOVAKIA **SLOVENIA** SOUTH AFRICA SPAIN SRI LANKA **SUDAN SWAZILAND SWEDEN** SWITZERLAND SYRIAN ARAB REPUBLIC TAJIKISTAN THAILAND THE FORMER YUGOSLAV REPUBLIC OF MACEDONIA TOGO TRINIDAD AND TOBAGO TUNISIA TURKEY TURKMENISTAN UGANDA UKRAINE UNITED ARAB EMIRATES UNITED KINGDOM OF GREAT BRITAIN AND NORTHERN IRELAND UNITED REPUBLIC OF TANZANIA UNITED STATES OF AMERICA URUGUAY **UZBEKISTAN** VANUATU VENEZUELA, BOLIVARIAN **REPUBLIC OF** VIET NAM YEMEN ZAMBIA ZIMBABWE

The Agency's Statute was approved on 23 October 1956 by the Conference on the Statute of the IAEA held at United Nations Headquarters, New York; it entered into force on 29 July 1957. The Headquarters of the Agency are situated in Vienna. Its principal objective is "to accelerate and enlarge the contribution of atomic energy to peace, health and prosperity throughout the world".

EPR-NPP-OILs (2017)

# OPERATIONAL INTERVENTION LEVELS FOR REACTOR EMERGENCIES

AND METHODOLOGY FOR THEIR DERIVATION

INTERNATIONAL ATOMIC ENERGY AGENCY VIENNA, 2017

#### **COPYRIGHT NOTICE**

All IAEA scientific and technical publications are protected by the terms of the Universal Copyright Convention as adopted in 1952 (Berne) and as revised in 1972 (Paris). The copyright has since been extended by the World Intellectual Property Organization (Geneva) to include electronic and virtual intellectual property. Permission to use whole or parts of texts contained in IAEA publications in printed or electronic form must be obtained and is usually subject to royalty agreements. Proposals for non-commercial reproductions and translations are welcomed and considered on a case-by-case basis. Enquiries should be addressed to the IAEA Publishing Section at:

Marketing and Sales Unit, Publishing Section International Atomic Energy Agency Vienna International Centre PO Box 100 1400 Vienna, Austria fax: +43 1 2600 29302 tel.: +43 1 2600 22417 email: sales.publications@iaea.org http://www.iaea.org/books

For further information on this publication, please contact:

Incident and Emergency Centre International Atomic Energy Agency Vienna International Centre PO Box 100 1400 Vienna, Austria Email: Official.Mail@iaea.org

OPERATIONAL INTERVENTION LEVELS FOR REACTOR EMERGENCIES AND METHODOLOGY FOR THEIR DERIVATION IAEA, VIENNA, 2017 EPR-NPP-OILs (2017) ISSN 2518–685X © IAEA, 2017 Printed by the IAEA in Austria March 2017

#### FOREWORD

The IAEA safety standards require the establishment of predetermined default operational criteria as part of the protection strategy for nuclear and radiological emergencies. The IAEA Safety Standards Series No. GSR Part 7, Preparedness and Response for a Nuclear or Radiological Emergency, in para. 4.28(4) stipulates that: "Once the protection strategy has been justified and optimized and a set of national generic criteria has been developed, pre-established operational criteria (conditions on the site, emergency action levels (EALs) and operational intervention levels (OILs)) for initiating the different parts of an emergency plan and for taking protective actions and other response actions shall be derived from the generic criteria". This publication provides selected default OIL values for taking protective actions and other response actions to protect the public in an emergency involving a severe release of radioactive material from a light water reactor or its spent fuel, the methodological approach for their derivation, as well as practical tools and recommendations for their use, which will assist IAEA Member States in meeting the requirement of deriving pre-established OILs.

The default OIL values provided in this publication will contribute to the harmonization of national criteria for implementing protective actions and other response actions, a need that was emphasized by the 55th IAEA General Conference in resolution GC(55)/RES/9. It was again highlighted by the 56th IAEA General Conference in resolution GC(56)/RES/9, which stressed "the importance of the establishment and implementation of national emergency preparedness and response measures, based on the Agency's Safety Standards and relevant action plans, for improving preparedness and response, including communication in an emergency, and contributing to the harmonization of national criteria for protective and other actions".

The IAEA officers responsible for this publication were T. Mc Kenna and P. Vilar Welter of the Incident and Emergency Centre, Department of Nuclear Safety and Security.

#### EDITORIAL NOTE

This publication (including the figures, tables and references) has undergone only the minimum copy editing considered necessary for the reader's assistance.

The views expressed remain the responsibility of the contributors and do not necessarily reflect those of the IAEA or the governments of its Member States.

Neither the IAEA nor its Member States assume any responsibility for consequences which may arise from the use of this publication. This publication does not address questions of responsibility, legal or otherwise, for acts or omissions on the part of any person.

The use of particular designations of countries or territories does not imply any judgement by the publisher, the IAEA, as to the legal status of such countries or territories, of their authorities and institutions or of the delimitation of their boundaries.

The mention of names of specific companies or products (whether or not indicated as registered) does not imply any intention to infringe proprietary rights, nor should it be construed as an endorsement or recommendation on the part of the IAEA.

The IAEA has no responsibility for the persistence or accuracy of URLs for external or third party Internet web sites referred to in this publication and does not guarantee that any content on such web sites is, or will remain, accurate or appropriate.

1.	INT	RODUCTION	1
	1.1.	BACKGROUND	1
	1.2.	SCOPE	2
	1.3.	OBJECTIVE	2
	1.4.	STRUCTURE	3
2.	OIL	CHARTS AND DEFAULT OIL VALUES FOR USE IN THE RESPONSE	5
	2.1.	ROLE OF THE OILS WITHIN THE PROTECTION STRATEGY	5
	2.2.	EVALUATING MONITORING RESULTS	8
	2.3.	COMMUNICATING WITH DECISION MAKERS	10
	2.4.	COMMUNICATING WITH PUBLIC INFORMATION OFFICERS	12
	2.5.	HOW TO USE THE OIL CHARTS	13
	2.6.	OIL CHARTS FOR USE IN THE RESPONSE	14
		2.6.1. Ground monitoring (OIL1 <sub>y</sub> , OIL2 <sub>y</sub> and OIL3 <sub>y</sub> )	14
		2.6.2. Skin monitoring (OIL4 <sub><math>\gamma</math></sub> and OIL4 <sub><math>\beta</math></sub> )	16
		2.6.3. Food, milk and drinking water samples (OIL7)	18
		2.6.4. Thyroid monitoring (OIL8 $_{\gamma}$ )	19
	2.7.	INTEGRATION OF THE CHARTS INTO NATIONAL ARRANGEMENTS	20
3.	ME	THODOLOGY FOR DERIVING THE DEFAULT OIL VALUES	21
	3.1.	OVERVIEW	21
	3.2.	GENERIC CRITERIA	24
	3.3.	RADIONUCLIDE MIX	26
	3.4.	REPRESENTATIVE PERSON, EXPOSURE SCENARIOS AND PATHWAYS	37
	3.5.	BEHAVIOUR OF THE RADIONUCLIDES	43
		3.5.1. Ground weathering and decay	43
		3.5.2. Resuspension and decay	45
		3.5.3. Inadvertent ingestion from the ground and decay	46
		3.5.4. Transfer from the ground to vegetables and pasture	50
		3.5.5. Removal from vegetation by weathering and decay	50
		3.5.6. Transfer from cow's feed to cow's milk	
		3.5.7. Reduction due to processing and delay before consumption	52
		3.5.8. I ransfer from the skin to the GI tract by inadvertent ingestion	
		3.5.9. Removal from the skin by weathering and decay	33 55
		3.5.10. Reduction of the measured dose rate due to the decay of I-131 in the thyroid	55 56
	36	DOSE CONVERSION FACTORS	50 57
	2.0.	3.6.1 Dose conversion factors for the 'ground' scenario	57
		3.6.2 Dose conversion factors for the 'food pre-analysis' scenario	<i>51</i> 66
		3.6.3. Dose conversion factors for the 'skin' scenario	
		3.6.4. Dose conversion factors for the 'food post-analysis' scenario	73
		3.6.5. Dose conversion factors for the 'thyroid' scenario	74

# CONTENTS

3.7. INSTRUMENT RESPONSE	76
3.7.1. Ambient dose equivalent rate	
3.7.2. Beta monitoring instrument response	79
3.8. OIL(t,mix) FUNCTIONS AND DEFAULT OIL VALUES	
3.8.1. OIL1 <sub>γ</sub>	
3.8.2. OIL2 <sub>γ</sub>	
3.8.3. $OIL3_{\gamma}$	
3.8.4. OIL4 $_{\gamma}$	
$3.8.5.$ OIL $4_{\beta}$	
3.8.7 OIL 8.	102
3.9 WEIGHTING FACTORS	104
3.10.CONSERVATISM	
4. SPREADSHEETS	
4.1. MACROS	
4.2. 'OIL CALCULATION' SPREADSHEET	
4.2.1. Welcome tab	
4.2.2. Control tab	
4.2.3. Calculation tab	
4.2.4. OIL8 tab	
4.3. 'SPECTRA' SPREADSHEET	
4.3.1. Welcome tab	
4.3.2. Calculation tab	
4.3.3. KAD tab	
4.4. REVISING THE SPREADSHEETS	
5. REVISING THE DEFAULT OIL VALUES	
5.1. GENERAL CONSIDERATIONS	117
5.2. REVISING OIL4 <sup><math>\beta</math></sup> FOR A SPECIFIC BETA MONITORING INSTRUMENT	
APPENDIX: RELATIONSHIP WITH OTHER IAEA PUBLICATIONS	
REFERENCES	
NOTATION	
ABBREVIATIONS	147
CONTRIBUTORS TO DRAFTING AND REVIEW	149

#### 1. INTRODUCTION

This publication provides selected default OIL values, together with a detailed description of the methodology for their derivation, as well as practical tools and recommendations for their use. The tools and default OIL values provided here may be directly integrated into national emergency arrangements or reviewed and modified as necessary to meet the specific emergency preparedness and response arrangements of the country in which they will be applied.

#### 1.1. BACKGROUND

The use of OILs as part of the protection strategy for nuclear and radiological emergencies is required by IAEA Safety Standards Series No. GSR Part 7 [1], and addressed by IAEA Safety Standards Series No. GSG-2 [2] (Criteria for Use in Preparedness and Response for a Nuclear or Radiological Emergency) and No. GS-G-2.1 [3] (Arrangements for Preparedness for a Nuclear or Radiological Emergency), as well as by EPR Series publications (e.g. EPR-NPP Public Protective Actions 2013 [4]) and IAEA TECDOCs (e.g. IAEA-TECDOC-955 [5]).

OILs are operational criteria that allow the prompt implementation of protective actions and other response actions<sup>1</sup> on the basis of monitoring results that are readily available during a nuclear or radiological emergency. 'Operational' refers to the need for the OILs to be practical and reflect the realities of the response to an emergency, such as the need for the measured quantities to be representative, easily measurable and readily available during a nuclear or radiological emergency.

A default OIL value is a specific value of such a measured quantity that indicates the need to implement predetermined response actions (e.g. evacuation, relocation, food restrictions). The response actions implemented based on the default OIL values are intended to minimize radiation induced health effects that would reduce quality of life.<sup>2</sup> The default OIL values provided in this publication follow a reasonably conservative<sup>3</sup> approach; they are established below those levels at which radiation induced health effects will be observed, even in a very large exposed group of people composed of the most sensitive members of the public.

However, it is also important to consider non-radiological consequences of the response actions to ensure they do more good than harm. This is achieved by the justification and optimization of the overall protection strategy, as established in Requirement 5 of the IAEA Safety Standards Series No. GSR Part 7 [1]. The default OIL values and methodology provided in this publication are generically justified and optimized on radiation protection grounds, but further optimization and justification may be necessary in consideration of the specific protection strategy in which the OILs will be applied. For example, using much lower default OIL values than those provided here could result in more harm than good when considering: (a) the health hazard associated with the response action itself; and (b) the diversion of limited resources from the highest priority actions.

<sup>&</sup>lt;sup>1</sup> For ease of reading, 'protective actions and other response actions' will hereafter be referred to as 'response actions'.

<sup>&</sup>lt;sup>2</sup> By avoiding severe deterministic effects or a discernible increase in the incidence of stochastic effects (e.g. cancers).

<sup>&</sup>lt;sup>3</sup> Conservative meaning that it will result in a projected dose higher than the dose actually expected to be received under real conditions.

Default OIL values need to be developed during the preparedness stage, (a) to allow taking decisions on response actions quickly in the urgent and early phases of an emergency for the actions to be effective,<sup>4</sup> and (b) to account for the limited availability and reliability of information at these phases of the emergency. During past nuclear and radiological emergencies, failure to pre-establish default OIL values has resulted in unnecessarily postponing warranted response actions and in taking damaging actions that were not warranted based on the radiological health hazard [6, 7, 8, 9, 10].<sup>5</sup>

Once actions have been completed based on the default operational criteria, and once the greatest risk to the public has therefore been alleviated, there will be time for more deliberate assessments. As the emergency progresses, further information may become available. Arrangements need to be established in advance to consider prevailing conditions as they evolve and, if justified, to revise the OILs and explain those changes to the public in a plain and understandable form.<sup>6</sup>

#### 1.2. SCOPE

This publication addresses OILs for a severe release of radioactive material from an LWR or its spent fuel<sup>7</sup>, for the following monitoring results: (a) dose rate measurements above the ground; (b) dose rate measurements and beta count rates from the skin; (c) concentrations of marker radionuclides <sup>8</sup> in food, milk and drinking water samples; and (d) dose rate measurements from the thyroid.

The general methodology presented in this publication is generically applicable for deriving default OIL values for other reactor types or for radiological emergencies, but needs to be adapted. Additional publications addressing other reactor types and radiological emergencies may be issued by the IAEA.

#### 1.3. OBJECTIVE

This publication seeks to assist IAEA Member States in meeting the requirement of para. 4.28(4) of the IAEA Safety Standards Series No. GSR Part 7 [1], and contribute to the harmonization of national criteria for response actions (a need emphasized by the 55th and 56th IAEA General Conferences in resolutions GC(55)/RES/9 and GC(56)/RES/9), by providing Member States with: (a) default OIL values that can be used immediately in the response to an emergency due to severe conditions at an LWR or its spent fuel; (b) a detailed explanation of the methodology for deriving the default OIL values; (c) practical tools and recommendations on how to use the OILs; and (d) considerations for revising the default OIL values.

<sup>&</sup>lt;sup>4</sup> Effective in keeping the dose below the pre-established generic criteria discussed in Section 3.2.

<sup>&</sup>lt;sup>5</sup> Section 2.4 provides examples of damaging actions taken during past nuclear and radiological emergencies that were not warranted based on the radiological health hazard.

<sup>&</sup>lt;sup>6</sup> Experience from past emergencies has shown that changing criteria for the implementation of response actions during an emergency may lead to confusion of decision makers and scepticism among the public [6].

<sup>&</sup>lt;sup>7</sup> Only spent fuel that is sufficiently heated (either by its own decay heat or another heat source) to reach zirconium ignition temperatures is expected to result in a significant release of radioactive material warranting response actions. This is typically spent fuel in the spent fuel pool of an NPP.

<sup>&</sup>lt;sup>8</sup> A marker radionuclide is easy to identify and is representative of all other radionuclides present, avoiding the need for costly and time intensive comprehensive isotopic analyses.

#### 1.4. STRUCTURE

Following the introductory section (Section 1), this publication is structured into four main parts:

- Section 2 provides stand-alone 'OIL charts' (a practical tool for using the default OIL values during the response to an emergency), together with important practical information for using the OILs and making recommendations on warranted response actions.
- Section 3 provides a detailed explanation of the methodology for deriving the default OIL values given in Section 2, to support Member States in following the underlying calculations and assumptions.
- Section 4 provides a description of the two spreadsheets included on the attached CD, which were used to calculate the default OIL values (following the method given in Section 3); they are included here in order to facilitate the revision of the OILs to consider different underlying assumptions or another methodological approach than the one used by the IAEA.
- Section 5 provides (a) general considerations concerning the revision of the default OIL values to consider different underlying assumptions or another methodological approach and (b) instructions for adapting the default OIL value provided for the beta count rate from activity on the skin to a specific instrument.<sup>9</sup>

<sup>&</sup>lt;sup>9</sup> Default OIL values are provided for the beta count rate from activity on the skin as requested by some Member States.

#### 2. OIL CHARTS AND DEFAULT OIL VALUES FOR USE IN THE RESPONSE

This section provides stand-alone 'OIL charts' (a practical tool for using the OILs during the response to an emergency), together with important practical information that needs to be considered when using the OILs. The charts are structured as shown in Fig. 1. Before using the charts, it is important to understand the role of the OILs within the protection strategy, the evaluation of monitoring results and the communication with decision makers and public information officers (as explained in the following sections). Section 2.5 provides a step by step instruction on how to use the OIL charts, and Section 2.6 provides the actual charts.

		OIL CHART FOR SKIN MONITORING [LWRs] - GAMMA	
Checklist to ensure the correct use of the OILs		ATTENTION: Only use this OIL if the answer to all the following questions is 'yes'.	
		Has there been a nifease of radioactive material from an LWR or its spent faul?   Yes No	
		Are you assessing the ambient dose equivalent rate from the bare skin of the hand.  Yes No and face?	
Purpose within the protection strategy	PURPOSE WITHIN THE PROTECTION STRATEOY	To fee each to interfly orientation with receipt inductions related on the relative to sense it response to dense to during and contractionality. Only the public preventation of incolutions Telescotte the panelshy have addressing and addressing and the public as a sense of the public as a measurement. On the regime and with their members of the public as a sense that the public public public public as a sense of the public as a sense that the relative to a sense of the public as a sense of the public as a discussion of the sense of the relative to a sense of the theory of the public as a sense of the relative to a sense of the theory of the public as a discussion of the sense of the off the relative to a sense. Add Theory of the discussion of the sense of the off the relative to a sense of the theory of the other relative to the relative to a sense of the theory of the other public theory of the relative to a sense of the other other other other other other other relative to a sense of the public the relative to a sense of the theory of the other other other relative to a sense other other other other other other other other other other relative to a sense other other other other other other other other other other relative to a sense other other other other other other other other other relative to a sense other other other other other other other other other relative to a sense other o	
Description of the monitoring type	MONITORING TYPE	Another does equivalent rate at 10 cm from the bare skin of the frame set	
Default OIL value(s)	DEFAULT OIL VALUE	OE4, = 1 µd/vh slove badgmand SHM MONITORING	
	RESPONSE ACTIONS FOR ALL THOSE THAT MAY BE MONITORED	Whith the first hears after togetime of the accesses before resolution is industremental on the present constraints in the present of the risk in the method the present of the nations. Thus, a proven one bip present by taking and non-big present of and the present of the nations. Thus, a proven one bip present by taking and methods and non-big present of the national straints and the present of the national straints. If can be down safely is g do instant of methods and present on the national straints and the national straints and the instant of the national temporal or national straints. If can be down safely is g do the means how treating under temporation constraints of indexia has the first and the subty of they are national presenting straints and straints (in the nation is a software the means the temporation constraints of indexia has the first and the subty of they are not presenting straints and straints (in the nation is a software they in the straints and the temporation constraints of indexia has the subty of they are not presenting straints and straints (in the constraints of the straints) and the straints and they match to a straint straints and the straints and a straint straints and they match to a straint straints and the straints and a straints and they match to a straint straints and they match to a straint straints and they match to a straint straints and they are a straints and a straint straints and a straints	
Recommended response actions		Notice 54 and 54 a	
	EXCEEDED	Check OLLSy	
Putting the health hazard in perspective	HEALTH	Chart 2 in Section 7 of IAEA EPRONP Public Protective Actions (2015) (4) can be used to place the health hazard in perspective when communicating with decision makers and public information officiers.	

FIG. 1. General structure of the OIL charts.

#### 2.1. ROLE OF THE OILS WITHIN THE PROTECTION STRATEGY

The role of the OILs provided in this publication (Table 1) is outlined in Fig. 2. As learned from past emergencies [6, 7, 8, 9, 10], for those response actions that in order to be effective<sup>10</sup> need to be taken before or shortly after a release of radioactive material, monitoring results will not be available in time to support decision making. Therefore, initial response actions need to be implemented within pre-established emergency planning zones and planning distances based on an emergency classification system (i.e. based on the detection of conditions that indicate the need for response actions, as described in Refs [1, 2, 4, 5, 11]).<sup>11</sup>

<sup>&</sup>lt;sup>10</sup> Effective in keeping the dose below the pre-established generic criteria discussed in Section 3.2.

<sup>&</sup>lt;sup>11</sup> For example, for a severe release of radioactive material from an NPP, the patterns of deposition are very complex and extensive and will change if there is an ongoing release, making it very difficult to promptly identify the areas warranting a restriction of food, milk and drinking water based on monitoring alone. Therefore, upon declaration of a General Emergency, provisions need to be in place within a pre-established emergency planning distance, i.e. the Ingestion and Commodities Planning Distance (ICPD) outlined in Refs [1, 4], to automatically restrict nonessential local produce (e.g. leafy vegetables) and wild-grown products (e.g. mushrooms) that may be directly contaminated by the release of radioactive material and consumed within weeks, milk from grazing animals, animal feed and directly collected rainwater (other sources of drinking water, e.g. wells, reservoirs or rivers, will only need to be restricted if the samples exceed OIL7). Failure to do so may result in significant doses to the public



FIG. 2. Role of the OILs provided in this publication within the protection strategy.

Once monitoring results become available, OILs will be essential in first expanding and later adjusting the initial response actions. Owing to the lack of reliable information and the urgency of the response actions early in an emergency, default OIL values need to be pre-established to allow taking prompt and confident actions.<sup>12</sup>

Once actions have been completed based on these default operational criteria, and once the greatest risk to the public has therefore been alleviated, there is time for more deliberate assessments. As prevailing conditions evolve, OILs may be revised according to pre-established arrangements, as appropriate, ensuring that the response actions are optimized and justified and that the reasons for the revisions are explained to the public in a plain and understandable form.<sup>13</sup>

and in a discernible increase in the incidence of thyroid cancers as observed during past emergencies. Restrictions should only be implemented if alternative food, milk and drinking water are available.

<sup>&</sup>lt;sup>12</sup> Following the previous example, restrictions on local produce, wild-grown products, milk from grazing animals, directly collected rainwater and animal feed would be extended based on ground monitoring results in areas exceeding  $OIL1_{\gamma}$ ,  $OIL2_{\gamma}$  or  $OIL3_{\gamma}$ . OIL7 would be used to confirm the adequacy of these restrictions based on food, milk and drinking water samples.

<sup>&</sup>lt;sup>13</sup> Following the previous example, the food, milk and drinking water affected by the release and/or the consumption behaviour may be so well characterized late in the emergency that it may be justified to revise the default OIL7 values to reflect actual conditions. This revision will need to be based on a defensible basis, agreed with all relevant stakeholders and clearly communicated to the public.

OIL	Default OIL value	Monitoring type		Charts in section	
$OIL1_{\gamma}$	1000 µSv/h				
	<b>100 μSv/h</b> (for the first 10 days after reactor shutdown <sup>a</sup> )		GROUND MONITORING		
OIL2 <sub>γ</sub>	<b>25 μSv/h</b> (later than 10 days after reactor shutdown <sup>a</sup> or for spent fuel)	1m		Ambient dose equivalent rate at 1 m above ground level	2.6.1
$OIL3_{\gamma}^{b}$	1 μSv/h				
OIL4 <sub>γ</sub> <sup>c</sup>	1 μSv/h	10 cm	SKIN MONITORING Ambient dose equivalent rate at 10 cm from the bare skin of the hand and face	2 ( 2	
OIL4 <sub>β</sub> <sup>c</sup>	<b>1000 cps</b> <sup>d</sup>	2 cm B	SKIN MONITORING Beta count rate at 2 cm from the bare skin of the hand and face (The use of OIL4 <sub><math>\gamma</math></sub> is preferable over OIL4 <sub><math>\beta</math></sub> )	- 2.6.2	
OIL7	1000 Bq/kg of I-131 and 200 Bq/kg of Cs-137	<sup>90</sup> Sr. (131], 114Cs (137CS)	MONITORING OF FOOD, MILK <sup>e</sup> AND DRINKING WATER SAMPLES Activity concentration of I-131 <sup>f</sup> and Cs-137 <sup>f</sup> in food, milk and drinking water samples	2.6.3	
OIL8 <sub>7</sub>	0.5 μSv/h	CONTACT WITH SKIN	THYROID MONITORING Ambient dose equivalent rate in front of the thyroid in contact with the skin	2.6.4	

#### TABLE 1. DEFAULT OIL VALUES PROVIDED IN THIS PUBLICATION

<sup>a</sup> Time after the nuclear reaction in the core was stopped.

<sup>b</sup> The advantage of  $OIL3_{\gamma}$  when compared with OIL7 is that, based on  $OIL3_{\gamma}$ , restrictions can be implemented early on in the emergency (i.e. when most needed) with readily available and easy to obtain ambient dose equivalent rates from deposition on the ground.

<sup>c</sup> The ambient dose equivalent rate  $OIL4_{\gamma}$  is sufficient and preferable to assess the levels of radioactive material on the skin for a release of radioactive material from a nuclear power plant (NPP) or its spent fuel, because it is less dependent on the measurement technique and instrument characteristics. However, beta count rate  $OIL4_{\beta}$  is also provided, since it may be used by some response organizations for skin monitoring.

<sup>d</sup> The default value is provided in cps and not in Bq/cm<sup>2</sup> because instruments natively provide cps. When providing Bq/cm<sup>2</sup>, an implicit assumption is made on the radionuclides being monitored. However, the radionuclide mix released from an NPP will be complex and changing, making the use of Bq/cm<sup>2</sup> impractical.

- <sup>e</sup> Milk is mentioned separately because of its key role in radiation induced thyroid cancers following the accident at the Chernobyl NPP.
- <sup>f</sup> I-131 and Cs-137 serve as marker radionuclides. A marker radionuclide is easier to identify and is representative of all the other radionuclides present, avoiding the need for costly and time intensive comprehensive isotopic analyses. The contribution of the other radionuclides expected to be present after a release of radioactive material due to severe fuel damage was considered.

The default OIL values provided in this publication were developed for the concept of operations described in Ref. [4]<sup>14</sup> and were generically justified and optimized on radiation protection grounds. Further justification and optimization within the overall protection strategy may be necessary during the preparedness stage to account for local and national conditions. The justification of the protection strategy requires risk informed decision making (as described in Refs [12, 13]), i.e. the participation of all those involved in the decision making process during the preparedness stage and late response phases. References [1, 2, 3] provide further details on the protection strategy for nuclear and radiological emergencies.

It needs to be highlighted that OILs indicate the need for promptly implementing response actions to protect the public during a nuclear or radiological emergency, and are not intended for the detection, classification or declaration of an emergency. Other operational criteria are intended for this purpose; namely observables/indicators and emergency action levels (EALs). In addition, it is inappropriate to use OILs for situations that are not emergency exposure situations (for example, during the regular transport of radioactive material or during medical exposure) since other criteria and assumptions apply.

#### 2.2. EVALUATING MONITORING RESULTS

Early in an emergency, monitoring results will most likely be limited, confusing and possibly inconsistent (as observed during past emergencies [6, 7, 8, 9, 10]) due to:

- Complex and varying deposition patterns resulting from natural processes (e.g. variations in deposition due to meteorological or ground conditions);
- Considerable uncertainties in the monitoring results due to field and emergency conditions (even professional teams may provide significantly different results for the same area); and
- Complex and varying spatial radionuclide mix distributions resulting from changing fuel damage states and plant conditions (that result in different radionuclide mixes being released) combined with changing meteorological conditions (e.g. wind or precipitation).

This will be particularly problematic early on in the emergency, when decisions need to be made quickly in order to be effective. Arrangements need to be in place to initially enable prompt decision making, based on early monitoring results, taking into account the characteristics of the measurements, the natural and social environment of the area and the representativeness of the measurement, as described in Ref. [4]. Examples are: (a) the number of readings exceeding the OIL; (b) the magnitude of the reading and whether it slightly exceeds the OIL or is greatly in excess; (c) the reliability of the measurement (Have confirmatory measurements been performed? Does the instrument cover the ranges relevant for the OIL values?); (d) the population in the area; (e) the consequence of no action (or action); (f) the use of the land; (g) conditions that may make implementation of response actions hazardous (e.g. restricting essential food or water, movement of patients, evacuation or sheltering under hazardous conditions); (h) social conditions; (i) the ability to define the area in a way understandable to the public; and (j) administrative and jurisdictional boundaries.

<sup>&</sup>lt;sup>14</sup> Some minor modifications to the assumptions made in Ref. [4] were introduced here (as indicated in Section 3) in order to reflect more realistic emergency conditions and to improve clarity.

The OIL charts provide minimum requirements that need to be met by the monitoring results for the OILs to be applicable. For example, the monitoring result needs to be representative of the dose received by the public (e.g. a dose rate from a ditch or sewer is not representative of an area inhabited by people). Selecting representative monitoring results will be heavily based on expert judgement, and providing a detailed guidance lies outside the scope of this publication.

Monitoring needs to be integrated into decision making systems in such a way that its use will not impede the implementation of effective response actions, especially for those actions that need to be taken before or shortly after a release to be effective. Not every type of monitoring will allow the effective implementation of response actions during a nuclear or radiological emergency:

- The ambient dose equivalent rate is readily available, easy to obtain (with simple instruments and little training) and representative of the expected radionuclide mixes. For a release of radioactive material from an NPP or its spent fuel, the expected radionuclide mixes will emit sufficient gamma radiation for the ambient dose equivalent rate to be an appropriate and effective indicator under emergency conditions for the amount of radioactive material present on the ground (OIL1<sub>γ</sub>, OIL2<sub>γ</sub> and OIL3<sub>γ</sub>), on the skin (OIL4<sub>γ</sub>) and in the thyroid (OIL8<sub>γ</sub><sup>15,16</sup>), making it unnecessary to measure beta or alpha radiation for this purpose. Although the use of the gamma dose rate is preferable over the beta count rate, an OIL4<sub>β</sub> for skin monitoring is provided because it may be used by some Member States.
- After the initial restrictions on food, milk and drinking water implemented on the basis of the declaration of a general emergency and on ground monitoring results (i.e. based on OIL1<sub>γ</sub>, OIL2<sub>γ</sub> and OIL3<sub>γ</sub>), food, milk and drinking water samples may be analysed to confirm these initial restrictions (as indicated in Fig. 2).<sup>17</sup> For this purpose, it is convenient to use marker radionuclides, which are representative of all the other radionuclides expected to be present and are easier to identify, avoiding the need for costly and time intensive comprehensive isotopic analyses. This publication provides OIL7 for the marker radionuclides I-131 and Cs-137. The use of OIL7 during a nuclear emergency is preferable over the use of OIL5 and OIL6 (given in Ref. [2]) because of the limited availability of time and resources early in an emergency. Once sufficient resources and time become available, OIL5 and OIL6 may be used, if considered necessary and justified, keeping in mind that the default OIL5 and OIL6 values provided in Ref. [2] (a) are applicable to any type of nuclear or radiological emergency; (b) are more conservative than OIL7; and (c) require determining the activity concentrations of all radionuclides present in food, milk and drinking water (and not only the activity concentration of a few radionuclides).

<sup>&</sup>lt;sup>15</sup> Only useful during the first week after intake of radioiodine.

<sup>&</sup>lt;sup>16</sup> Skin activity levels greater than the OIL4 $_{\gamma}$  or OIL4 $_{\beta}$  values indicate that the individual may have inhaled or ingested sufficient radioactive iodine before the monitoring has taken place to warrant monitoring of the thyroid with OIL8 $_{\gamma}$  and registration for later medical follow-up due to the risk of developing thyroid cancer and the need for early detection and effective treatment. It is early on (within the first few days) when valuable information can be easily obtained (e.g. the ambient dose equivalent rate from the thyroid, based on OIL8 $_{\gamma}$ ).

<sup>&</sup>lt;sup>17</sup> The analysis of samples will become relevant once sufficient samples have been collected, and time and resources are available for their analysis.

- OILs for rates or air concentrations in a plume<sup>18</sup> resulting from an ongoing release are not provided in this publication because, as indicated in Ref. [2]: (a) in many cases the significant release will be over by the time results of environmental measurements are available; (b) it is difficult to take and analyse air concentrations in a sample in a timely manner; and (c) there is a great variation in time and location of the plume concentrations at any location during a release. During the period of significant release, therefore, protective actions (e.g. evacuation) are best taken on the basis of the emergency classification system. Operating organizations of facilities at which there could be emergencies that might result in airborne releases of long duration need to develop EALs and possibly facility specific OILs for measurements taken in a plume, for possible airborne releases from the facilities.<sup>19</sup> Examples of OILs for dose rates in a release from an LWR resulting from core melt are provided in Ref. [5]. OILs for air concentrations arising from resuspension are not provided, because doses arising from resuspension have been considered in the OILs. Although not calculated for the purpose of in plume measurements, the default  $OIL1_{\gamma}$  and  $OIL2_{\gamma}$  values and associated response actions provided in this publication are consistent with the OILs for in plume measurements provided in Table B2 of Ref. [5]. There is no need to confirm that the measurement is not being taken in the plume for the default  $OIL1_{\gamma}$  and  $OIL2_{\gamma}$  values provided in this publication.
- Further OILs will need to be developed based on the new generic criteria given in Ref. [1] for enabling the transition to an existing exposure situation (para. II.15 of Ref. [1]), for vehicles, equipment and other items (Table II.4 of Ref. [1]) and for food and other commodities traded internationally (Table II.5 of Ref. [1]). International criteria for trade are established in Refs [1, 14, 15].

Once actions have been taken based on representative, readily available and practical measurement, and once the greatest risk to the public has thus been alleviated, there may be time and resources available to perform more detailed measurements, such as spectrometry or radiochemical analyses.

#### 2.3. COMMUNICATING WITH DECISION MAKERS

Experience has shown that decision makers take actions best when they understand how the actions contribute to the safety of the public [16]. The information needs to be conveyed to decision makers in plain and simple language, indicating (a) what response actions need to be implemented when and where; (b) why they need to be implemented; and (c) how they will contribute to the safety of the public.<sup>20</sup> When communicating the assessment of the monitoring results to decision makers, the following points ought to be clearly addressed:

• The importance of avoiding unnecessary delays in decision making. Postponing response actions may result in radiation induced health effects that could have been avoided.

<sup>&</sup>lt;sup>18</sup> Airborne radioactive material released from the NPP.

<sup>&</sup>lt;sup>19</sup> In plume measurements are considered within the emergency classification system to detect a release of radioactive material (and classify the emergency accordingly), but are not considered suitable to determine specific locations warranting response actions.

<sup>&</sup>lt;sup>20</sup> For example, a map showing monitoring results in combination with default OIL values does not explain to a layperson what response actions need to be implemented where, when, how and why.

- The need to implement the response actions safely and without causing more harm than good (e.g. avoiding dangerous travel conditions).
- The quality and reliability of the monitoring results being used and whether future refinements are expected.
- Using default OIL values above or below those agreed upon during the preparedness stage needs to be avoided unless there is a clear justification for it.<sup>21</sup> There is no need to add conservatism at the time of the emergency (like using half the default OIL value). Doing so may result in more harm than good. If changes are made to the default OIL values to account for prevailing conditions during an emergency, the reasons need to be based on defensible considerations and explained to the public in a plain and understandable form.
- The goal of the response actions is to protect all members of the public (including children and pregnant women) by minimizing radiation induced health effects.<sup>22</sup> Below the OIL values, no radiation induced health effects are expected to be observed, even in a very large exposed group composed of the most sensitive members of the public.
- Radiation induced health effects can only be assessed properly by experts (others, such as local physicians, may not have the necessary expertise).<sup>23</sup> This needs to be clearly conveyed to pregnant women of the affected areas to prevent actions that do more harm than good.
- Radiation induced thyroid cancers are the greatest concern among the possible radiation induced health effects following a release of radioactive material from a reactor core or spent fuel.<sup>24</sup> A discernible increase in the incidence of any other radiation induced cancers (e.g. leukaemia) among the public after such a release is considered very unlikely and would require that many people receive doses sufficient to result in severe deterministic effects [17, 18, 19]. There is no scientific evidence of an increase in the incidence of radiation induced cancers, other than thyroid cancers, among the public that could be related to radiation exposure from the Chernobyl accident [20, 21, 22, 23].<sup>25</sup>
- Emphasis has to be placed on the need to inform medical practitioners that universal precautions against infection (gloves, mask, etc.) will provide sufficient protection from the radioactive material (irrespective of the level of contamination).<sup>26</sup>

<sup>&</sup>lt;sup>21</sup> Section 5 addresses what needs to be considered before modifying the default OIL values.

<sup>&</sup>lt;sup>22</sup> By avoiding severe deterministic effects or a discernible increase in the incidence of stochastic effects (e.g. cancers).
<sup>23</sup> There are only a limited number of experts in diagnosing and treating the health effects of radiation exposure in the

<sup>&</sup>lt;sup>23</sup> There are only a limited number of experts in diagnosing and treating the health effects of radiation exposure in the world. Medical examinations, treatments and counselling need to be done only after consultation with those experts. Assistance can be obtained through the IAEA or WHO.

<sup>&</sup>lt;sup>24</sup> This is because large amounts of radioactive iodine may be released which may be inhaled or ingested. Once inhaled or ingested, the radioactive iodine concentrates in the thyroid resulting in very high doses to the organ.

<sup>&</sup>lt;sup>25</sup> To date, only a discernible increase in radiation induced thyroid cancers among the population group aged 0–18 years old (in 1986) living in the areas of Belarus, Russian Federation and Ukraine affected by the Chernobyl accident has been clearly established [23]. These cancers were due principally to doses received from drinking milk from cows grazing on pasture contaminated with radioactive iodine and were observed out to very long distances from the site (i.e. hundreds of kilometres). Radiation induced thyroid cancers started to appear in 1990, four years following the Chernobyl accident [22, 23]. These cancers are usually not life threatening if detected and treated early. For this reason, people who may have inhaled or ingested radioactive iodine need to be registered and have their dose estimated to determine whether a medical follow-up is warranted. OIL8, was developed for this purpose.

<sup>&</sup>lt;sup>26</sup> In several past emergencies [6, 8, 10], some medical staff refused to treat potentially contaminated patients, because they did not understand how to protect themselves from contamination.

#### 2.4. COMMUNICATING WITH PUBLIC INFORMATION OFFICERS

The decisions taken by the decision maker need to be explained in plain and simple language to public information officers, who will be responsible for the direct communication with the public. The information provided to the public information officers needs to address the main concern of the public, i.e. **"Am I safe? And if not, what should I do to be safe?"** The system for putting the radiological health hazard in perspective in emergency situations provided in Refs [4, 24, 25] can be used for this purpose. The so called 'perspective charts' provided in Sections 7.2. and 7.3. of Ref. [4] (and displayed as thumbnails in Fig. 3 below) can be used to relate a measurement result with the associated health hazard. Further guidance regarding the communication with the public is available in Refs [26, 27].

The importance of public communication is highlighted when considering examples of actions taken by the public and decision makers during previous nuclear and radiological emergencies that were not warranted based on the radiological health hazard and resulted in unnecessary harm, such as [25]:

- Artificially terminating pregnancies [28];
- Performing unsafe evacuations [10];
- Medical staff being reluctant to treat possibly contaminated patients [29, 30, 31];
- Demanding unwarranted medical examinations that interfere with the treatment of those who are most at risk [29];
- Pregnant women worrying about their fetus [31, 32, 33, 34];
- Using inappropriate substitutes for iodine thyroid blocking (ITB) agents [31];
- Stigmatizing those from the affected area [7, 29, 32];
- Worrying about the possibility of radiation induced cancers [31];
- Rejecting products from the affected area [32];
- Limiting tourism and implementing restrictions on agriculture [29, 35];
- Cancelling necessary nuclear medical treatments due to fear of radiation [36].



FIG. 3. Charts for placing the radiological health hazard in perspective (provided in Ref. [4]).

#### 2.5. HOW TO USE THE OIL CHARTS

The charts can be used by following seven simple steps:

- Step 1 Obtain monitoring results for one of the monitoring types listed in Table 1 and check their quality and reliability, as described in Section 2.2.
- **Step 2** Select the OIL chart applicable to the monitoring type, as indicated in Table 1.
- Step 3 Check if the chart is applicable to your situation by using the checklist at the beginning of the chart.
- Step 4 Check the purpose within the protection strategy. Section 2.1 provides further advice in this regard.
- Step 5 Compare your monitoring results with the default OIL value(s), ensuring they have the same units and prefixes as the default OIL value(s).
- **Step 6** Recommend taking the response actions indicated in the charts to the decision maker(s). When communicating with the decision maker(s), ensure that the information is clear, concise and goal oriented, as described in Section 2.3. Note that some actions are recommended to be taken whether or not the default OIL value is exceeded.
- Step 7 Place the health hazard in perspective when communicating with decision makers and public information officers, as described in Section 2.4.





# 2.6. OIL CHARTS FOR USE IN THE RESPONSE

# 2.6.1. Ground monitoring (OIL1 $_{\gamma}$ , OIL2 $_{\gamma}$ and OIL3 $_{\gamma}$ )

OIL CHART FOR GROUND MONITORING (LWRs)					
	ATTENTION: Only use this OIL if the answer to all the following questions is 'yes'.				
	Has there been a release of radioactive material from an LWR or its spent fuel?				
CHECKLIST	Are you assessing the ambient dose equivalent rate at 1 m above ground level? Ures I No				
	Is the measurement representative of an area inhabited or frequented by the public or of an area from where the public consumes or distributes local produce, wild-grown products (e.g. mushrooms or berries), milk from grazing animals, rainwater, local animals or animal feed?				
PURPOSE WITHIN THE PROTECTION STRATEGY	<ul> <li>To be used to identify areas (beyond those for response actions have been taken based on the emergency classification) where the ground deposition of radioactive material warrants:</li> <li>Protection of the public frequenting or living in the area (by using OIL1<sub>Y</sub> for immediate urgent response actions and OIL2<sub>Y</sub> for early response actions);</li> <li>Restricting the consumption, distribution and sale of non-essential local produce, wild-grown products, milk from grazing animals, directly collected rainwater, local animals and animal feed (by using OIL3<sub>Y</sub> for immediate urgent response actions).</li> <li>Avoid delays in decision making and implement response actions as soon as possible. Living in areas exceeding OIL1<sub>Y</sub> for more than 1 day may result in radiation induced health effects. Those living in areas exceeding OIL2<sub>Y</sub> will receive a large fraction of the annual dose in the first month. Consuming local produce, wild-grown products, milk from grazing animals from an area exceeding OIL3<sub>Y</sub>, may result in radiation induced health effects.</li> </ul>				
MONITORING TYPEAmbient dose equivalent rate at 1 m above ground level in a populated or frequented area or in an area used for farming or for grazing, ideally with low or no vegetation and away from roads, trees and buildings.					
	OIL1 <sub>γ</sub> = 1000 μSv/h				
DEFAULT OIL VALUE	$OIL2_{v} = \frac{100 \ \mu Sv/h}{= 25 \ \mu Sv/h} = \frac{For the first 10 days after reactor shutdown}{Later than 10 days after reactor shutdown or for spent fuel.}$				
	$OIL3_{\gamma} = 1 \mu Sv/h$ above background.				
<ul> <li>Response actions to be implemented upon declaration of a General Emergency and following a release (before monitoring is implemented):</li> <li>Within the Extended Planning Distance (EPD): Instruct the public to reduce inadvertent ingestion, by advising: (a) to wash hands before drinking, eating or smoking or touching the face; (b) not to let children play on the ground; and (c) to avoid activities resulting in the creation of dust that could be ingested or inhaled.</li> <li>Within the Ingestion and Commodities Planning Distance (ICPD): Instruct the public to stop consumption, distribution and sale of non-essential local produce, wild-grown products, milk from grazing animals, directly collected rainwater, local animals (unless fed with protected feed) and animal feed, until the activity concentrations have been assessed by using OIL7. If the restricted food, milk or drinking water is essential, replace it.</li> <li>Within the ICPD: Instruct the public to stop distribution of commodities that may have been contaminated until they have been assessed.</li> </ul>					
	THE CHART CONTINUES ON THE NEXT PAGE				

THE CHART BEGINS ON THE PREVIOUS PAGE		
	Within the first day after the beginning of the exposure:	
RESPONSE ACTIONS IF OIL1 <sub>Y</sub> IS EXCEEDED	<ul> <li>Instruct the public to safely evacuate (only if it does not endanger those being evacuated; for example, patients in hospitals or care homes do not need to be immediately evacuated if this will put them at risk), if possible in combination with iodine thyroid blocking (only if it will not delay evacuation). If immediate evacuation is not possible or safe (e.g. for special facilities or owing to snow, floods or lack of transport), instruct the public to shelter preferably in large buildings in combination with iodine thyroid blocking, until safe evacuation is possible.</li> <li>Once evacuated, provide registration, skin and thyroid monitoring (by using OIL4<sub>Y</sub> and OIL8<sub>Y</sub>), decontamination and medical screening for the evacuees. Instruct them to shower and change clothing, if it can be done safely (e.g. do not change or shower in cold temperatures). Skin and thyroid monitoring is not essential and does not warrant delaying other urgent response actions.</li> </ul>	
	Within weeks after the beginning of the exposure:	
	<ul> <li>Estimate the dose from all exposure pathways for those who were in the area to determine if medical follow-up is warranted in accordance with the IAEA Safety Standards Series Nos GSR Part 7 [1] and GSG-2 [2].</li> </ul>	
	Also implement the response actions indicated for $OIL3_{\gamma}$ .	
RESPONSE ACTIONS IF OIL2γ IS EXCEEDED	<ul> <li>Within weeks after the beginning of the exposure:</li> <li>Register those living in the area.</li> <li>Safely relocate those living in the area (i.e. do not endanger those being relocated). Prioritize those in the areas of highest potential exposure ensuring that those areas approaching OIL1<sub>Y</sub> will be relocated at least within the first days and those areas approaching OIL2<sub>Y</sub> will be relocated at least within the first month.</li> <li>Estimate the dose from all exposure pathways for those who were living in the area to determine if medical follow-up is warranted in accordance with the IAEA Safety Standards Series Nos. GSR Part 7 [1] and GSG-2 [2].</li> <li>Also implement the response actions indicated for OIL3<sub>Y</sub>.</li> </ul>	
	Within the first day after the beginning of the exposure:	
RESPONSE ACTIONS IF OIL3 <sub>V</sub> IS	<ul> <li>If not already implemented based on the declaration of a General Emergency, implement the response actions indicated in the "RESPONSE ACTIONS BASED ON GENERAL EMERGENCY" section on the previous page regardless of the distance from the NPP.</li> <li>Within weeks after the beginning of the exposure:</li> </ul>	
EXCEEDED	Estimate the dose from all exposure pathways for those who may have consumed local produce, wild-grown products milk from grazing animals, directly collected rainwater and local animals from the area where restrictions were implemented to determine if medical follow-up is warranted in accordance with the IAEA Safety Standards Series Nos. GSR Part 7 [1] and GSG-2 [2].	
IF NONE OF THE OILS IS EXCEEDED	Adjust food, milk and drinking water restrictions by using OIL7 once food, milk and drinking water samples have been taken and analysed.	
HEALTH HAZARD	Chart 1 in Section 7 of IAEA EPR-NPP Public Protective Actions (2013) [4] can be used to place the <b>health hazard in perspective</b> when communicating with decision makers and public information officers.	

## 2.6.2. Skin monitoring (OIL4 $_{\gamma}$ and OIL4 $_{\beta}$ )

	OIL CHART FOR SKIN MONITORING (LWRs) - GAMMA	
	ATTENTION: Only use this OIL if the answer to all the following questions is 'yes'.	
CHECKLIST	Has there been a release of radioactive material from an LWR or its spent fuel?	
	Are you assessing the ambient dose equivalent rate from the bare skin of the hand	
PURPOSE WITHIN THE PROTECTION STRATEGY	To be used to identify individuals with enough radioactive material on the skin to warrant response actions (such as decontamination). Only the public being evacuated or relocated is expected to possibly have sufficient radioactive material on the skin to warrant response actions. However, for reassurance, OIL4 <sub>y</sub> may be used with other members of the public as well. <b>Monitoring of the skin will only be effective over the first few days.</b> After a few days, most of the radioactive material will have been removed from the skin by natural processes. Keep in mind that the risk to health from skin contamination is small, and therefore monitoring or decontamination of the skin does not warrant delaying or interfering with more important response actions (e.g. sheltering, evacuation, treatment of injured individuals or patients).	
MONITORING TYPE	Ambient dose equivalent rate at 10 cm from the bare skin of the hand and face conducted in an area with a background of less than $0.5 \mu$ Sv/h.	
DEFAULT OIL VALUE	OIL4 <sub>γ</sub> = 1 μSv/h above background.	
	Within the first hours after beginning of the exposure (before monitoring is implemented):	
RESPONSE ACTIONS FOR ALL THOSE THAT MAY BE MONITORED	<ul> <li>The primary concern from radioactive material on the skin is from inadvertent ingestion of the material. Thus, a person can be protected by taking such simple and non-disruptive measures as: (a) washing the hands before drinking, eating, smoking or touching the face; (b) not letting children play on the ground; and (c) avoiding activities resulting in the creation of dust that could be ingested or inhaled.</li> <li>Instruct to change clothing and shower as soon as possible, if it can be done safely (e.g. do not change or shower in cold temperatures).</li> <li>Reassure those treating and/or transporting contaminated individuals that they can do so safely if they use universal precautions against infection (i.e. gloves, mask, etc.).</li> </ul>	
	Within the first days after the beginning of the exposure:	
RESPONSE ACTIONS IF OIL4 <sub>Y</sub> IS EXCEEDED	<ul> <li>Register all those being monitored and record the monitoring result (if practical).</li> <li>Provide for additional decontamination (apart from the simple decontamination measures mentioned above) by means considered appropriate and safe.</li> <li>Monitor the thyroid by using OIL8<sub>y</sub>.</li> <li>Provide medical screening.</li> <li>Instruct to take iodine thyroid blocking agents (if not already taken and only within the first days after reactor shutdown) to reduce further uptake of radioiodine. WHO guidance needs to be followed in this regard [37].</li> </ul>	
	Within weeks after the beginning of the exposure:	
	<ul> <li>Estimate the dose from all exposure pathways for those exceeding OIL4<sub>Y</sub> to determine if medical follow-up is warranted in accordance with the IAEA Safety Standards Series Nos. GSR Part 7 [1] and GSG-2 [2].</li> </ul>	
IF OIL4 <sub>Y</sub> IS <u>NOT</u> EXCEEDED	Check OIL8 <sub>y</sub> .	
HEALTH HAZARD	Chart 2 in Section 7 of IAEA EPR-NPP Public Protective Actions (2013) [4] can be used to place the <b>health hazard in perspective</b> when communicating with decision makers and public information officers.	

OIL CHART FOR SKIN MONITORING (LWRs) - BETA					
<b>CAUTION:</b> The ambient dose equivalent rate <b>OIL4</b> <sub>Y</sub> is sufficient and preferable to assess the levels of radioactive material on the skin for a release of radioactive material from an LWR or its spent fuel, because it is less dependent of the measurement technique and instrument characteristics. However, beta count rate OIL4 <sub>β</sub> is also provided, since may be used by some response organizations for skin monitoring.					
	ATTENTION: Only use this OIL if the answer to all the following questions is 'yes'.				
	Has there been a release of radioactive material from an LWR or its spent fuel?				
CHECKLIST	Are you assessing the beta count rate from the bare skin of the hand and face?				
	Does the beta monitoring instrument meet <b>all</b> of the suitability criteria of Section $\Box$ Yes $\Box$ No 3.7.2.3 below (e.g. the effective window area of the instrument is less than or equal to 50 cm <sup>2</sup> )?				
PURPOSE WITHIN THE PROTECTION STRATEGY	The same as indicated in the gamma chart for skin monitoring.				
MONITORING TYPE	Beta count rate at 2 cm from the bare skin of the hand and face conducted in an area with a background of less than 0.5 µSv/h.				
	OIL4 <sub>β</sub> = 1000 cps				
DEFAULT OIL VALUE	The default OIL4 $_{\beta}$ provided here can be used for a large variety of instruments. However, if the specific properties (e.g. efficiency and detector area) of the instrument are known, its suitability needs to be confirmed, or an instrument specific default OIL4 $_{\beta}$ value needs to be calculated during the preparedness stage following Section 5.2 of this publication.				
RESPONSE ACTIONS FOR ALL THOSE BEING MONITORED	The same as indicated in the gamma chart for skin monitoring.				
RESPONSE ACTIONS IF OIL4 <sub>β</sub> IS EXCEEDED	The same as indicated in the gamma chart for skin monitoring.				
IF OIL4 <sub>β</sub> IS <u>NOT</u> EXCEEDED	The same as indicated in the gamma chart for skin monitoring.				
HEALTH HAZARD	The same as indicated in the gamma chart for skin monitoring.				

### 2.6.3. Food, milk and drinking water samples (OIL7)

OIL CHART FOR FOOD, MILK AND DRINKING WATER SAMPLES (LWRs)					
	ATTENTION: Only use this OIL if the answer to all the following questions is 'yes'.				
	Has there been a release of radioactive material from an	LWR or its spent fuel?			
	Are you assessing the activity concentrations of radioact or drinking water?	ive material in food, milk 🗌 Yes 🗌 No			
CHECKLIST	Have you determined the activity concentration of <u>both</u> marker radionuclides I-131				
	Is the measurement representative of the food, milk or drinking water that is being Yes No consumed?				
	If you are analysing milk: Are you aware that the activity concentration of I-131 Yes No and Cs-137 in milk will not reach its maximum in the first day(s) after grazing on contaminated pasture (e.g. approximately two days for cows)?				
	To be used to confirm and adjust initial restrictions on initial restrictions that were implemented based on the d $OIL2_{\gamma}$ or $OIL3_{\gamma}$ ).	food, milk and drinking water (i.e. those eclaration of the emergency or on $OIL1_{\gamma},$			
PROTECTION STRATEGY	The collection and analysis of samples will require Comprehensive results are not expected within the fin emergency; hence, there is time for a more detailed asso or water restrictions are implemented, the lower the dose	much time and substantial resources. rst weeks (and even months) after the essment. However, the sooner food, milk received by the public will be.			
MONITORING TYPE	Activity concentration of I-131 and Cs-137 in food, milk and drinking water samples (these are marker radionuclides, i.e. the contribution of all other radionuclides expected to be present was considered in the calculation of OIL7).	<sup>90</sup> Sr, <sup>131</sup> I, <sup>134</sup> Cs <sup>137</sup> CS,			
DEFAULT OIL VALUE	The activity concentration of <b>both</b> marker radionuclides I-131 and Cs-137 needs to be determined, and the OIL is exceeded if <b>either</b> the I-131 or Cs-137 value is exceeded.				
	OIL7 = 1000 Bq/kg of I-131 or 200 Bq/kg of Cs-137	MONITORING OF FOOD, MILK AND DRINKING WATER SAMPLES			
	Within days after obtaining the results:				
RESPONSE	<ul> <li>Instruct the public to stop consumption, distribution and sale of the affected food, milk or drinking water (only if it can be implemented safely). If the restricted food, milk or drinking water is essential, replace it.</li> </ul>				
ACTIONS IF OIL7 IS	Within weeks after obtaining the results:				
EXCEEDED	<ul> <li>Estimate the dose from all exposure pathways for those who may have consumed food, milk or drinking water with activity concentrations greater than OIL7 to determine if medical follow-up is warranted in accordance with the IAEA Safety Standards Series Nos. GSR Part 7 [1] and GSG-2 [2].</li> </ul>				
IF OIL7 IS <u>NOT</u> EXCEEDED	IF OIL7 IS NOT EXCEEDED No need for further restrictions.				
HEALTH HAZARD	Charts 3A, 3B, 4A and 4B in Section 7 of IAEA EPR-NPP Public Protective Actions (2013) [4] can be used to place the <b>health hazard in perspective</b> when communicating with decision makers and public information officers.				

### **2.6.4.** Thyroid monitoring (OIL8 $_{\gamma}$ )

OIL CHART FOR THYROID MONITORING (LWRs)				
	ATTENTION: Only use this OIL if the answer to all the following questions is 'yes'.			
	Has there been a release of radioactive material from an LWR or its spent fuel?			
	Are you assessing the ambient dose equivalent rate measured in front of the $\Box$ Yes $\Box$ No thyroid in contact with the skin?			
CHECKLIST	Was the person decontaminated and the contaminated outer clothing removed $\Box$ Yes $\Box$ No before monitoring?			
	Was the measurement taken in the first week after the possible intake of I-131?			
	Did you use an instrument with an effective window area of $\leq 15 \text{ cm}^2$ and a $\square$ Yes $\square$ No response of $\geq 0.1 \ \mu$ Sv/h per kBq of I-131 activity in the thyroid? Further details are given in Section 3.7.1.3.			
	To be used to identify individuals warranting registration and medical follow-up due to the intake of radioiodine, i.e. evacuated public or those that have ingested local produce, wild-grown products, milk from grazing animals, directly collected rainwater or local animals from an area exceeding $OIL3_{\gamma}$ . For reassurance, $OIL8_{\gamma}$ may be used with other members of the public as well.			
PROTECTION STRATEGY	The thyroid needs to be monitored within the first week to detect if an individual has inhaled or ingested sufficient radioiodine to warrant medical follow-up. Identifying the individuals is difficult later on. The early identification of those with an increased risk of developing thyroid cancer is paramount in their later medical follow-up and treatment. However, keep in mind that monitoring of the thyroid does not warrant delaying or interfering with other urgent response actions.			
	Ambient dose equivalent rate in front of the thyroid:			
MONITORING TYPE	<ul> <li>In contact with the skin;</li> <li>Measured within the first week after the intake of radioiodine;</li> <li>Conducted in an area with a background of less than 0.25 μSv/h;</li> <li>Measured after the person has been decontaminated and contaminated outer clothing has been removed; and</li> <li>Measured with an instrument with an effective window area of ≤ 15 cm<sup>2</sup> and a response of ≥ 0.1 μSv/h (ambient dose equivalent rate in front of the thyroid in contact with the skin) per kBq of I-131 activity in the thyroid (as described in Section 3.7.1.3.).</li> </ul>			
DEFAULT OIL VALUE	<b>OIL8</b> <sub>y</sub> = 0.5 $\mu$ Sv/h above background.			
	Before monitoring:			
RESPONSE ACTIONS FOR ALL THOSE TO BE MONITORED	<ul> <li>Instruct those to be monitored to reduce inadvertent ingestion by: (a) washing their hands before drinking, eating or smoking or touching the face; (b) not letting children play on the ground; and (c) avoiding activities resulting in the creation of dust that could be ingested or inhaled.</li> <li>Instruct those to be monitored to change clothing and shower as soon as possible, if it can be done safely (e.g. do not change or shower in cold temperatures).</li> <li>Reassure those treating and/or transporting contaminated individuals that they can do so safely if they use universal precautions against infection (gloves, mask, etc.).</li> </ul>			
	Immediately following the monitoring:			
RESPONSE ACTIONS IF OIL8 <sub>Y</sub> IS EXCEEDED	<ul> <li>Register all those monitored and record the monitoring result.</li> <li>Instruct to take iodine thyroid blocking agents to reduce further uptake of radioiodine (if not already taken and only within the first days after reactor shutdown). WHO guidance needs to be followed in this regard [37].</li> <li>Provide medical screening.</li> </ul>			
	Within weeks after the beginning of the exposure: Estimate the dose from all exposure pathways for those exceeding $OIL8_{V}$ to determine if a medical follow-up is warranted in accordance with the IAEA Safety Standards Series Nos. GSR Part 7 [1] and GSG-2 [2].			
<b>IF OIL8</b> <sub>Y</sub> <b>IS</b> <u>NOT</u> Register all those monitored and record the monitoring result (if practical). No further action necessary.				

#### 2.7. INTEGRATION OF THE CHARTS INTO NATIONAL ARRANGEMENTS

These charts need to be integrated in the overall protection strategy, incorporated into national emergency arrangements and adapted to local and site specific circumstances, ensuring that they remain practical for the response to an actual emergency (e.g. that the instructions are clear and unambiguous, and that the units are applicable to the monitoring results expected to be available during an emergency). Once revised, the charts need to be tested in drills and exercises and included in training programmes. Revisions of the default OIL values need to be considered with caution following the recommendations given in Section 5.

#### 3. METHODOLOGY FOR DERIVING THE DEFAULT OIL VALUES

#### 3.1. OVERVIEW

This section provides the methodology for deriving the default OIL values provided in this publication, giving a detailed explanation of the underlying assumptions, references and calculations. The default OIL values are intended to be used directly and confidently during the response. This requires making a series of considerations during the preparedness stage, as indicated in Fig. 4. The considered elements are shortly described below and discussed in detail in the sections indicated in Fig. 4:

- Establish generic criteria at which to implement response actions: The OILs provided in this publication are based on the generic criteria given in the IAEA Safety Standards Series No. GSR Part 7 [1] and described in Section 3.2. Generic criteria are projected or received doses at which response actions are to be taken in a nuclear or radiological emergency. Below the generic criteria, no radiation induced health effects are expected to be observed, even in a very large exposed group composed of the most sensitive members of the public. Being projected or received doses (i.e. calculated quantities), generic criteria cannot be used directly in the response, hence the need to develop operational criteria (such as OILs).
- **Consider all relevant radionuclide mixes:** The potential health effects from exposure to radioactive material, as well as the response of the monitoring instrument and the selection of marker radionuclides, depend on the radionuclides present. The OILs are therefore calculated for all the radionuclide mixes expected to be released from an LWR or its spent fuel during a severe emergency which may be significant contributors to the dose of the public or to the instrument response, as described in Section 3.3.
- Consider all individuals being exposed: Experience from past nuclear and radiological emergencies [6] has shown that assessments not considering all members of the public (especially those most sensitive, such as children and pregnant women), or not clearly stating that they have been considered, may result in unwarranted actions being taken by the public and/or decision makers that do more harm than good (examples of such actions are provided in Section 2.4). All members of the public have been considered in the calculation of the OILs by taking response actions based on the dose projected or received by the representative person, as described in Section 3.4.
- Consider all relevant exposure scenarios and associated pathways: The exposure scenario is a postulated set of conditions, circumstances, events and behaviour of the public that characterizes the exposure situation. It is the basis for determining the potentially exposed individuals, the relevant exposure pathways and the effectiveness of the response actions. Five exposure scenarios resulting from deposited radioactive material are considered for the OILs provided in this publication, i.e. the 'ground', 'food pre-analysis', 'skin', 'food post-analysis' and 'thyroid' scenarios, as described in Section 3.4. For each exposure scenario, different exposure pathways are considered. These pathways describe the routes by which radiation or radionuclides can reach humans and cause exposure.



FIG. 4. Elements considered at the preparedness stage for developing the OILs and OIL charts.

- **Consider the behaviour of the radionuclides:** Any behaviour of the radionuclides that could have a significant impact on the dose or the OILs needs to be considered, such as the decrease in the dose rate due to decay and weathering, resuspension, transfer from the ground to milk or food or inadvertent ingestion. Section 3.5 considers the behaviour of the radionuclides (after being released into the environment)<sup>27</sup> and its impact on the dose by deriving a series of factors used in the OIL calculations.
- Determine the dose conversion factors to perform the relevant organ dose calculations: Dose conversion factors relate the activity of a certain radionuclide with the projected dose, which is needed to determine if the generic criteria may be exceeded. Section 3.6 provides the dose conversion factors for the exposure scenarios described in Section 3.4. The dose conversion factor will depend on the radionuclide, the dose quantity, the exposure pathways, the exposure scenario, the exposed individual and other considerations.
- **Consider the instrument response:** The instrument response will affect the default OIL values and needs to be considered in the calculations. Section 3.7 describes the basis for the instrument response assumed in the calculation of the different OILs.

 $<sup>^{27}</sup>$  The behaviour of the radionuclides within the NPP is addressed through the fuel damage scenarios given in Section 3.3.

- Calculate the time and mix dependent OIL(t,mix) functions and select a default OIL value: OIL values depend on the radionuclide mix that is considered, which will vary over time (due to processes such as decay). Therefore, for each OIL a set of time and mix dependent OIL(t,mix) functions is calculated, based on which a default OIL value is chosen, as described in Section 3.8. Choosing a default value over a time and/or mix dependant OIL is considered necessary because (a) the mix can vary considerably during an emergency with time and location [6, 7, 8, 9, 10]; and (b) not having a default criterion for the implementation of response actions has led to confusion of decision makers and scepticism among the public in past emergencies, delaying urgently required response actions [6, 16].
- Develop a set of response actions within a justified and optimized protection strategy: The charts in Section 2 provide the default OIL values, together with associated response actions. These response actions were developed by taking into consideration the overall protection strategy outlined in Ref. [1] and the concept of operations outlined in Ref. [4]. For a detailed description of the response actions, refer to Section 5 of Ref. [4].
- Prepare to communicate with decision makers and public information officers: During past nuclear and radiological emergencies, members of the public, those responsible for protecting the public and others (e.g. medical staff) have, on occasion, taken damaging actions that were not warranted based on the radiological health hazard. In some cases, these actions have resulted in avoidable deaths, injuries and an increased risk to health. In addition, these actions had adverse economic, social and psychological consequences. These adverse effects are also known as 'non-radiological' effects, which can be the most severe consequences of an emergency [6]. Experience has shown that decision makers take actions best when they understand how the actions contribute to the safety of the public [16]. Similarly, the public are more likely to follow instructions when they are clear and understandable in their own best interests. For this reason, the IAEA Safety Standards Series No. GSR Part 7 [1] requires in para. 5.71: "Arrangements shall be made so that in a nuclear or radiological emergency information is provided to the public in plain and understandable language." Further details can be found in Section 2.3. Providing detailed guidance on public communication lies outside of the scope of this publication.<sup>28</sup>

Due to the large number of assumptions and uncertainties taken into account in the derivation of the OILs (a) the results should not be considered as an exact value, but rather as a rough estimate, and (b) a reasonably conservative<sup>29</sup> approach is needed (as described in Section 3.10) to ensure that the response actions protect the public effectively from the radiological health hazard. However, using much lower default OIL values than those provided in this publication (e.g. one half of a default OIL value) may not be justified, since it could result in response actions that do more harm than good when considering: (a) the health hazard associated with the response action itself, and (b) the diversion of limited resources from the highest priority actions during an emergency.

<sup>&</sup>lt;sup>28</sup> Further guidance regarding the communication with the public is available in Refs [26, 27].

<sup>&</sup>lt;sup>29</sup> Conservative meaning that it will result in a projected dose higher than the dose actually expected to be received under real conditions. Section 3.10 discusses the considered conservatism.

#### 3.2. GENERIC CRITERIA

The OILs provided in this publication are based on the generic criteria given in Tables II.1, II.2 and II.3 of the IAEA Safety Standards Series No. GSR Part 7 [1].<sup>30</sup> Only a subset of those generic criteria is needed for deriving the OILs given in this publication, as described in Table 2. Keeping the dose below the generic criteria given in Table 2 will ensure that none of the generic criteria listed in Tables II.1, II.2 and II.3 of Ref. [1] will be exceeded for a release from an LWR or its spent fuel.

Generic criteria are projected or received doses at which response actions are to be taken in a nuclear or radiological emergency. The generic criteria are established at doses below those at which radiation induced health effects would be expected to be observed, even in a very large exposed group composed of the most sensitive members of the public (e.g. children and pregnant women). Therefore, while implementing response actions above those generic criteria would almost always be justified on radiation protection grounds [38], below these generic criteria response actions may not be justified on radiation protection grounds, requiring a special and careful consideration before their implementation. In all cases, regardless whether the projected or received doses are above or below the generic criteria, the response actions need to be justified (i.e. do more good than harm) and optimized, taking the overall protection strategy into account.

Projected or received doses cannot be directly measured or easily calculated early in an emergency when information is limited and uncertainties significant, and decisions need to be made quickly in order for the actions to be effective. Hence the need to develop operational criteria, such as OILs, that can be used directly in the response. If a response action is implemented soon enough, the majority of the projected dose can be averted and the risk of suffering severe deterministic effects or incurring an increased risk of stochastic effects significantly reduced.

A detailed description of the basis for the generic criteria can be found in Ref. [19]. Member States may decide to adopt the IAEA's generic criteria directly or develop national generic criteria on the basis of the outcome of the justification and the optimization of their protection strategy.

<sup>&</sup>lt;sup>30</sup> This publication does not contain OILs for the following generic criteria provided in Ref. [1]: (a) for vehicles, equipment and other items (table II.4 of Ref. [1]); (b) for food and other commodities traded internationally (Table II.5 of Ref. [1]); and (c) for enabling transition to an existing exposure situation (para. II.15 of Ref. [1]).

Actions <sup>a</sup>	Generic criterion (GC)	Used for	References
Take response actions under any circumstance to avoid or minimize severe deterministic effects	$GC(Acute,AD_{skin-ext},10h) = 10 \text{ Gy}$ RBE weighted absorbed dose to 100 cm <sup>2</sup> of the skin dermis of the representative person <sup>b</sup> from acute external exposure in the first 10 hours	$\begin{array}{c} {\rm OIL4}_{\gamma} \\ {\rm OIL4}_{\beta} \end{array}$	Table II.1 of Ref. [1]
	<b>GC(Urgent,E,7d) = 0.1 Sv</b> total <sup>c</sup> effective <sup>d</sup> dose to the representative person <sup>b</sup> in the first 7 days	$\begin{array}{c} OIL1_{\gamma}\\ OIL4_{\gamma}\\ OIL4_{\beta} \end{array}$	Table II.2 of Ref. [1]
Take <b>urgent</b> response actions to reduce the risk of stochastic effects	$GC(Urgent, H_{fetus}, 7d) = 0.1 Sv$ total <sup>e</sup> equivalent dose to the fetus <sup>f</sup> in the first 7 days	$\begin{array}{c} OIL1_{\gamma}\\ OIL4_{\gamma}\\ OIL4_{\beta} \end{array}$	Table II.2 of Ref. [1]
	$GC(Urgent,h_{thyroid,thy-burden}) = 0.1 \text{ Sv}^{g}$ committed <sup>h</sup> equivalent dose to the thyroid from radioiodine in the thyroid (thyroid burden)	OIL8 <sub>γ</sub>	Ref. [1] <sup>i</sup>
Take <b>early</b> response actions	GC(Early,E,1a) = 0.1 Sv total <sup>c</sup> effective <sup>d</sup> dose to the representative person <sup>b</sup> in the first year	$OIL2_{\gamma}$	Table II.2 of Ref. [1]
stochastic effects	$GC(Early, H_{fetus}, 9mo) = 0.1 \text{ Sv}$ total <sup>e</sup> equivalent dose to the fetus <sup>f</sup> in the full period of in utero development	OIL2 <sub>γ</sub>	Table II.2 of Ref. [1]
Take response actions to reduce the risk of stochastic	<b>GC(Ingestion, <math>e_{ing}</math>, 1a) = 0.01 Sv<sup>g</sup></b> committed <sup>h</sup> effective <sup>d</sup> dose to the representative person <sup>b</sup> from ingestion of food, milk and drinking water during the first year	OIL3 <sub>γ</sub> OIL7	Table II.3. of Ref. [1]
of food, milk or drinking water	<b>GC(Ingestion, <math>h_{fetus,ing}</math>, 9mo) = 0.01 Sv<sup>g</sup></b> committed <sup>h</sup> equivalent dose to the fetus <sup>f</sup> from ingestion of food, milk and drinking water	$\begin{array}{c} \text{OIL3}_{\gamma} \\ \text{OIL7} \end{array}$	Table II.3. of Ref. [1]

#### TABLE 2. GENERIC CRITERIA USED AS A BASIS FOR THE OILS

<sup>a</sup> Response actions are implemented based on projected doses. Received doses are used to identify those warranting medical actions to detect and effectively treat radiation induced health effects.

during the full period of in utero development

<sup>b</sup> The representative person is described in Section 3.4.

<sup>c</sup> The total effective dose includes the effective dose from external exposure and the committed effective dose from intake of radioactive material during the exposure period in accordance with Table II.2 of Ref. [1].

<sup>d</sup> Effective dose alone cannot be used to ensure that the doses to the specific organ may not exceed the threshold for severe deterministic effects resulting from intake (inhalation or ingestion) or radioactive material on the skin. However, keeping the projected equivalent dose to the fetus below 100 mSv for the exposure scenarios of interest will ensure that the RBE weighted dose from intake for any organ or tissue (including the fetus and the skin) will not exceed the generic criteria for severe deterministic effects, as listed in Table II.1 of Ref. [1].

<sup>e</sup> The total equivalent dose to the fetus includes: (a) the maximum committed equivalent dose to any organ from intake to the fetus for different chemical compounds and time relative to conception; and (b) the equivalent dose to the fetus from external exposure during the exposure period, in accordance with Table II.2 of Ref. [1].

<sup>f</sup> In this publication the term 'fetus' encompasses both the embryo and the fetus.

<sup>g</sup> For the notation of the dose, a lower case letter is used (i.e. e, h or ad) to indicate that only a single exposure pathway is considered, as opposed to the total dose from all relevant exposure pathways, for which an upper case letter is used (i.e. E, H or AD).

<sup>h</sup> For all committed doses addressed in this publication, the integration time given in the respective references is used, i.e. typically 50 a for the adult, 70 a for the infant and the period of in utero development for the fetus.

<sup>1</sup> The generic criterion of 50 mSv committed equivalent dose to the thyroid given in Table II.2 of Ref. [1] was not used because it is intended for implementation of ITB and not for the urgent identification of those that may need medical follow-up. The criterion of 100 mSv committed equivalent dose to the thyroid for medical follow-up was determined based on consideration of: (a) the equivalent dose to the fetus warranting medical follow-up as given in Ref. [1] (i.e. 100 mSv); (b) footnote e; (c) the controlling organ dose to the fetus for intake of iodine being the thyroid [39]; and (d) the assumption that the equivalent dose to the pregnant woman's thyroid is approximately equal to the equivalent dose to the fetal thyroid [40].

#### 3.3. RADIONUCLIDE MIX

Knowing the radionuclide mix is necessary to perform the dose calculations, determine the response of the monitoring instruments and select marker radionuclides. Releases of radioactive material from an NPP are composed of a large number of radionuclides. All the radionuclide mixes expected to be released which may be significant contributors to the dose of the public or to the instrument response need to be considered. Radionuclides expected not to be deposited in significant quantities to cause health effects off the site or not to have a significant impact on the OILs provided in this publication, such as noble gases, have not been included neither here nor in the attached spreadsheets. No exposure of the public is expected to occur before 30 minutes after reactor shutdown [41], eliminating the need to consider a broad range of short-lived radionuclides in the fuel that are of no consequence in a release of radioactive material.

Many factors influence the mix released from an NPP, including the amount of the radionuclides in the fuel, the type of damage to the reactor fuel, the time of the release and the conditions in the plant during the release. At the time of the emergency, it will be very difficult to associate the mix of radioactive material deposited at a given location with a specific fuel damage state. For example, during the early stage of the accident at the Fukushima Daiichi nuclear power plant, it was not clear whether the spent fuel was contributing to the release (as described in Technical Volume 2 of Ref. [8]). In order to ensure that the default OIL values provide adequate protection, the radionuclide mixes representative of the whole range of postulated or observed fuel damage scenarios and releases to the atmosphere resulting from severe conditions at an LWR or its spent fuel have been considered, including older sources of information such as Ref. [42] (i.e. postulated core to containment mixes), which seem to fit the releases of radioactive material from the Fukushima Daiichi accident better than the postulated core to atmosphere estimates. The considered mixes are described in Table 3 and are grouped as follows:

- **Postulated core to containment release scenarios:** The fuel undergoes different phases of damage resulting in the release of radionuclides from the core to the containment (Mix 1-11):
  - Gap release phase: The fuel cladding (fuel pins) holding the fuel pellets fails and releases gaseous fission products such as noble gases, iodine and caesium located in the gap between the fuel pellets.
  - Early in-vessel melt release phase: The fuel heats up resulting in melting and slumping of core materials to the bottom of the reactor vessel. During this phase, virtually all noble gases and significant fractions of the volatile radionuclides such as iodine and caesium are released from the fuel.
  - Ex-vessel melt release and late in-vessel release phases: The molten core debris (including reactor fuel) melts through the reactor vessel and falls onto the concrete structural materials of the cavity below the reactor vessel. As a result, the less volatile nuclides may be released due to the interaction with the concrete. It is also assumed that vaporization of radionuclides that have been deposited on surfaces within the reactor coolant system during the early in-vessel melt release phase and the ex-vessel melt release phase contribute to the release during this phase. Therefore, the ex-vessel melt release and late in-vessel release are added together.
- **Postulated core to atmosphere release scenarios:** The fuel undergoes different phases of damage resulting in the release of radionuclides from the core into the containment, which are then affected by conditions within the plant (e.g. by sprays or plate-out) (Mix 12-15).
- **Postulated spent fuel release scenario:** Spent fuel not being cooled, heating up and resulting in a self-sustained zirconium oxidation reaction (Mix 16).
- Actual emergencies: Estimated mix from the Chernobyl accident (Mix 17), estimated mix for the release of radioactive material from the Fukushima Daiichi accident (Mix 18) and mix considering the measured deposition resulting from the Fukushima Daiichi accident (Mix 19).

Mix	Description	Reactor type	Fuel inventory
	Postulated core to containment release scenarios		
1	Estimated release fractions from the core during the <b>gap release</b> phase of fuel damage for a pressurized water reactor (PWR) or boiling water reactor (BWR) with standard fuel, obtained from Tables 3.12 and 3.13 of Ref. [42].	PWR/ BWR	
2	Estimated release fractions from the core during the <b>early in-vessel melt release</b> phase of fuel damage for a BWR with standard fuel, obtained from Table 3.12 of Ref. [42].		
3	Sum of the estimated release fractions from the core during the <b>ex-vessel melt</b> <b>and late in-vessel release</b> phases of fuel damage for a BWR with standard fuel, obtained from Table 3.12 of Ref. [42]. The release fractions of these two phases were added together, because it is likely that both would be occurring at the time of a release from the containment to the atmosphere.	BWR	standard
4	Estimated release fractions from the core during the <b>early in-vessel melt release</b> phase of fuel damage for a PWR with standard fuel, obtained from Table 3.13 of Ref. [42].		
5	Sum of the estimated release fractions from the core during the <b>ex-vessel melt</b> <b>and late in-vessel release</b> phases of fuel damage for a PWR with standard fuel, obtained from Table 3.13 of Ref. [42]. The release fractions of these two phases were added together, because it is likely that both would be occurring at the time of a release from the containment to the atmosphere.	PWR	
6	Estimated release fractions from the core during the <b>gap release</b> phase of fuel damage for a BWR with high burnup fuel, obtained from Table 12 of Ref. [43].		
7	Estimated release fraction from the core during the <b>early in-vessel melt</b> release phase of fuel damage for a BWR with high burnup fuel, obtained from Table 12 of Ref. [43].	BWR	
8	Sum of the estimated release fractions from the core during the <b>ex-vessel melt</b> <b>and late in-vessel release</b> phases of fuel damage for a BWR with high burnup fuel, obtained from Table 12 of Ref. [43]. The release fractions of these two phases were added together, because it is likely that both would be occurring at the time of a release from the containment to the atmosphere.		high
9	Estimated release fractions from the core during the <b>gap release</b> phase of fuel damage for a PWR with high burnup fuel, obtained from Table 13 of Ref. [43].		burnup
10	Estimated release fractions from the core during the <b>early in-vessel melt</b> release phase of fuel damage for a PWR with high burnup fuel, obtained from Table 13 of Ref. [43].	PWR	
11	Sum of the estimated release fractions from the core during the <b>ex-vessel melt</b> <b>and late in-vessel release</b> phases of fuel damage for a PWR with high burnup fuel, obtained from Table 13 of Ref. [43]. The release fractions of these two phases were added together, because it is likely that both would be occurring at the time of a release from the containment to the atmosphere.		

# TABLE 3. DESCRIPTION OF THE CONSIDERED RADIONUCLIDE MIXES

Mix	Description	Reactor type	Fuel inventory
	Postulated core to atmosphere release scenarios		v
12	Estimated atmospheric release fractions from a BWR following core melt and <b>melt through of the reactor vessel</b> resulting from long term station blackout. These are the total release fractions projected at about <b>22 hours</b> following start of the accident and about 2 hours after the containment fails resulting in the first phase of the major release. The values are obtained from Fig. 5-11 of Vol. 1 of Ref. [44] Ref. [44] does not provide a release fraction applicable to Sr. Therefore, the Sr release fraction used is based on the ratio of the Sr and Cs fractions for a BWR core melt from Mix 8 (Sr/Cs $\approx$ 0.5), and, consequently, the Sr release fraction was set at 0.5 times the release fraction for Cs.		
13	Estimated atmospheric release fractions from a BWR following core melt and <b>melt through of the reactor vessel</b> resulting from long term station blackout. These are total release fractions projected at about <b>24 hours</b> following the start of the accident and include the releases projected to occur during the second phase of the major release, which occurs when volatile radionuclides deposited on surfaces in the plant vaporize and are released. The values are obtained from Fig. 5-11 of Vol. 1 of Ref. [44].	BWR	
	release fraction used is based on the ratio of the Sr and Cs fractions for a BWR core melt from Mix 8 (Sr/Cs $\approx$ 0.5), and, consequently, the Sr release fraction was set at 0.5 times the release fraction for Cs.		standard and high burnup <sup>a</sup>
14	Estimated atmospheric release fractions from a PWR following core melt and <b>melt through of the reactor vessel</b> resulting from long-term station blackout. These are total release fractions projected to occur by normal leakage at about <b>1.5 days</b> after the start of the accident. The values are obtained from Fig. 5-9 of Vol. 2 of Ref. [44].		
	Ref. [44] does not provide a release fraction applicable to Sr. Therefore, the Sr release fraction used is based on the ratio of the Sr and Cs fractions for a PWR core melt from Mix 11 (Sr/Cs $\approx$ 0.04), and, consequently, the Sr release fraction was set 0.04 times the release fraction for Cs.	DWD	
	Estimated atmospheric release fractions from a PWR following core melt and <b>melt through of the reactor vessel</b> resulting from long term station blackout. These are the total release fractions projected at about <b>4 days</b> after the start of the accident and following containment failure. The values are obtained from Fig. 5-9 of Vol. 2 of Ref. [44].	- PWK	
	Ref. [44] does not provide a release fraction applicable to Sr. Therefore, the Sr release fraction used is based on the ratio of the Sr and Cs fractions for a PWR core melt from Mix 11 (Sr/Cs $\approx$ 0.04), and, consequently, the Sr release fraction was set at 0.04 times the release fraction for Cs.		
	Postulated spent fuel release scenario		
16	Estimated release fractions from spent fuel. The radionuclide mix is assumed to be similar to the releases of the volatile fission products during core melt, except that it may result in higher releases of ruthenium during severe overheating of spent fuel as postulated in Ref. [45]. The mix was based on Mix 7 but with a Ru release faction that is the same as for Cs to account for the much higher releases of Ru. Only spent fuel that is sufficiently heated (either by its own decay heat or another heat source) to reach zirconium ignition temperatures is expected to result in a significant release of radioactive material warranting response actions. This is typically spent fuel in the spent fuel pool of an NPP	-	standard and high burnup <sup>a</sup>

Mix	Description	Reactor type	Fuel inventory
	Actual emergencies		
17	Estimated release fractions from the Chernobyl accident based on the total release to the atmosphere estimated in Table 4 of Ref. [46]. For the radionuclides not mentioned in Ref. [46] the release fraction of the chemically similar radionuclides is used following Table 3.8 of Ref. [42].	-	standard and high burnup <sup>a</sup>
18	Estimated release amounts from the Fukushima Daiichi accident based on the average estimates of the total release to the atmosphere of Table 4.1–2 of Technical Volume 4 of Ref. [8]. The release amounts were normalized to the time of shutdown of the reactor (assuming 40 hours, since at this time the iodine ratios are similar). For the radionuclides not mentioned in Ref. [8], if available, the release fraction of an isotope is used, and, if not available, the release fractions for chemically similar radionuclide(s) are used following Table 3.8 of Ref. [42].	-	
19	Mix 19 is used to examine the measured values of Sr-90, Te-129m, Te-132, Cs-134 and I-131 relative to Cs-137 on the ground following the atmospheric releases from the Fukushima Daiichi nuclear power plant, as provided in Refs [47, 48, 49, 50, 51]. These values, normalized to March 11, were: (a) Sr-90/Cs-137, in most cases at around 0.001 [47]; (b) Te-129m/Cs-137, in most cases around 0.5 [48, 49]; (c) Te-132/Cs-137, in most cases between 15–20 [50, 51]; (d) Cs-134/Cs-137, in most cases around 1.0 [47, 49]; and (e) I-131/Cs-137, in most cases between 15–25 [49, 50, 51]. Mix 19 was developed based on Mix 18, by modifying the release amounts of Sr-90, Te-129m, Te-132, Cs-134 and I-131 relative to Cs-137 according to the ratios given above. <sup>c</sup>	-	NA <sup>b</sup>

<sup>&</sup>lt;sup>a</sup> Both inventories (corresponding to standard and high burnup fuel) were considered in the calculations of the OILs. The results for both inventories only differ slightly and do not affect the final default OIL values; only the results for the standard fuel inventory are provided in Section 3.8. Providing both sets of results was considered to be of no added value.

<sup>b</sup> Not applicable. Table 4.1–2 of the Technical Volume 4 of Ref. [8] provides the estimated release in Bq and not the release fraction, making it unnecessary to choose between standard and high burnup fuel.

<sup>&</sup>lt;sup>c</sup> The uncertainty for the estimates is greater for radionuclides other than iodine and caesium, reflecting the lack of direct measurements of these radionuclides immediately after the accident. Mix 19 needs to be treated with care and only considered as an approximation within orders of magnitude.

OIL values are calculated for each of the 19 radionuclide mixes described in Table 3 by using  $RA_i(t,mix)$  [unitless], which is the relative activity of radionuclide i at time t after reactor shutdown (not the absolute activity in Bq). It is determined by Eq. (1):

$$RA_{i}(t, mix) = \frac{A_{i}(t, mix)}{\sum_{i=1}^{n} A_{i}(t, mix)}$$
(1)

where:

- n [unitless] is the number of considered radionuclides.
- t [s] is the time after shutdown of the reactor (i.e. after the fuel was last irradiated in a reactor).
- 'mix' refers to the mix number as provided in Table 3, Table 7, Table 8 and Table 9.
- A<sub>i</sub>(t,mix) [Bq] is the activity of radionuclide i at time t after reactor shutdown, for a specific radionuclide mix. It is determined by combining release fractions with core inventories for two operating conditions (standard and high burnup), as described in Eq. (2):

$$A_{i}(t, mix) = I_{fuel-type,i}(t) \times RF_{i}(mix)$$
<sup>(2)</sup>

where:

 $\circ$  I<sub>fuel-type,i</sub>(t) [Bq] is the inventory of radionuclide i in the fuel at time t after reactor shutdown, which is calculated by Eq. (3):

$$I_{\text{fuel-type},i}(t) = I_{\text{fuel-type},i}(t_0) \times e^{-\lambda_i \times (t-t_0)}$$
(3)

where:

Ifuel-type,i(t<sub>0</sub>) [Bq] is the inventory of radionuclide i in the fuel at time t<sub>0</sub> = 30 min = 1800 s after reactor shutdown. As mentioned before, no exposure of the public is expected to occur before 30 minutes after reactor shutdown [41]. The values are given in Table 4 for a 3000 MW(th) power plant and are provided for two different fuel types: (a) standard fuel and (b) a high burnup fuel. Also mixed oxide fuel (MOX) was considered, but it did not have a significant impact for those radionuclides important in terms of health effects [43]; thus, it is not addressed in further detail. For those radionuclides marked with a '+' in Table 4, the progenies are assumed to be in equilibrium with the parent (Table 5 provides the ratio between the activity of the progeny and the parent radionuclide). Therefore, these particular radionuclides do not need to be considered separately.

Both inventories (corresponding to standard and high burnup fuel) were considered in the calculations of the OILs. The results for both inventories only differ slightly and do not affect the final default OIL values.

- $\lambda_i [s^{-1}]$  is the decay constant for radionuclide i. The values are given in Table 6.
- $\circ$  RF<sub>i</sub>(mix) [unitless] is the release fraction of radionuclide i from the fuel for a specific radionuclide mix. The values are given in Table 7, Table 8 and Table 9.

Dadianualida	$I_{\text{fuel-type},i}(t_0)$ [Bq]				
Kadionuciide –	Standard fuel <sup>a</sup>	High burnup fuel <sup>b,c</sup>			
Rb-86	9.6E+14	3.7E+15			
Sr-89	3.5E+18	3.2E+18			
Sr-90+	1.4E+17	2.5E+17			
Sr-91	4.1E+18	3.9E+18			
Y-91	4.4E+18	4.1E+18			
Zr-95+	5.6E+18	5.2E+18			
Zr-97+	5.6E+18	5.1E+18			
Mo-99+	5.9E+18	5.5E+18			
Ru-103+	4.1E+18	4.1E+18			
Ru-105	2.7E+18	2.4E+18			
Ru-106+	9.3E+17	1.1E+18			
Rh-105	1.8E+18	2.3E+18			
Te-127m+	4.1E+16	3.7E+16			
Te-127	2.2E+17	2.2E+17			
Te-129m+	2.0E+17	1.3E+17			
Te-131m	4.8E+17	5.1E+17			
Te-132+	4.4E+18	4.1E+18			
I-131	3.2E+18	2.9E+18			
I-133	6.3E+18	6.0E+18			
I-134	7.0E+18	4.7E+18			
I-135	5.6E+18	5.5E+18			
Cs-134	2.8E+17	3.1E+17			
Cs-136	1.1E+17	1.2E+17			
Cs-137+	1.7E+17	3.2E+17			
Ba-140+	5.9E+18	5.3E+18			
Ce-141	5.6E+18	5.0E+18			
Ce-143	4.8E+18	4.8E+18			
Ce-144+	3.2E+18	3.6E+18			
Pr-143	4.8E+18	4.7E+18			
Nd-147	2.2E+18	2.0E+18			
Np-239	5.9E+19	4.8E+19			
Pu-238	2.1E+15 5.8E+15				
Pu-239	7.8E+14	1.2E+15			
Pu-240	7.8E+14	9.6E+14			
Pu-241	1.3E+17	3.3E+17			
Am-241	6.3E+13	4.5E+14			
Cm-242	1.9E+16	8.2E+16			
Cm-244	8.5E+14	4.0E+15			

TABLE 4. INVENTORY OF RADIONUCLIDE i IN THE FUEL FOR A 3000 MW(th) LWR

<sup>a</sup>  $I_{\text{fuel-type,i}}(t_0)$  for the standard fuel is obtained from Table ID of Ref. [5], which is based on Table VI-3-1 of Ref. [41]. <sup>b</sup>  $I_{\text{fuel-type,i}}(t_0)$  for the high burnup fuel is obtained from the inventory in Tables A-3 to A-10 of Vol. 1 of Ref. [44],

 $f_{\text{fuel-type,i}}(0)$  for the high burnup rule is obtained from the inventory in Fables A-5 to A-10 of Vol. 1 of Ker. [44], which was normalized to time  $t_0 = 30$  min after reactor shutdown and to 3000 MW(th) to allow for a direct comparison. This was done in consideration of 3000 MW(th) being approximately equivalent to 1000 MW(e).

<sup>c</sup> The high burnup inventory shows reasonable agreement with the inventory for Unit 2 of the Fukushima Daiichi NPP at the time of the accident, as provided in Technical Volume 1 of Ref. [8].

TABLE 5. PROGENIES CONSIDERED FOR THE RADIONUCLIDES MARKED WITH '+' (*The ratio between the activity of the progeny and the parent radionuclide is given in the parentheses and is obtained from Table 11 of Ref. [2].*)

Parent radionuclide	Progenies considered to be in equilibrium with the parent radionuclide in the fuel
Sr-90	Y-90 (1.0)
Zr-95	Nb-95 (2.2)
Zr-97	Nb-97m (0.95), Nb-97 (1.0)
Mo-99	Tc-99m (0.96)
Ru-103	Rh-103m (1.0)
Ru-106	Rh-106 (1.0)
Te-127m <sup>a</sup>	Te-127 (1.0)
Te-129m	Te-129 (0.65)
Te-132	I-132 (1.0)
Cs-137	Ba-137m (1.0)
Ba-140	La-140 (1.2)
Ce-144	Pr-144m (0.018), Pr-144 (1.0)

<sup>a</sup> Te-127 is considered as being in equilibrium with Te-127m after its release. Te-127 is considered separately in the fuel inventory because it is not in equilibrium with the Te-127m in the fuel.

# TABLE 6. HALF-LIFE AND DECAY CONSTANT

<b>D</b> I' I' I	Half-life	$e(T_{1/2})^{a}$	Decay constant $(\lambda_i)^b$
Radionuclide -	Value	Unit	- [s <sup>-1</sup> ]
Rb-86	18.642	[d]	4.3E-07
Sr-89	50.53	[d]	1.6E-07
Sr-90	28.79	[a]	7.6E-10
Sr-91	9.63	[h]	2.0E-05
Y-91	58.51	[d]	1.4E-07
Zr-95	64.032	[d]	1.3E-07
Zr-97	16.744	[h]	1.1E-05
Mo-99	65.94	[h]	2.9E-06
Ru-103	39.26	[d]	2.0E-07
Ru-105	4.44	[h]	4.3E-05
Ru-106	373.59	[d]	2.1E-08
Rh-105	35.36	[h]	5.4E-06
Te-127m	109	[d]	7.4E-08
Te-127	9.35	[h]	2.1E-05
Te-129m	33.6	[d]	2.4E-07
Te-131m	30	[h]	6.4E-06
Te-132	3.204	[d]	2.5E-06
I-131	8.0207	[d]	1.0E-06
I-133	20.8	[h]	9.3E-06
I-134	52.5	[m]	2.2E-04
I-135	6.57	[h]	2.9E-05
Cs-134	2.0648	[a]	1.1E-08
Cs-136	13.16	[d]	6.1E-07
Cs-137	30.1671	[a]	7.3E-10
Ba-140	12.752	[d]	6.3E-07
Ce-141	32.508	[d]]	2.5E-07
Ce-143	33.039	[h]	5.8E-06
Ce-144	284.91	[d]	2.8E-08
Pr-143	13.57	[d]	5.9E-07
Nd-147	10.98	[d]	7.3E-07
Np-239	2.3565	[d]	3.4E-06
Pu-238	87.7	[a]	2.5E-10
Pu-239	24110	[a]	9.1E-13
Pu-240	6564	[a]	3.3E-12
Pu-241	14.35	[a]	1.5E-09
Am-241	432.2	[a]	5.1E-11
Cm-242	162.8	[d]	4.9E-08
Cm-244	18.1	[a]	1.2E-09

<sup>a</sup> The values are obtained from Table A.1. of Ref. [52]. <sup>b</sup>  $\lambda_i$  is calculated based on the half-life:  $\lambda_i = ln(2) / T_{1/2.}$ 

TABLE 7. RELEASE FRACTION OF RADIONUCLIDE i FROM THE FUEL FOR THE CONSIDERED RADIONUCLIDE MIXES (MIXES 1–7)

(The basis for these values are described in Table 3.)

(0.00E+00 means that only a negligible release is predicted to occur.)

Dadionualida	RF <sub>i</sub> (mix1)	RF <sub>i</sub> (mix2)	RF <sub>i</sub> (mix3)	RF <sub>i</sub> (mix4)	RF <sub>i</sub> (mix5)	RF <sub>i</sub> (mix6)	RF <sub>i</sub> (mix7)
Kaulonuchue	[unitless]						
Rb-86	5.0E-02	2.0E-01	3.6E-01	2.5E-01	4.5E-01	2.0E-03	1.3E-01
Sr-89	0.0E+00	2.0E-02	1.0E-01	2.0E-02	1.0E-01	0.0E+00	5.0E-03
Sr-90+	0.0E+00	2.0E-02	1.0E-01	2.0E-02	1.0E-01	0.0E+00	5.0E-03
Sr-91	0.0E+00	2.0E-02	1.0E-01	2.0E-02	1.0E-01	0.0E+00	5.0E-03
Y-91	0.0E+00	2.0E-04	5.0E-03	2.0E-04	5.0E-03	0.0E+00	1.4E-07
Zr-95+	0.0E+00	2.0E-04	5.0E-03	2.0E-04	5.0E-03	0.0E+00	1.3E-07
Zr-97+	0.0E+00	2.0E-04	5.0E-03	2.0E-04	5.0E-03	0.0E+00	1.3E-07
Mo-99+	0.0E+00	2.5E-03	2.5E-03	2.5E-03	2.5E-03	0.0E+00	2.0E-02
Ru-103+	0.0E+00	2.5E-03	2.5E-03	2.5E-03	2.5E-03	0.0E+00	2.7E-03
Ru-105	0.0E+00	2.5E-03	2.5E-03	2.5E-03	2.5E-03	0.0E+00	2.7E-03
Ru-106+	0.0E+00	2.5E-03	2.5E-03	2.5E-03	2.5E-03	0.0E+00	2.7E-03
Rh-105	0.0E+00	2.5E-03	2.5E-03	2.5E-03	2.5E-03	0.0E+00	2.7E-03
Te-127m+	0.0E+00	5.0E-02	2.6E-01	5.0E-02	2.6E-01	2.0E-03	3.9E-01
Te-127	0.0E+00	5.0E-02	2.6E-01	5.0E-02	2.6E-01	2.0E-03	3.9E-01
Te-129m+	0.0E+00	5.0E-02	2.6E-01	5.0E-02	2.6E-01	2.0E-03	3.9E-01
Te-131m	0.0E+00	5.0E-02	2.6E-01	5.0E-02	2.6E-01	2.0E-03	3.9E-01
Te-132+	0.0E+00	5.0E-02	2.6E-01	5.0E-02	2.6E-01	2.0E-03	3.9E-01
I-131	5.0E-02	2.5E-01	3.1E-01	3.5E-01	3.5E-01	2.0E-03	4.7E-01
I-133	5.0E-02	2.5E-01	3.1E-01	3.5E-01	3.5E-01	2.0E-03	4.7E-01
I-134	5.0E-02	2.5E-01	3.1E-01	3.5E-01	3.5E-01	2.0E-03	4.7E-01
I-135	5.0E-02	2.5E-01	3.1E-01	3.5E-01	3.5E-01	2.0E-03	4.7E-01
Cs-134	5.0E-02	2.0E-01	3.6E-01	2.5E-01	4.5E-01	2.0E-03	1.3E-01
Cs-136	5.0E-02	2.0E-01	3.6E-01	2.5E-01	4.5E-01	2.0E-03	1.3E-01
Cs-137+	5.0E-02	2.0E-01	3.6E-01	2.5E-01	4.5E-01	2.0E-03	1.3E-01
Ba-140+	0.0E+00	2.0E-02	1.0E-01	2.0E-02	1.0E-01	0.0E+00	5.0E-03
Ce-141	0.0E+00	5.0E-04	5.0E-03	5.0E-04	5.0E-03	0.0E+00	1.3E-07
Ce-143	0.0E+00	5.0E-04	5.0E-03	5.0E-04	5.0E-03	0.0E+00	1.3E-07
Ce-144+	0.0E+00	5.0E-04	5.0E-03	5.0E-04	5.0E-03	0.0E+00	1.3E-07
Pr-143	0.0E+00	2.0E-04	5.0E-03	2.0E-04	5.0E-03	0.0E+00	1.4E-07
Nd-147	0.0E+00	2.0E-04	5.0E-03	2.0E-04	5.0E-03	0.0E+00	1.4E-07
Np-239	0.0E+00	5.0E-04	5.0E-03	5.0E-04	5.0E-03	0.0E+00	1.3E-07
Pu-238	0.0E+00	5.0E-04	5.0E-03	5.0E-04	5.0E-03	0.0E+00	1.3E-07
Pu-239	0.0E+00	5.0E-04	5.0E-03	5.0E-04	5.0E-03	0.0E+00	1.3E-07
Pu-240	0.0E+00	5.0E-04	5.0E-03	5.0E-04	5.0E-03	0.0E+00	1.3E-07
Pu-241	0.0E+00	5.0E-04	5.0E-03	5.0E-04	5.0E-03	0.0E+00	1.3E-07
Am-241	0.0E+00	2.0E-04	5.0E-03	2.0E-04	5.0E-03	0.0E+00	1.4E-07
Cm-242	0.0E+00	2.0E-04	5.0E-03	2.0E-04	5.0E-03	0.0E+00	1.4E-07
Cm-244	0.0E+00	2.0E-04	5.0E-03	2.0E-04	5.0E-03	0.0E+00	1.4E-07

TABLE 8. RELEASE FRACTION OF RADIONUCLIDE i FROM THE FUEL FOR THE CONSIDERED RADIONUCLIDE MIXES (MIXES 8–13)

(The basis for these values are described in Table 3.)

(0.00E+00 means that only a negligible release is predicted to occur.)

Dedlemarki	RF <sub>i</sub> (mix8)	RF <sub>i</sub> (mix9)	RF <sub>i</sub> (mix10)	RF <sub>i</sub> (mix11)	RF <sub>i</sub> (mix12)	RF <sub>i</sub> (mix13)
Radionuclide	[unitless]	[unitless]	[unitless]	[unitless]	[unitless]	[unitless]
Rb-86	6.0E-02	3.0E-03	2.3E-01	8.0E-02	3.0E-03	3.8E-03
Sr-89	3.4E-02	6.0E-04	4.0E-03	3.0E-03	1.5E-03	1.9E-03
Sr-90+	3.4E-02	6.0E-04	4.0E-03	3.0E-03	1.5E-03	1.9E-03
Sr-91	3.4E-02	6.0E-04	4.0E-03	3.0E-03	1.5E-03	1.9E-03
Y-91	5.0E-05	0.0E+00	1.5E-07	1.3E-05	5.0E-06	1.0E-05
Zr-95+	2.1E-03	0.0E+00	1.5E-07	2.4E-04	6.0E-05	7.5E-05
Zr-97+	2.1E-03	0.0E+00	1.5E-07	2.4E-04	6.0E-05	7.5E-05
Mo-99+	8.5E-03	0.0E+00	8.0E-02	4.0E-02	8.0E-04	8.5E-04
Ru-103+	2.6E-03	0.0E+00	6.0E-03	2.5E-03	2.0E-06	3.0E-06
Ru-105	2.6E-03	0.0E+00	6.0E-03	2.5E-03	2.0E-06	3.0E-06
Ru-106+	2.6E-03	0.0E+00	6.0E-03	2.5E-03	2.0E-06	3.0E-06
Rh-105	2.6E-03	0.0E+00	6.0E-03	2.5E-03	2.0E-06	3.0E-06
Te-127m+	3.3E-01	4.0E-03	3.0E-01	1.0E-01	2.2E-03	1.0E-02
Te-127	3.3E-01	4.0E-03	3.0E-01	1.0E-01	2.2E-03	1.0E-02
Te-129m+	3.3E-01	4.0E-03	3.0E-01	1.0E-01	2.2E-03	1.0E-02
Te-131m	3.3E-01	4.0E-03	3.0E-01	1.0E-01	2.2E-03	1.0E-02
Te-132+	3.3E-01	4.0E-03	3.0E-01	1.0E-01	2.2E-03	1.0E-02
I-131	4.0E-01	4.0E-03	3.7E-01	2.2E-01	4.0E-03	1.0E-02
I-133	4.0E-01	4.0E-03	3.7E-01	2.2E-01	4.0E-03	1.0E-02
I-134	4.0E-01	4.0E-03	3.7E-01	2.2E-01	4.0E-03	1.0E-02
I-135	4.0E-01	4.0E-03	3.7E-01	2.2E-01	4.0E-03	1.0E-02
Cs-134	6.0E-02	3.0E-03	2.3E-01	8.0E-02	3.0E-03	3.8E-03
Cs-136	6.0E-02	3.0E-03	2.3E-01	8.0E-02	3.0E-03	3.8E-03
Cs-137+	6.0E-02	3.0E-03	2.3E-01	8.0E-02	3.0E-03	3.8E-03
Ba-140+	3.4E-02	6.0E-04	4.0E-03	3.0E-03	4.5E-03	5.0E-03
Ce-141	2.1E-03	0.0E+00	1.5E-07	2.4E-04	6.0E-05	7.5E-05
Ce-143	2.1E-03	0.0E+00	1.5E-07	2.4E-04	6.0E-05	7.5E-05
Ce-144+	2.1E-03	0.0E+00	1.5E-07	2.4E-04	6.0E-05	7.5E-05
Pr-143	5.0E-05	0.0E+00	1.5E-07	1.3E-05	5.0E-06	1.0E-05
Nd-147	5.0E-05	0.0E+00	1.5E-07	1.3E-05	5.0E-06	1.0E-05
Np-239	2.1E-03	0.0E+00	1.5E-07	2.4E-04	6.0E-05	7.5E-05
Pu-238	2.1E-03	0.0E+00	1.5E-07	2.4E-04	6.0E-05	7.5E-05
Pu-239	2.1E-03	0.0E+00	1.5E-07	2.4E-04	6.0E-05	7.5E-05
Pu-240	2.1E-03	0.0E+00	1.5E-07	2.4E-04	6.0E-05	7.5E-05
Pu-241	2.1E-03	0.0E+00	1.5E-07	2.4E-04	6.0E-05	7.5E-05
Am-241	5.0E-05	0.0E+00	1.5E-07	1.3E-05	5.0E-06	1.0E-05
Cm-242	5.0E-05	0.0E+00	1.5E-07	1.3E-05	5.0E-06	1.0E-05
Cm-244	5.0E-05	0.0E+00	1.5E-07	1.3E-05	5.0E-06	1.0E-05

TABLE 9. RELEASE FRACTION OF RADIONUCLIDE i FROM THE FUEL FOR THE CONSIDERED RADIONUCLIDE MIXES (MIX 14 TO 19)

(*The basis for these values are described in Table 3.*)

Dadianualida	RF <sub>i</sub> (mix14)	RF <sub>i</sub> (mix15)	RF <sub>i</sub> (mix16)	RF <sub>i</sub> (mix17)	RF <sub>i</sub> (mix18) <sup>a</sup>	RF <sub>i</sub> (mix19) <sup>a</sup>
Kaulonuchue	[unitless]	[unitless]	[unitless]	[unitless]	[unitless]	[unitless]
Rb-86	4.0E-05	7.5E-04	1.3E-01	3.3E-01	5.7E+00	5.7E+00
Sr-89	1.6E-06	3.0E-05	5.0E-03	5.0E-02	2.1E-03	2.1E-03
Sr-90+	1.6E-06	3.0E-05	5.0E-03	5.0E-02	2.8E-04	5.3E-05
Sr-91	1.6E-06	3.0E-05	5.0E-03	5.0E-02	2.9E-03	2.9E-03
Y-91	0.0E+00	0.0E+00	1.4E-07	3.5E-02	4.2E-06	4.2E-06
Zr-95+	1.0E-06	2.0E-05	1.3E-07	3.5E-02	3.3E-06	3.3E-06
Zr-97+	1.0E-06	2.0E-05	1.3E-07	3.5E-02	3.4E-06	3.4E-06
Mo-99+	9.0E-06	1.0E-04	2.0E-02	3.5E-02	2.4E-11	2.4E-11
Ru-103+	0.0E+00	1.5E-05	1.3E-01	3.5E-02	9.8E-09	9.8E-09
Ru-105	0.0E+00	1.5E-05	1.3E-01	3.5E-02	8.9E-09	8.9E-09
Ru-106+	0.0E+00	1.5E-05	1.3E-01	3.5E-02	1.9E-09	1.9E-09
Rh-105	0.0E+00	1.5E-05	2.7E-03	3.5E-02	4.6E-09	4.6E-09
Te-127m+	4.0E-05	2.0E-02	3.9E-01	4.3E-01	2.2E-01	2.2E-01
Te-127	4.0E-05	2.0E-02	3.9E-01	4.3E-01	3.6E-02	3.6E-02
Te-129m+	4.0E-05	2.0E-02	3.9E-01	4.3E-01	6.1E-02	5.1E-02
Te-131m	4.0E-05	2.0E-02	3.9E-01	4.3E-01	1.6E-02	1.6E-02
Te-132+	4.0E-05	2.0E-02	3.9E-01	4.3E-01	2.8E-02	5.7E-02
I-131	6.0E-05	5.0E-03	4.7E-01	5.5E-01	1.0E-01	9.4E-02
I-133	6.0E-05	5.0E-03	4.7E-01	5.5E-01	9.5E-02	9.5E-02
I-134	6.0E-05	5.0E-03	4.7E-01	5.5E-01	9.2E-02	9.2E-02
I-135	6.0E-05	5.0E-03	4.7E-01	5.5E-01	7.8E-02	7.8E-02
Cs-134	4.0E-05	7.5E-04	1.3E-01	3.3E-01	9.5E-02	4.4E-02
Cs-136	4.0E-05	7.5E-04	1.3E-01	3.3E-01	1.8E-01	1.8E-01
Cs-137+	4.0E-05	7.5E-04	1.3E-01	3.3E-01	4.2E-02	4.2E-02
Ba-140+	3.0E-06	7.5E-04	5.0E-03	5.0E-02	2.2E-03	2.2E-03
Ce-141	1.0E-06	2.0E-05	1.3E-07	3.5E-02	3.7E-06	3.7E-06
Ce-143	1.0E-06	2.0E-05	1.3E-07	3.5E-02	3.1E-06	3.1E-06
Ce-144+	1.0E-06	2.0E-05	1.3E-07	3.5E-02	3.1E-06	3.1E-06
Pr-143	0.0E+00	0.0E+00	1.4E-07	3.5E-02	3.7E-06	3.7E-06
Nd-147	0.0E+00	0.0E+00	1.4E-07	3.5E-02	8.8E-06	8.8E-06
Np-239	1.0E-06	2.0E-05	1.3E-07	3.5E-02	2.6E-06	2.6E-06
Pu-238	1.0E-06	2.0E-05	1.3E-07	3.5E-02	1.8E-06	1.8E-06
Pu-239	1.0E-06	2.0E-05	1.3E-07	3.5E-02	1.5E-06	1.5E-06
Pu-240	1.0E-06	2.0E-05	1.3E-07	3.5E-02	1.9E-06	1.9E-06
Pu-241	1.0E-06	2.0E-05	1.3E-07	3.5E-02	1.8E-06	1.8E-06
Am-241	0.0E+00	0.0E+00	1.4E-07	3.5E-02	3.9E-02	3.9E-02
Cm-242	0.0E+00	0.0E+00	1.4E-07	3.5E-02	6.8E-07	6.8E-07
Cm-244	0.0E+00	0.0E+00	1.4E-07	3.5E-02	4.3E-03	4.3E-03

(0.00E+00 means that only a negligible release is predicted to occur.)

<sup>a</sup> The release fraction shown here is applicable for high burnup and is only provided for consistency. Technical Volume 4 of Ref. [8] provides the estimated release in Bq and not the release fraction, making it unnecessary to choose between standard and high burnup fuel.

#### 3.4. REPRESENTATIVE PERSON, EXPOSURE SCENARIOS AND PATHWAYS

All members of the public have been considered in the calculation of the OILs by taking response actions based on the dose projected or received by the representative person and its fetus, as recommended by the International Commission on Radiological Protection (ICRP) in Refs [38, 53], which gives reasonable assurance that the actions will also effectively protect any other member of the public. Using the representative person is a conservative approach: No member of the public is expected to receive a dose during an actual emergency close to that calculated for the representative person for the conditions described in the scenarios.

The representative person (which replaces the concept of 'critical group') does not represent any specific person from a particular age group; it is a theoretical construct defined by a combination of dosimetric and scenario factors (e.g. breathing and ingestion rates) that will result in the highest dose (effective or organ) reasonably expected to be received by any member of the public. It aggregates the dose models for the internal and external exposure of ICRP reference persons of different age groups, together with relevant exposure scenario, providing the highest dose estimate for particular pathways. Depending on the type of the exposure characteristic, e.g. effective dose or organ dose, the different sets of dose models have to be aggregated to assess the exposure of the representative person, as described in Table 10.

The scenarios used for the calculation of the OILs provided in this publication are described in Table 11. For a severe release of radioactive material from an LWR or its spent fuel the total effective dose to the representative person and the equivalent dose to its fetus<sup>31</sup> will be controlling for the 'ground', 'food pre-analysis', 'skin' and 'food post-analysis' scenarios. For the 'thyroid scenario', it will be the equivalent dose to the thyroid of the representative person. The exposure pathways for each exposure scenario and the associated parameters of the effective dose to the representative person and its fetus are described in Table 12 to Table 16. The figures shown in black represent the controlling member of the public for the corresponding exposure pathway and scenario, while the figures in grey indicate that the member of the public is not controlling for the exposure scenario (but was still considered).

For the 'ground', 'food pre-analysis', 'skin' and 'food post-analysis' scenarios, the exposure starts from deposition of radioactive material (on the ground or skin). The contribution from the plume (cloud shine and inhalation) is not considered in the OIL calculations, since the response actions to protect the public from these exposure pathways needs to be triggered by the emergency classification system to be effective and not by monitoring results (as discussed in Ref. [4]).

The 1-year-old infant, 10-year-old child and the adult ( > 17 a) were used to represent the three age categories recommended in Ref. [53]. In this publication, the term 'fetus' encompasses both the embryo and the fetus.

<sup>&</sup>lt;sup>31</sup> Keeping the equivalent dose to the fetus from intake below the generic criteria will ensure that the dose to other organs will also stay below the generic criteria. The equivalent dose to the fetus is defined as the highest equivalent dose to any organ in the fetus during the period of in utero development.

### TABLE 10. DOSIMETRIC CHARACTERISTICS OF THE REPRESENTATIVE PERSON FOR THE CONSIDERED EXPOSURE PATHWAYS

Dose to the representative person	Ingestion	Inadvertent ingestion	External exposure	Inhalation	Radioactive iodine in the thyroid
Effective dose	Infant or adult <sup>a</sup>	Infant	Infant	Adult	_
Equivalent dose to the fetus	Adult <sup>b</sup>	Adult <sup>b</sup>	Adult <sup>b</sup>	Adult <sup>b</sup>	_
Equivalent dose to the thyroid	Infant <sup>c</sup>	Infant <sup>c</sup>	_	Adult <sup>c</sup>	Infant
RBE weighted absorbed dose to the skin dermis	_	_	Age independent	_	_

<sup>a</sup> Depending on the radionuclide.

<sup>b</sup> Pregnant woman.
 <sup>c</sup> Not directly considered in the calculation of the OILs given in this publication.

# TABLE 11. DESCRIPTION OF THE EXPOSURE SCENARIOS CONSIDERED IN THE CALCULATION OF THE OILS

OII	Exposure scenarios					
UIL	Name	Summary	pathways			
$\begin{array}{c} OIL1_{\gamma}\\ OIL2_{\gamma} \end{array}$	Ground	It is assumed that all members of the public live normally for 7 days (for the calculation of $OIL1_{\gamma}$ ) and for 1 year (for the calculation of $OIL2_{\gamma}$ ) in an area affected by a release of radioactive material, carrying out normal activities (such as children playing on the ground and people being inside normal structures part of the time). However, it is assumed that the public is not consuming food, milk or drinking water from the affected area, because restrictions have been implemented on the basis of the emergency classification (as described in Ref. [4]).	Table 12			
OIL3 <sub>γ</sub>	Food pre- analysis	It is assumed that the public will consume local food and milk (e.g. from their garden or cow) that is directly affected by the deposition of radioactive material before entering the food distribution system and thus before it is sampled or controlled in any way. It is assumed that (a) there will be no reduction in the concentration of radioactive materials due to preparation of the food (e.g. peeling, washing) before consumption; (b) the affected food and milk is consumed over 1 year; and (c) 50% of the food and milk is affected.	Table 13			
$OIL4_{\gamma}$ $OIL4_{\beta}$	Skin	It is assumed that the person has radioactive material on the skin (including the face and hands), and that no actions are taken to remove the material from the skin (e.g. washing) or to reduce inadvertent ingestion (e.g. keeping the hand away from the mouth).	Table 14			
OIL7	Food post- analysis	It is assumed that the food, milk or drinking water in the food supply is affected, that samples are taken and that the activity concentrations of the marker radionuclides I-131 and Cs-137 in the food are known (e.g. by laboratory analysis). It is assumed that (a) 50% of the diet is affected; (b) it is consumed over 1 year; and (c) there is no reduction due to processing or preparation.	Table 15			
$OIL8_{\gamma}$	Thyroid	It is assumed that the person has radioactive iodine in the thyroid resulting from inhalation or ingestion.	Table 16			

22 30

	Exposure pathways	Effective dose to the representative person	Equivalent dose to the fetus
1	External exposure from radioactive material deposited on the ground (i.e. ground shine)	Effective dose to the infant	Equivalent dose to the fetus of the pregnant woman from external exposure to ground shine
2	External exposure from resuspended <sup>a</sup> radioactive material (i.e. air shine)	Effective dose to the infant	Equivalent dose to the fetus of the pregnant woman from external exposure to air shine
3	Inhalation of resuspended <sup>a</sup> radioactive material	Breathing rate for an adult performing light activity combined with the committed effective dose to the adult	Breathing rate for an adult (pregnant woman) performing light activity combined with the committed equivalent dose to the fetus from inhalation by the pregnant woman
4	Inadvertent ingestion of soil (e.g. from dirt on the hands)	Inadvertent ingestion rates by an infant during normal activity combined with the committed effective dose to the infant <sup>b</sup>	Inadvertent ingestion rates of dust and soil by an adult (pregnant woman) during normal activity combined with the committed equivalent dose to the fetus from ingestion by the pregnant woman

<sup>a</sup> Resuspended due to natural processes.
 <sup>b</sup> For this scenario, the combination of infant ingestion rates and dose conversion factors is controlling for all radionuclides.

TABLE 13. EXPOSURE PATHWAYS AND PARAMETERS FOR THE 'FOOD PRE-ANALYSIS' SCENARIO



# TABLE 14. EXPOSURE PATHWAYS AND PARAMETERS FOR THE 'SKIN' SCENARIO

	Exposure pathways	Effective dose to the representative person	Equivalent dose to the fetus
6	External exposure to the dermis from radioactive material on the skin	RBE weighted absorbed dose to the skin <sup>a</sup>	Not applicable
7	Inadvertent ingestion of radioactive material on the skin	Inadvertent ingestion rates by an infant combined with the committed effective dose to the infant <sup>b</sup>	Inadvertent ingestion rates of material on the skin by an adult (pregnant woman) combined with the committed equivalent dose to the fetus from inadvertent ingestion by the pregnant woman

<sup>a</sup> The external exposure to the dermis from radioactive material deposited on the skin is considered to be age

<sup>b</sup> For this scenario, the combination of infant ingestion rates and dose conversion factors is controlling for all radionuclides.

# TABLE 15. EXPOSURE PATHWAYS AND PARAMETERS FOR THE 'FOOD POST-ANALYSIS' SCENARIO



# TABLE 16. EXPOSURE PATHWAYS AND PARAMETERS FOR THE 'THYROID' SCENARIO



	Exposure pathways	Equivalent dose to the thyroid
9	Exposure of the thyroid from radioactive iodine in the thyroid	The committed equivalent dose to the thyroid of an infant is used

#### 3.5. BEHAVIOUR OF THE RADIONUCLIDES

This section considers the behaviour of the radionuclides (after being released into the environment) by deriving a series of factors used in the OIL calculations. Only the most relevant factors (i.e. those that could have a significant impact on the final default OIL value) are provided. This section is not a comprehensive description of radionuclide behaviour. Depending on the specific circumstances, other behaviours may need to be considered.

#### 3.5.1. Ground weathering and decay

The reduction in the external exposure rate from deposited radioactive material (ground shine) by ground weathering and decay is considered in the calculation of  $OIL1_{\gamma}$  and  $OIL2_{\gamma}$  by making use of  $WI_{G,i}(\Delta)$  [s], which is the time integrated external ground dose rate weathering factor, for radionuclide i, integrated over the exposure period  $\Delta$  (i.e. the period of time a person is living in the affected area). The values are given for the exposure periods of 7 d and 1 a in Table 17 and are determined by Eq. (4):

$$WI_{G,i}(\Delta) = \int_0^{\Delta} W_G(\tau) \times e^{-\lambda_i \times \tau} \times d\tau$$
(4)

where:

- $\tau$  [s] is the time after the deposition.
- $\lambda_i [s^{-1}]$  is the decay constant for radionuclide i. The values are given in Table 6.
- W<sub>G</sub>(τ) [unitless] is the external ground dose rate weathering factor at time τ after deposition. It is displayed in Fig. 5, and determined by Eq. (5), which is obtained from Ref. [41]:
- The use of newer models, such as those provided in Section 4.5 of Ref. [54], have little impact considering the exposure periods discussed here (i.e. 7 days and 1 year).

$$W_{\rm G}(\tau) = 0.63 \times e^{(-\beta_1 \times \tau)} + 0.37 \times e^{(-\beta_2 \times \tau)}$$
(5)

$$\circ$$
  $\beta_1 = 1.13 \text{ a}^{-1} = 3.59\text{E-}08 \text{ s}^{-1} \text{ and } \beta_2 = 7.48\text{E-}03 \text{ a}^{-1} = 2.37\text{E-}10 \text{ s}^{-1}$ 



*FIG. 5. External ground dose rate weathering factor*  $W_G(\tau)$  *[unitless].* 

TABLE 17. TIME INTEGRATED EXTERNAL GROUND DOSE RATE WEATHERING FACTOR AND TIME INTEGRATED TRANSFER FACTORS FROM THE GROUND TO THE AIR BY NATURAL RESUSPENSION

D	$WI_{G,i}(7d)^{a}$	$WI_{G,i}(1a)^{a}$	$TI_{grd \rightarrow air,i}(7d)^{b}$	TI <sub>grd→air,i</sub> (1a) <sup>b</sup>
Kaulonucilde	[s]	[ <b>s</b> ]	[s/m]	[s/m]
Rb-86	5.3E+05	2.2E+06	2.4E+00	3.2E+00
Sr-89	5.7E+05	5.5E+06	2.5E+00	4.1E+00
Sr-90+	6.0E+05	2.3E+07	2.5E+00	6.0E+00
Sr-91	5.0E+04	5.0E+04	4.7E-01	4.7E-01
Y-91	5.8E+05	6.3E+06	2.5E+00	4.2E+00
Zr-95+	5.8E+05	6.8E+06	2.5E+00	4.3E+00
Zr-97+	8.7E+04	8.7E+04	7.4E-01	7.4E-01
Mo-99+	2.8E+05	3.4E+05	1.6E+00	1.7E+00
Ru-103+	5.7E+05	4.4E+06	2.4E+00	3.9E+00
Ru-105	2.3E+04	2.3E+04	2.3E-01	2.3E-01
Ru-106+	6.0E+05	1.8E+07	2.5E+00	5.5E+00
Rh-105	1.8E+05	1.8E+05	1.2E+00	1.2E+00
Te-127m+	5.9E+05	1.0E+07	2.5E+00	4.7E+00
Te-127	4.9E+04	4.9E+04	4.6E-01	4.6E-01
Te-129m+	5.6E+05	3.8E+06	2.4E+00	3.7E+00
Te-131m	1.5E+05	1.6E+05	1.1E+00	1.1E+00
Te-132+	3.1E+05	4.0E+05	1.7E+00	1.8E+00
I-131	4.5E+05	9.8E+05	2.1E+00	2.5E+00
I-133	1.1E+05	1.1E+05	8.6E-01	8.6E-01
I-134	4.5E+03	4.5E+03	4.5E-02	4.5E-02
I-135	3.4E+04	3.4E+04	3.4E-01	3.4E-01
Cs-134	6.0E+05	2.0E+07	2.5E+00	5.7E+00
Cs-136	5.0E+05	1.6E+06	2.3E+00	2.9E+00
Cs-137+	6.0E+05	2.3E+07	2.5E+00	6.0E+00
Ba-140+	5.0E+05	1.5E+06	2.3E+00	2.9E+00
Ce-141	5.6E+05	3.7E+06	2.4E+00	3.7E+00
Ce-143	1.7E+05	1.7E+05	1.2E+00	1.2E+00
Ce-144+	6.0E+05	1.6E+07	2.5E+00	5.4E+00
Pr-143	5.1E+05	1.6E+06	2.3E+00	3.0E+00
Nd-147	4.9E+05	1.3E+06	2.2E+00	2.8E+00
Np-239	2.6E+05	2.9E+05	1.5E+00	1.5E+00
Pu-238	6.0E+05	2.3E+07	2.5E+00	6.0E+00
Pu-239	6.0E+05	2.4E+07	2.5E+00	6.0E+00
Pu-240	6.0E+05	2.4E+07	2.5E+00	6.0E+00
Pu-241	6.0E+05	2.3E+07	2.5E+00	6.0E+00
Am-241	6.0E+05	2.3E+07	2.5E+00	6.0E+00
Cm-242	5.9E+05	1.3E+07	2.5E+00	5.0E+00
Cm-244	6.0E+05	2.3E+07	2.5E+00	6.0E+00

<sup>a</sup> WI<sub>G,i</sub>(Δ) is calculated based on Eq. (4).
 <sup>b</sup> TI<sub>grd→air,i</sub>(Δ) is calculated based on Eq. (6).
 <sup>+</sup> The progenies that are in equilibrium with this parent radionuclide (as listed in Table 5) have been considered in the OIL calculations as being in equilibrium with the parent and do not need to be considered separately.

#### 3.5.2. Resuspension and decay

The influence of decay and resuspension of radioactive material from the ground to the air (by wind or other natural processes) is considered in the calculation of  $OIL1_{\gamma}$  and  $OIL2_{\gamma}$  by making use of  $TI_{grd \rightarrow air,i}(\Delta)$  [s/m], which is the time integrated transfer factor of radionuclide i from the ground to the air, integrated over the exposure period  $\Delta$  (i.e. the period of time living in the affected area). The values are given for an exposure period of 7 d and 1 a in Table 17 and are determined by Eq. (6):

$$TI_{grd \to air,i}(\Delta) = \int_0^{\Delta} T_{grd \to air}(\tau) \times e^{-\lambda_i \times \tau} \times d\tau$$
 (6)<sup>32</sup>

where:

- $\tau$  [s] is the time after deposition.
- $\lambda_i [s^{-1}]$  is the decay constant for radionuclide i. The values are given in Table 6.
- $T_{grd \rightarrow air}(\tau) \text{ [m}^{-1}\text{]}$  is the transfer factor from the ground to the air (by wind or other natural processes) at time  $\tau$  after deposition, also called resuspension factor. It is displayed in Fig. 6, determined by Eq. (7) and is consistent with Refs [55, 57, 58]. The use of newer models like those from Ref. [56] has little impact on the OIL values for LWRs.<sup>33,34</sup>

$$T_{\text{grd}\to\text{air}}(\tau) = \begin{cases} T_{\text{grd}\to\text{air}}(\tau_0) & \text{for } 0 < \tau \le \tau_0 \\ T_{\text{grd}\to\text{air}}(\tau_0) \times \frac{\tau_0}{\tau} + 10^{-9} & \text{for } \tau > \tau_0 \end{cases}$$
(7)

- $\circ$   $\tau_0 = 1 \ d = 86 \ 400 \ s.$
- $T_{grd \to air}(\tau_0) = 10^{-5} \text{ m}^{-1}$  is the transfer factor from the ground to the air (by wind or other natural processes), at time  $\tau_0$ . The default value was based on an examination of the results of Eq. (8) shown in Table 18 for different conditions. The default value was selected, because it would be conservative compared with most of those normally encountered by the public while living normally in an area. However, the inhalation and cloud shine doses from resuspended material are not significant contributors following a severe release of radioactive material from an LWR or its spent fuel.

$$T_{\text{grd}\to\text{air}}(\tau_0) \equiv \frac{\text{Concentration in the air}}{\text{Concentration on the ground}} \quad \frac{[\text{Bq}/\text{m}^3]}{[\text{Bq}/\text{m}^2]} \tag{8}$$

<sup>&</sup>lt;sup>32</sup> The integral was solved numerically (by using a common mathematical computer program).

<sup>&</sup>lt;sup>33</sup> For the exposure periods considered here (i.e. 7 days and 1 year).

<sup>&</sup>lt;sup>34</sup> The model used here is initially conservative by around an order of magnitude when compared with those normally encountered by the public while living normally in an area.



FIG. 6. Transfer factor from the ground to the air  $T_{grd \to air}(\tau)$  [m<sup>-1</sup>], also called resuspension factor.

# TABLE 18. TRANSFER FACTORS FROM THE GROUND TO THE AIR (ALSO CALLED RESUSPENSION FACTORS) FOR DIFFERENT REPRESENTATIVE CONDITIONS

Conditions	${{\operatorname{T}}_{{\operatorname{grd}} \to {\operatorname{air}}}({{ au }_0})^a} \ [{{{\mathbf{m}}^{ - 1}}}]$
Wet — rain (minimum value)	1E-09
Normal conditions and rural conditions with light to moderate wind	1E-06
Arid and urban conditions (default value)	1E-05
Very dusty conditions (e.g. ploughing in dry conditions)	1E-04

<sup>a</sup>  $T_{grd \rightarrow air}(\tau_0)$  is obtained from Refs [57, 58, 59].

### 3.5.3. Inadvertent ingestion from the ground and decay

The influence of decay and the transfer of radioactive material from the ground to the gastrointestinal (GI) tract by inadvertent ingestion is considered in the calculation of  $OIL1_{\gamma}$  and  $OIL2_{\gamma}$  by making use of  $TI_{grd \rightarrow GI,i}(\Delta,g)$  [m<sup>2</sup>], which is the time integrated transfer factor of radionuclide i from the ground to the GI tract by inadvertent ingestion for an individual in age group g (i.e. infants for the representative person and adults for the fetal exposure), integrated over the exposure period  $\Delta$  (i.e. period of time living in the affected area).

Since no model for the weathering effects on inadvertent ingestion was found, it was assumed that the material available for inadvertent ingestion is reduced in a similar way as for resuspension (as discussed in Eq. (7)), which is supported by the following considerations:

• As the material migrates into the soil, the transfer of radioactivity from the ground to the air or to the GI tract by inadvertent ingestion is reduced by the same factors: (a) dilution by unaffected soil; and (b) movement of the material beyond the depth that is likely to be resuspended or inadvertently ingested. The use of Eqs (9), (10) and (11) to model this

reduction with time is consistent with the traditional assumption that, on average, deposited material moves to a depth of about 1 cm within the first month  $[55]^{35}$  and thus below a depth available for resuspension or inadvertent ingestion of 1 mm (see Eq. (11)). It is also consistent with data from the Chernobyl accident indicating that, after a few weeks, it is not uncommon for downward migration of the activity in the soil to result in a relaxation length<sup>36</sup> of 1 cm, and, after a few months, to result in a relaxation length of 3 cm, before the downward migration begins to slow down [60].

For material deposited on pavement, the material available for resuspension or inadvertent ingestion is reduced rapidly by either natural processes (e.g. wind or rain) or other processes (e.g. road traffic) or becomes fixed [61].

The values for  $TI_{grd \rightarrow Gl,i}(\Delta,g)$  [m<sup>2</sup>] are given for an exposure period of 7 d and 1 a and for the infant and adult in Table 19 and are determined by Eq. (9):

$$TI_{grd \to GI,i}(\Delta, g) = \int_0^{\Delta} T_{grd \to GI}(\tau, g) \times e^{-\lambda_i \times \tau} \times d\tau$$
(9)<sup>37</sup>

where:

- $\tau$  [s] is the time after deposition.
- $\lambda_i$  [s<sup>-1</sup>] is the decay constant for radionuclide i. The values are given in Table 6.
- $T_{\text{erd}\to GI}(\tau,g)$  [m<sup>2</sup>/s] is the transfer factor from the ground to the GI tract by inadvertent ingestion, for an individual in age group g (i.e. the infant for the representative person and the adult for the fetus), at time  $\tau$  after deposition. It is displayed in Fig. 7 and determined by Eq. (10):

$$T_{\text{grd}\to\text{GI}}(\tau,g) = \begin{cases} T_{\text{grd}\to\text{GI}}(\tau_0,g) & \text{for } 0 < \tau \le \tau_0 \\ T_{\text{grd}\to\text{GI}}(\tau_0,g) \times \frac{\tau_0}{\tau} & \text{for } \tau > \tau_0 \end{cases}$$
(10)

- $\tau_0 = 1 \text{ d} = 86 400 \text{ s}$  (it is normally provided in days). 0
- $T_{erd \rightarrow GI}(\tau_0,g)$  [m<sup>2</sup>/s] is the transfer factor from the ground to the GI tract by inadvertent ingestion (i.e. the infant for the representative person and the adult for the fetus), for an individual in age group g, at time  $\tau_0$ . The values are given below and determined by Eq. (11):
  - 7.2E-10  $m^2/s$  for intake by the infant (considered for the representative person);
  - $3.6\text{E-10} \text{ m}^2/\text{s}$  for intake by adults (considered for the fetal exposure).

<sup>&</sup>lt;sup>35</sup> Inadvertent ingestion of material at depths larger than those assumed would not greatly affect the  $T_{grd \rightarrow Gl}(\tau,g)$ value, because this would be offset by the lower activity concentration of the material ingested due to dilution (see Eq. (11)).

<sup>&</sup>lt;sup>36</sup> The relaxation length is the thickness of a shielding material (e.g. soil) that will reduce the intensity of the radiation (e.g. ground shine) to 1/e (37%) of its original. <sup>37</sup> The integral was solved numerically (by using a common mathematical computer program).

$$T_{\text{grd}\to\text{GI}}(\tau_0, g) = \frac{Q_{\text{soil}}(g)}{\rho_{\text{dep}} \times d_{\text{inadv-ing}} \times F_{\text{soil,inadv-ing}}(\tau_0)}$$
(11)

- $\rho_{dep} = 1600 \text{ mg/cm}^3 = 1600 \text{ kg/m}^3$  is the density of the deposition. It is obtained from Ref. [62].
- d<sub>inadv-ing</sub> = 1 mm = 0.001 m is the depth available for inadvertent ingestion. It is obtained from Ref. [62].
- $F_{soil,inadv-ing}(\tau_0) = 1$  [unitless] is the fraction of the deposited material available for ingestion (e.g. material that is within the d<sub>inadv-ing</sub> or not adhered to a surface) at time  $\tau_0$ .
- Q<sub>soil</sub>(g) [kg/s] is the inadvertent ingestion rate of ground deposition for an individual in age group g living under normal conditions. The default values are given below, obtained from Ref. [63], and are consistent with findings of studies for dust and soil ingestion under normal circumstances, such as infants playing outside or being in camp grounds<sup>38</sup> [62]:
  - 100 mg/d = 1.2E-09 kg/s for intake by infants (considered for the representative person);
  - 50 mg/d = 5.8E-10 kg/s for intake by adults (considered for the fetal exposure).



FIG. 7. Transfer factors from the ground to the GI tract by inadvertent ingestion,  $T_{grd \rightarrow Gl}(\tau,g) [m^2/s]$ .

<sup>&</sup>lt;sup>38</sup> Soil-pica, which is the recurrent ingestion of unusually high amounts of soil (i.e. around 1 to 5 g/day), and usually involves the consumption of the top 5 to 8 cm of surface soils [63], would not affect the default  $T_{grd \rightarrow GI}(\tau_0,g)$  value, because the higher intake rates are offset by the larger inadvertent ingestion depth  $d_{inadv-ing}$  of 5 to 8 cm.

Dadianualida	TI <sub>grd→GI,i</sub> (7d, infant) <sup>a</sup>	$TI_{grd \rightarrow GI,i}(7d, adult)^{a}$	TI <sub>grd→GI,i</sub> (1a, infant) <sup>a</sup>	$TI_{grd \rightarrow GI,i}(1a, adult)^{a}$
Radionucide	[m <sup>2</sup> ]	[ <b>m</b> <sup>2</sup> ]	[m <sup>2</sup> ]	[m <sup>2</sup> ]
Rb-86	1.7E-04	8.5E-05	2.3E-04	1.2E-04
Sr-89	1.8E-04	8.9E-05	2.9E-04	1.5E-04
Sr-90+	1.8E-04	9.2E-05	4.3E-04	2.1E-04
Sr-91	3.4E-05	1.7E-05	3.4E-05	1.7E-05
Y-91	1.8E-04	9.0E-05	3.0E-04	1.5E-04
Zr-95+	1.8E-04	9.0E-05	3.1E-04	1.5E-04
Zr-97+	5.4E-05	2.7E-05	5.4E-05	2.7E-05
<b>Mo-99</b> +	1.2E-04	5.8E-05	1.2E-04	6.0E-05
Ru-103+	1.8E-04	8.9E-05	2.8E-04	1.4E-04
Ru-105	1.7E-05	8.3E-06	1.7E-05	8.3E-06
Ru-106+	1.8E-04	9.2E-05	3.9E-04	2.0E-04
Rh-105	8.7E-05	4.3E-05	8.7E-05	4.4E-05
Te-127m+	1.8E-04	9.1E-05	3.4E-04	1.7E-04
Te-127	3.3E-05	1.7E-05	3.3E-05	1.7E-05
Te-129m+	1.8E-04	8.8E-05	2.7E-04	1.3E-04
Te-131m	7.9E-05	3.9E-05	7.9E-05	4.0E-05
Te-132+	1.2E-04	6.2E-05	1.3E-04	6.5E-05
I-131	1.5E-04	7.7E-05	1.8E-04	9.1E-05
I-133	6.2E-05	3.1E-05	6.2E-05	3.1E-05
I-134	3.3E-06	1.6E-06	3.3E-06	1.6E-06
I-135	2.4E-05	1.2E-05	2.4E-05	1.2E-05
Cs-134	1.8E-04	9.2E-05	4.1E-04	2.1E-04
Cs-136	1.6E-04	8.2E-05	2.1E-04	1.1E-04
Cs-137+	1.8E-04	9.2E-05	4.3E-04	2.1E-04
Ba-140+	1.6E-04	8.2E-05	2.1E-04	1.1E-04
Ce-141	1.8E-04	8.8E-05	2.7E-04	1.3E-04
Ce-143	8.3E-05	4.2E-05	8.4E-05	4.2E-05
Ce-144+	1.8E-04	9.2E-05	3.9E-04	1.9E-04
Pr-143	1.7E-04	8.3E-05	2.1E-04	1.1E-04
Nd-147	1.6E-04	8.1E-05	2.0E-04	1.0E-04
Np-239	1.1E-04	5.4E-05	1.1E-04	5.6E-05
Pu-238	1.8E-04	9.2E-05	4.3E-04	2.2E-04
Pu-239	1.8E-04	9.2E-05	4.3E-04	2.2E-04
Pu-240	1.8E-04	9.2E-05	4.3E-04	2.2E-04
Pu-241	1.8E-04	9.2E-05	4.3E-04	2.1E-04
Am-241	1.8E-04	9.2E-05	4.3E-04	2.2E-04
Cm-242	1.8E-04	9.1E-05	3.6E-04	1.8E-04
Cm-244	1.8E-04	9.2E-05	4.3E-04	2.1E-04

TABLE 19. TIME INTEGRATED TRANSFER FACTORS FROM THE GROUND TO THE GI TRACT BY INADVERTENT INGESTION

<sup>a</sup> TI<sub>grd→GI,i</sub>(Δ,g) is calculated based on Eq. (9).
 + The progenies that are in equilibrium with this parent radionuclide (as listed in Table 5) have been considered in the OIL calculations as being in equilibrium with the parent and do not need to be considered separately.

#### 3.5.4. Transfer from the ground to vegetables and pasture

The fraction of deposited activity intercepted by the edible portion of vegetation, per unit mass, as the result of deposition processes is considered in the calculation of  $OIL3_{\gamma}$  by making use of mass interception factors. For pasture forage, conventionally the dry weight is used, while for fresh vegetables, the wet weight is used:

- $\Phi_1 = 3 \text{ m}^2/\text{kg}$  is the mass interception factor for pasture grass in dry weight, i.e. the ratio between the activity concentration on the plant [Bq/kg] and the unit ground surface activity on the terrestrial surface [Bq/m<sup>2</sup>] (soil plus vegetation). It is obtained from Table VII of Ref. [64].
- $\Phi_2 = 0.3 \text{ m}^2/\text{kg}$  is the mass interception factor for leafy vegetables (fresh or wet weight), i.e. the ratio between the activity concentration on the plant [Bq/kg] and the unit ground surface activity on the terrestrial surface [Bq/m<sup>2</sup>] (soil plus vegetation). It is obtained from Table VII of Ref. [64].

#### 3.5.5. Removal from vegetation by weathering and decay

The reduction of the gross activity from vegetation due to natural processes (i.e. weathering, but not intentional washing) is considered in the calculation of  $OIL3_{\gamma}$  by making use of  $\Delta_{eff,OIL3,i}$  [s], which is the effective availability period for radionuclide i for ingestion in the 'food pre-analysis' scenario. The values are given in Table 20 and are determined by Eq. (12):

$$\Delta_{eff,OIL3,i} = \int_{0}^{\Delta_{available-ing}} e^{-(\lambda_{w-vegetation} + \lambda_i) \times t} dt = \frac{1 - e^{-(\lambda_{w-vegetation} + \lambda_i) \times \Delta_{available-ing}}}{\lambda_{w-vegetation} + \lambda_i}$$
(12)

where:

- $\Delta_{\text{available-ing}} = 365 \text{ d} = 3.15\text{E}+07 \text{ s}$  is the period of time that the affected food or forage is available for consumption. This is the exposure period of the GC(Ingestion, e<sub>ing</sub>, 1a).
- $\lambda_i [s^{-1}]$  is the decay constant for radionuclide i. The values are given in Table 6.
- λ<sub>w-vegetation</sub> [s<sup>-1</sup>] is the removal constant from vegetation due to weathering. It is determined by Eq. (13):

$$\lambda_{\text{w-vegetation}} = \frac{\ln(2)}{T_{1/2,\text{w-vegetation}}}$$
(13)

where:

•  $T_{1/2,w-vegetation} = 14 d = 1.2E+06 s$  is the half-life for removal of surface deposition from vegetation due to weathering. It is obtained from Ref. [64].

Padionualida	$\Delta_{ m eff,OIL3,i}^{a}$	$\Delta_{ m eff,OIL4-Urgent,i}^{ m b}$	$\Delta_{ m eff,OIL4-Acute,i}^{ m c}$	$\Delta_{ m eff,OIL7,i}^{d}$
Kaulollucliue	[s]	[s]	[s]	[s]
Rb-86	1.0E+06	7.4E+04	2.8E+04	2.3E+06
Sr-89	1.4E+06	7.5E+04	2.9E+04	6.3E+06
Sr-90+	1.7E+06	7.6E+04	2.9E+04	3.1E+07
Sr-91	4.9E+04	3.0E+04	2.1E+04	5.0E+04
Y-91	1.4E+06	7.6E+04	2.9E+04	7.2E+06
Zr-95+	1.4E+06	7.6E+04	2.9E+04	7.8E+06
Zr-97+	8.3E+04	4.1E+04	2.4E+04	8.7E+04
Mo-99+	2.9E+05	6.2E+04	2.7E+04	3.4E+05
Ru-103+	1.3E+06	7.5E+04	2.9E+04	4.9E+06
Ru-105	2.3E+04	1.8E+04	1.5E+04	2.3E+04
Ru-106+	1.7E+06	7.6E+04	2.9E+04	2.3E+07
Rh-105	1.7E+05	5.4E+04	2.6E+04	1.8E+05
Te-127m+	1.5E+06	7.6E+04	2.9E+04	1.2E+07
Te-127	4.7E+04	3.0E+04	2.1E+04	4.9E+04
Te-129m+	1.2E+06	7.5E+04	2.9E+04	4.2E+06
Te-131m	1.4E+05	5.1E+04	2.6E+04	1.6E+05
Te-132+	3.2E+05	6.4E+04	2.8E+04	4.0E+05
I-131	6.4E+05	7.1E+04	2.8E+04	1.0E+06
I-133	1.0E+05	4.5E+04	2.5E+04	1.1E+05
I-134	4.5E+03	4.3E+03	4.3E+03	4.5E+03
I-135	3.3E+04	2.4E+04	1.8E+04	3.4E+04
Cs-134	1.7E+06	7.6E+04	2.9E+04	2.7E+07
Cs-136	8.5E+05	7.3E+04	2.8E+04	1.6E+06
Cs-137+	1.7E+06	7.6E+04	2.9E+04	3.1E+07
Ba-140+	8.3E+05	7.3E+04	2.8E+04	1.6E+06
Ce-141	1.2E+06	7.5E+04	2.9E+04	4.1E+06
Ce-143	1.6E+05	5.3E+04	2.6E+04	1.7E+05
Ce-144+	1.7E+06	7.6E+04	2.9E+04	2.1E+07
Pr-143	8.6E+05	7.3E+04	2.8E+04	1.7E+06
Nd-147	7.7E+05	7.2E+04	2.8E+04	1.4E+06
Np-239	2.5E+05	6.1E+04	2.7E+04	2.9E+05
Pu-238	1.7E+06	7.6E+04	2.9E+04	3.1E+07
Pu-239	1.7E+06	7.6E+04	2.9E+04	3.2E+07
Pu-240	1.7E+06	7.6E+04	2.9E+04	3.2E+07
Pu-241	1.7E+06	7.6E+04	2.9E+04	3.1E+07
Am-241	1.7E+06	7.6E+04	2.9E+04	3.2E+07
Cm-242	1.6E+06	7.6E+04	2.9E+04	1.6E+07
Cm-244	1.7E+06	7.6E+04	2.9E+04	3.1E+07

# TABLE 20. EFFECTIVE AVAILABILITY PERIODS

 <sup>a</sup> Δ<sub>eff,OIL3,i</sub> is calculated based on Eq. (12).
 <sup>b</sup> Δ<sub>eff,OIL4-Urgent,i</sub> is calculated based on Eq. (16).
 <sup>c</sup> Δ<sub>eff,OIL4-Acute,i</sub> is calculated based on Eq. (18).
 <sup>d</sup> Δ<sub>eff,OIL7,i</sub> is calculated based on Eq. (19).
 <sup>+</sup> The progenies that are in equilibrium with this parent radionuclide (as listed in Table 5) have been considered in the progenies that are in equilibrium with this parent radionuclide (as listed in Table 5) have been considered in the progenies that are in equilibrium with this parent radionuclide (as listed in Table 5) have been considered in the progenies that are in equilibrium with this parent radionuclide (as listed in Table 5) have been considered in the progenies that are in equilibrium with this parent radionuclide (as listed in Table 5) have been considered in the progenies that are in equilibrium with this parent radionuclide (as listed in Table 5) have been considered in the progenies that are in equilibrium with this parent radionuclide (as listed in Table 5) have been considered in the progenies that are in equilibrium with this parent radionuclide (as listed in Table 5) have been considered in the progenies that are in equilibrium with this parent radionuclide (as listed in Table 5) have been considered in the progenies that are in equilibrium with this parent radionuclide (as listed in Table 5) have been considered in the progenies that are in equilibrium with the pro the OIL calculations as being in equilibrium with the parent and do not need to be considered separately.

#### 3.5.6. Transfer from cow's feed to cow's milk

The reduction in the gross activity from feed to cow's milk is considered in the calculation of  $OIL3_{\gamma}$  by making use of the transfer factor  $T_{feed \rightarrow cow-milk,i} [s/L] ([(Bq/L)/(Bq/s)] = [s/L])$ . The values are given in Table 21.

#### 3.5.7. Reduction due to processing and delay before consumption

The reduction in the gross activity due to the period of time between deposition and consumption is considered in the calculation of  $OIL3_{\gamma}$  by making use of  $F_{consumption,i}$  [unitless], which is the fraction of the radionuclide i remaining at the time of human consumption. The values are given in Table 22 and determined by Eq. (14):

$$F_{\text{consumption,i}} = e^{-\lambda_i \times \Delta_{\text{dep} \to \text{consumption}}} \times F_{\text{prep}}$$
(14)

where:

- $\lambda_i [s^{-1}]$  is the decay constant for radionuclide i. The values are given in Table 6.
- $\Delta_{dep \rightarrow consumption} = 1$  d = 86 400 s is the period of time assumed between deposition and consumption.
- F<sub>prep</sub> = 1 [unitless] is the fraction of the radioactive material remaining for ingestion after preparations. It assumes no reduction due to normal preparations/processing (e.g. washing and cooking), which, in some cases, substantially reduces the concentration of radionuclides in the food [65].

#### 3.5.8. Transfer from the skin to the GI tract by inadvertent ingestion

The transfer of radioactive material from the skin to the GI tract is considered in the calculation of  $OIL4_{\gamma}$  and  $OIL4_{\beta}$  by making use of  $T_{skin\rightarrow GI}(g)$  [m<sup>2</sup>/s], which is the transfer factor from the skin to the GI tract due to inadvertent ingestion for an individual in age group g (i.e. the infant for the representative person and the adult for the fetus). It is considered a reasonable upper limit, because it assumes that no actions are taken to limit inadvertent ingestion (e.g. washing hands). The default values are given below and determined by Eq. (15):

- 3.2E-08 m<sup>2</sup>/s for intake by adults (considered for the fetal exposure);
- 6.4E-08 m<sup>2</sup>/s for intake by infants (considered for the representative person).

$$T_{skin \to GI}(g) = \frac{Q_{soil}(g)}{L_{hand}}$$
(15)

- Q<sub>soil</sub>(g) [kg/s] is the inadvertent ingestion rate of ground deposition for an individual in age group g living under normal conditions. Its basis is described in Section 3.5.3., which also provides the default values.
- $L_{hand} = 1.8 \text{ mg/cm}^2 = 0.018 \text{ kg/m}^2$  is the amount of soil loaded on the hand of the maximally exposed individual (e.g. children playing outdoors). It is obtained from Ref. [62].

TABLE 21. TRANSFER FACTOR OF RADIONUCLIDE i FROM FEED TO THE MILK OF THE COW

Radionuclida	T <sub>feed→cow-milk,i</sub> <sup>a</sup>	$T_{feed \rightarrow cow-milk,i}^{a}$	Doforonco <sup>b</sup>
Kaulonuchue	[d/L]	[s/L]	Kelefence
Rb-86	1.0E-01	8.6E+03	[64]
Sr-89	1.3E-03	1.1E+02	[65]
Sr-90+	1.3E-03	1.1E+02	[65]
Sr-91	1.3E-03	1.1E+02	[65]
Y-91	6.0E-05	5.2E+00	[64]
Zr-95+	3.6E-06	3.1E-01	[65]
Zr-97+	3.6E-06	3.1E-01	[65]
Mo-99+	1.1E-03	9.5E+01	[65]
Ru-103+	9.4E-06	8.1E-01	[65]
Ru-105	9.4E-06	8.1E-01	[65]
Ru-106+	9.4E-06	8.1E-01	[65]
Rh-105	5.0E-04	4.3E+01	[64]
Te-127m+	3.4E-04	2.9E+01	[65]
Te-127	3.4E-04	2.9E+01	[65]
Te-129m+	3.4E-04	2.9E+01	[65]
Te-131m	3.4E-04	2.9E+01	[65]
Te-132+	3.4E-04	2.9E+01	[65]
I-131	5.4E-03	4.7E+02	[65]
I-133	5.4E-03	4.7E+02	[65]
I-134	5.4E-03	4.7E+02	[65]
I-135	5.4E-03	4.7E+02	[65]
Cs-134	4.6E-03	4.0E+02	[65]
Cs-136	4.6E-03	4.0E+02	[65]
Cs-137+	4.6E-03	4.0E+02	[65]
Ba-140+	1.6E-04	1.4E+01	[65]
Ce-141	2.0E-05	1.7E+00	[65]
Ce-143	2.0E-05	1.7E+00	[65]
Ce-144+	2.0E-05	1.7E+00	[65]
Pr-143	5.0E-06	4.3E-01	[66]
Nd-147	5.0E-06	4.3E-01	[66]
Np-239	5.0E-05	4.3E+00	[64]
Pu-238	1.0E-05	8.6E-01	[65]
Pu-239	1.0E-05	8.6E-01	[65]
Pu-240	1.0E-05	8.6E-01	[65]
Pu-241	1.0E-05	8.6E-01	[65]
Am-241	4.2E-07	3.6E-02	[65]
Cm-242	2.0E-06	1.7E-01	[64]
Cm-244	2.0E-06	1.7E-01	[64]

<sup>a</sup> The values for small grazing animals are assumed to be approximately 10 times higher [65].
<sup>b</sup> The mean values provided in Ref. [65] were used first, and, if not available in Ref. [65], the values were taken from Ref. [64]. If not available in Ref. [64] the values were taken from Ref. [66].
+ The progenies that are in equilibrium with this parent radionuclide (as listed in Table 5) have been considered in the formation of the formation of the formation of the formation of the formation.

the OIL calculations as being in equilibrium with the parent and do not need to be considered separately.

# TABLE 22. FRACTION OF THE RADIONUCLIDES REMAINING AT THE TIME OF HUMAN CONSUMPTION

Dadionualida	F <sub>consumption,i</sub> <sup>a</sup>
Kadionuciide	[unitless]
Rb-86	9.6E-01
Sr-89	9.9E-01
Sr-90+	1.0E+00
Sr-91	1.8E-01
Y-91	9.9E-01
Zr-95+	9.9E-01
Zr-97+	3.7E-01
Mo-99+	7.8E-01
Ru-103+	9.8E-01
Ru-105	2.4E-02
Ru-106+	1.0E+00
Rh-105	6.2E-01
Te-127m+	9.9E-01
Te-127	1.7E-01
Te-129m+	9.8E-01
Te-131m	5.7E-01
Te-132+	8.1E-01
I-131	9.2E-01
I-133	4.5E-01
I-134	5.5E-09
I-135	7.9E-02
Cs-134	1.0E+00
Cs-136	9.5E-01
Cs-137+	1.0E+00
Ba-140+	9.5E-01
Ce-141	9.8E-01
Ce-143	6.0E-01
Ce-144+	1.0E+00
Pr-143	9.5E-01
Nd-147	9.4E-01
Np-239	7.5E-01
Pu-238	1.0E+00
Pu-239	1.0E+00
Pu-240	1.0E+00
Pu-241	1.0E+00
Am-241	1.0E+00
Cm-242	1.0E+00
Cm-244	1.0E+00

<sup>a</sup> F<sub>consumption,i</sub> is calculated based on Eq. (14).
 + The progenies that are in equilibrium with this parent radionuclide (as listed in Table 5) have been considered in the OIL calculations as being in equilibrium with the parent and do not need to be considered separately.

#### 3.5.9. Removal from the skin by weathering and decay

The reduction of the gross activity on the skin surface due to natural processes (i.e. weathering, but not washing) is considered in the calculation of  $OIL4_{\gamma}$  and  $OIL4_{\beta}$  by making use of the effective availability period for inadvertent ingestion from the skin. Since two exposure periods for material on the skin (i.e. 7 d for urgent generic criteria and 10 h for acute generic criteria) are considered in the calculations, two corresponding effective availability periods for inadvertent ingestion of material on the skin are determined:

•  $\Delta_{\text{eff,OIL4-Urgent,i}}$  [s] is the effective availability period for radionuclide i for inadvertent ingestion from the skin in the 'skin' scenario for the urgent generic criteria. The values are given in Table 20 and are determined by Eq. (16):

$$\Delta_{\text{eff,OIL4-Urgent,i}} = \int_{0}^{\Delta_{\text{OIL4-Urgent}}} e^{-(\lambda_{i} + \lambda_{w-skin}) \times t} \times dt = \frac{1 - e^{-(\lambda_{i} + \lambda_{w-skin}) \times \Delta_{\text{OIL4-Urgent}}}{\lambda_{i} + \lambda_{w-skin}}$$
(16)

where:

- $\circ \Delta_{OIL4-Urgent} = 7 d = 604 800 s$  is the exposure period considered for the urgent generic criteria (given in Table 2). It is obtained from Table II.2 of Ref. [1].
- $\circ$   $\lambda_i$  [s<sup>-1</sup>] is the decay constant for radionuclide i. The values are given in Table 6.
- $\circ \lambda_{w-skin} [s^{-1}]$  is the removal constant from skin due to weathering. It is determined by Eq. (17):

$$\lambda_{\text{w-skin}} = \ln(2) / T_{1/2,\text{w-skin}}$$
(17)

where:

- $T_{1/2,w-skin} = 14.7 h = 52 920 s$  is the half-life for removal of surface deposition from the skin due to weathering. It is obtained from Ref. [67].
- Δ<sub>eff,OIL4-Acute,i</sub> [s] is the effective availability period for radionuclide i for inadvertent ingestion from the skin in the 'skin' scenario for the acute generic criteria. The values are given in Table 20 and are determined by Eq. (18):

$$\Delta_{\text{eff,OIL4-Acute,i}} = \int_{0}^{\Delta_{\text{OIL4-Acute}}} e^{-(\lambda_{i} + \lambda_{w-skin}) \times t} \times dt = \frac{1 - e^{-(\lambda_{i} + \lambda_{w-skin}) \times \Delta_{\text{OIL4-Acute}}}{\lambda_{i} + \lambda_{w-skin}}$$
(18)

where:

- $\circ \Delta_{OII.4-Acute} = 10$  h = 36 000 s is the exposure period considered for the acute generic criteria (Table 2).
- $\circ$   $\lambda_i$  [s<sup>-1</sup>] is the decay constant for radionuclide i. The values are given in Table 6.
- $\circ \lambda_{w-skin} [s^{-1}]$  is the removal constant from skin due to weathering. It is determined by Eq. (17).

#### 3.5.10. Reduction due to radioactive decay during the period of availability

The reduction of the gross activity in food, milk or drinking water due to decay is considered in the calculation of OIL7 by making use of  $\Delta_{eff,OIL7,i}$  [s], which is the effective availability period for

radionuclide i for ingestion in the 'food post-analysis' scenario. The values are given in Table 20 and are determined by Eq. (19):

$$\Delta_{\text{eff,OIL7},i} = \int_{0}^{\Delta_{\text{available-ing}}} e^{-\lambda_i \times t} \times dt = \frac{1 - e^{-\lambda_i \times \Delta_{\text{available-ing}}}}{\lambda_i}$$
(19)

where:

- $\lambda_i [s^{-1}]$  is the decay constant for radionuclide i. The values are given in Table 6.
- $\Delta_{\text{available-ing}} = 365 \text{ d} = 3.15\text{E}+07 \text{ s}$  is the period of time that the affected food or forage is available for consumption. This is the exposure period of the GC(Ingestion, e<sub>ing</sub>, 1a).

#### 3.5.11. Reduction of the measured dose rate due to the decay of I-131 in the thyroid

The reduction of the activity of I-131 in the thyroid due to decay and biological processes is considered in the calculation of  $OIL8_{\gamma}$  by making use of  $CorF_{I-131-thyroid-burden}(t_m)$  [unitless], which is displayed in Fig. 8 and determined by Eq. (20):

$$CorF_{I-131-thyroid-burden}(t_m) = e^{-(\lambda_{I-131} + \lambda_{bio-I-131}) \times t_m}$$
(20)

where:

- t<sub>m</sub> [s] is the time between intake and measurement of the thyroid.
- $\lambda_{I-131} = 1.0E-06 \text{ s}^{-1}$  is the decay constant of I-131. The value is taken from Table 6.
- $\lambda_{\text{bio-I-131}} = 1.0\text{E-07 s}^{-1}$  is the biological removal constant of I-131 in the thyroid. It is determined by Eq. (21):

$$\lambda_{\text{bio}-I-131} = \ln(2) / T_{1/2,\text{bio}-I-131}$$
(21)

where:

 $\circ$  T<sub>1/2,bio-I-131</sub> = 80 d = 6.9E+06 s is the biological half-life of I-131 provided in Ref. [68].



FIG. 8. Correction factor to account for the reduction in the dose rate from the thyroid due to the reduction of I-131,  $CorF_{I-131-thyroid-burden}(t_m)$  [unitless].

#### 3.6. DOSE CONVERSION FACTORS

This section provides the dose conversion factors used in the calculation of the OILs and their basis, following the exposure scenarios described in Section 3.4. Both effective and equivalent dose conversion factors are considered in the calculation of the OILs because the effective dose alone cannot be used as a basis for estimating the possible health hazard from radiation exposure [38], since it can greatly underestimate the possible radiation induced impact to individual organs following intake of radioactive material.<sup>39</sup> As mentioned in Section 3.4, the total effective dose to the representative person and the equivalent dose to its fetus<sup>40</sup> will be controlling for the 'ground', 'food pre-analysis', 'skin' and 'food post-analysis' scenarios, while for the 'thyroid scenario', it will be the equivalent dose to the thyroid of the representative person.

Sections 3.6.1 to 3.6.5 provide the dose conversion factors for the exposure scenarios described in Table 11 to Table 16. The most restrictive combinations of different factors such as age, public behaviour, exposure pathways, chemical forms of the radionuclides, radionuclide behaviour, times relative to conception, acute or chronic exposure and types of absorption are used. Section 3.10 discusses the considered conservatism in further detail.

In the notation of the dose conversion factors, the lower case letter is used to indicate that a single exposure pathway is considered when calculating the dose, as opposed to the total dose, for which an upper case letter is used (e.g.  $E_{grd-scenario,i}$ ,  $e_{grd-sh(grd-scenario),i}$ ).

#### 3.6.1. Dose conversion factors for the 'ground' scenario

Table 23 provides the dose conversion factors for the total effective dose to the representative person and the total equivalent dose to the fetus for the 'ground' scenario (needed in the calculation of  $OIL1_{\gamma}$  and  $OIL2_{\gamma}$ ), over an exposure period  $\Delta$  of 7 days and 1 year.<sup>41</sup> These dose conversion factors are determined as described in Sections 3.6.1.1 and 3.6.1.2.

#### 3.6.1.1. Total effective dose to the representative person for the 'ground' scenario

Table 23 provides  $E_{grd-scenario,i}(\Delta)$  [Sv/(Bq/m<sup>2</sup>)], which is the total effective dose to the representative person over an exposure period  $\Delta^{41}$  for the 'ground' scenario, per unit ground surface activity of radionuclide i. It is determined by Eq. (22):

<sup>&</sup>lt;sup>39</sup> For example: The intake of radioiodine resulting in an effective dose below the generic criteria for urgent protective actions could result in an equivalent dose to the thyroid at which radiation induced health effects may occur.

<sup>&</sup>lt;sup>40</sup> In this publication the term 'fetus' encompasses both the embryo and the fetus. Keeping the equivalent dose to the fetus from intake below the generic criteria will ensure that the dose to other organs will also stay below the generic criteria. The equivalent dose to the fetus is defined as the highest equivalent dose to any organ in the fetus during the period of in utero development.

<sup>&</sup>lt;sup>41</sup> 7 days and 1 year are used, since those are the exposure periods provided for the corresponding generic criteria in the IAEA Safety Standards Series No. GSR Part 7 [1], as shown in Table 2. 7 days is used to calculate the early response action  $OIL1_{\gamma}$ , and 1 year is used to calculate the urgent response action  $OIL2_{\gamma}$ .

$$E_{\text{grd-scenario},i}(\Delta) = e_{\text{grd-sh}(\text{grd-scenario}),i}(\Delta) + e_{\text{air-sh}(\text{grd-scenario}),i}(\Delta) + e_{\text{inh-resusp}(\text{grd-scenario}),i}(\Delta) + e_{\text{inadv-ing}(\text{grd-scenario}),i}(\Delta)$$
(22)

where:

e<sub>grd-sh(grd-scenario),i</sub>(Δ) [Sv/(Bq/m<sup>2</sup>)] is the external effective dose to the representative person (i.e. the infant for this exposure scenario and pathway), over an exposure period Δ, from ground shine resulting from ground deposition, for the 'ground' scenario, per unit ground surface activity of radionuclide i. It is determined by Eq. (23):

$$e_{\text{grd-sh(grd-scenario),i}}(\Delta) = e_{\text{plane-srf,i}}(\text{adult}) \times \text{CorF}_{\text{grd}} \times \\ \times \text{SF}_{e_{\text{ext}}(\text{adult} \rightarrow \text{infant})} \times \text{WI}_{G,i}(\Delta) \times (F_{\text{sf}} \times F_{\text{of}} + (1 - F_{\text{of}}))$$
(23)

where:

- e<sub>plane-srf,i</sub>(adult) [(Sv/s)/(Bq/m<sup>2</sup>)] is the effective dose rate to the adult from an infinite smooth plane source, per unit surface activity of radionuclide i. The values are given in Table 24 and are applicable to an exposed individual standing on an ideal plane, uniformly contaminated surface. This would overestimate the dose from a real (non-plane) surface of uniform concentration; therefore, a ground roughness correction factor is used, as described in the next bullet.
- $CorF_{grd} = 0.7$  [unitless] is the ground roughness correction factor to account for the dose rate reduction due to ground roughness, as described in Ref. [69].
- $SF_{ext(adult \rightarrow infant)} = 1.4$  [unitless] is the scaling factor to convert the effective dose to the adult to the effective dose to the infant for external exposure, as discussed in Ref. [69].
- $\circ$  WI<sub>G,i</sub>( $\Delta$ ) [s] is the time integrated external ground dose rate weathering factor, for radionuclide i, integrated over the exposure period  $\Delta$ . The values are given for the exposure periods of 7 d and 1 a in Table 17.
- $F_{sf} = 0.4$  [unitless] is the fraction of the dose rate received from ground deposition for a person located inside a building relative to the dose rate for that person located outside the building (shielding factor). It is obtained from Ref. [5] and is consistent with Ref. [70].<sup>42</sup>
- $\circ$  F<sub>of</sub> = 0.6 [unitless] is the fraction of the exposure period that the person is assumed to spend inside a building (occupancy fraction). It is obtained from Ref. [5].
- e<sub>air-sh(grd-scenario),i</sub>(Δ) [Sv/(Bq/m<sup>2</sup>)] is the external effective dose to the representative person (i.e. the infant for this exposure scenario and pathway), over an exposure period Δ, from air shine resulting from resuspended ground deposition, for the 'ground' scenario, per unit ground surface activity of radionuclide i. It is determined by Eq. (24):

 $e_{air-sh(grd-scenario),i}(\Delta) = e_{air-sh,i}(adult) \times TI_{grd \rightarrow air,i}(\Delta) \times SF_{e_{ext}(adult \rightarrow infant)}$ (24) where:

<sup>&</sup>lt;sup>42</sup> The default of 0.4 is for a wooden framed house with the less effective shielding of the range of structures provided in Ref. [5]. The other structures have shielding factors between 0.2, for a one storey block house, to 0.01 for a large multi-storey structure. The default value is also higher than that estimated for most structures in Ref. [70].

- e<sub>air-sh,i</sub>(adult) [(Sv/s)/(Bq/m<sup>3</sup>)] is the external effective dose rate to the adult from air submersion, per unit volume activity of radionuclide i. The values are given in Table 24, and are applicable to an exposed individual standing at the centre of a semi-infinite, uniformly contaminated atmospheric cloud. This would overestimate the dose from a real (finite) plume of uniform concentration.
- $\circ$  TI<sub>grd→air,i</sub>( $\Delta$ ) [s/m] is the time integrated transfer factor of radionuclide i from the ground to the air (resuspended into the air by wind or other natural processes), integrated over the exposure period  $\Delta$ . The values are given for an exposure period of 7 d and 1 a in Table 17.
- $SF_{ext(adult \rightarrow infant)} = 1.4$  [unitless] is the scaling factor to convert the effective dose to the adult to the effective dose to the infant for external exposure, as discussed in Ref. [69].
- e<sub>inh-resusp(grd-scenario),i</sub>(Δ) [Sv/(Bq/m<sup>2</sup>)] is the committed effective dose to the representative person (i.e. the adult for this exposure scenario and pathway), over an exposure period Δ, from inhalation of resuspended ground deposition, for the 'ground' scenario, per unit ground surface activity of radionuclide i. It is determined by Eq. (25):

 $e_{inh-resusp(grd-scenario),i}(\Delta) = e_{inh,i}(adult) \times TI_{grd \to air,i}(\Delta) \times F_{rf} \times Q_{air}$ (25)

where:

- $\circ$  e<sub>inh,i</sub>(adult) [Sv/Bq] is the committed effective dose to the adult from inhalation, per unit intake of radionuclide i. The values are given in Table 24.
- $TI_{grd \rightarrow air,i}(\Delta)$  [s/m] is the time integrated transfer factor of radionuclide i from the ground to the air (resuspended into the air by wind or other natural processes), integrated over the exposure period  $\Delta$ . The values are given for an exposure period of 7 d and 1 a in Table 17.
- $\circ$  F<sub>rf</sub> = 1 [unitless] is the fraction of the airborne radioactive material that is deposited in the pulmonary region of the lung (respirable fraction). In reality F<sub>rf</sub> would be expected to be less than one tenth of this default value in most cases [71].
- $Q_{air} = 1.2 \text{ m}^3/\text{h} = 3.3\text{E-04 m}^3/\text{s}$  is the breathing rate of an adult performing light activity and is obtained from Ref. [72].
- e<sub>inadv-ing(grd-scenario),i</sub>(Δ) [Sv/(Bq/m<sup>2</sup>)] is the committed effective dose to the representative person (i.e. the infant for this exposure scenario and pathway), over an exposure period Δ, from inadvertent ingestion of ground deposition, for the 'ground' scenario, per unit ground surface activity of radionuclide i. It is determined by Eq. (26):

$$e_{\text{inadv-ing}(\text{grd-scenario}),i}(\Delta) = e_{\text{ing},i}(\text{infant}) \times \text{TI}_{\text{grd}\to\text{GI},i}(\Delta,\text{infant})$$
(26)

- $\circ$  e<sub>ing,i</sub>(infant) [Sv/Bq] is the committed effective dose to the infant from ingestion, per unit intake of radionuclide i. The values are given in Table 24.
- $TI_{grd \rightarrow Gl,i}(\Delta,infant) [m^2]$  is the time integrated transfer factor of radionuclide i from the ground to the GI tract by inadvertent ingestion for an infant (i.e. the representative person for this exposure scenario and pathway), integrated over the exposure period  $\Delta$ . The values for an exposure period of 7 d and 1 a for the infant are given in Table 19.

#### 3.6.1.2. Total equivalent dose to the fetus for the 'ground' scenario

Table 23 provides  $H_{fetus,grd-scenario,i}(\Delta)$  [Sv/(Bq/m<sup>2</sup>)], which is the total equivalent dose to the fetus over an exposure period  $\Delta^{43}$  for the 'ground' scenario, per unit ground surface activity of radionuclide i. It is determined by Eq. (27):

$$H_{fetus,grd-scenario,i}(\Delta) =$$

$$= h_{fetus,grd-sh(grd-scenario),i}(\Delta) + h_{fetus,air-sh(grd-scenario),i}(\Delta) + (27)$$

$$+ h_{fetus,inh-resusp(grd-scenario),i}(\Delta) + h_{fetus,inadv-ing(grd-scenario),i}(\Delta)$$

where:

h<sub>fetus,grd-sh(grd-scenario),i</sub>(Δ) [Sv/(Bq/m<sup>2</sup>)] is the external equivalent dose to the fetus, over an exposure period Δ, from ground shine resulting from ground deposition, for the 'ground' scenario, per unit ground surface activity of radionuclide i. It is determined by Eq. (28):

$$\begin{aligned} h_{fetus,grd-sh(grd-scenario),i}(\Delta) &= h_{red marrow,plane-srf,i}(adult) \times CorF_{grd} \times \\ &\times WI_{G,i}(\Delta) \times SF_{(h_{red marrow} \rightarrow h_{fetus}),grd-sh} \times (F_{sf} \times F_{of} + (1 - F_{of})) \end{aligned}$$
(28)

- h<sub>red marrow,plane-srf,i</sub>(adult) [(Sv/s)/(Bq/m<sup>2</sup>)] is the equivalent dose rate to the red marrow of the adult from an infinite smooth plane source, for radionuclide i.<sup>44</sup> The values are given in Table 25, and are applicable to an exposed individual standing on an ideal plane, uniformly contaminated surface. This would overestimate the dose from a real (non-plane) surface of uniform concentration; therefore, a ground roughness correction factor is used, as described in the next bullet.
- $CorF_{grd} = 0.7$  [unitless] is the ground roughness correction factor to account for the dose rate reduction due to ground roughness, as described in Ref. [69].
- $SF_{(hredmarrow \rightarrow hfetus),grd-sh} = 0.9$  [unitless] is the scaling factor to convert the dose conversion factor for the equivalent dose rate to the red marrow into equivalent dose rate to the fetus, for ground shine. It is based on the ratio between the dose to the uterus and the red marrow from Table II.6 of Ref. [69] for the photon energies of concern.
- $\circ$  WI<sub>G,i</sub>( $\Delta$ ) [s] is the time integrated external ground dose rate weathering factor, for radionuclide i, integrated over the exposure period  $\Delta$ . The values are given for the exposure periods of 7 d and 1 a in Table 17.
- $\circ$  F<sub>sf</sub> = 0.4 [unitless] is the fraction of the dose rate received from ground deposition for a person located inside a building relative to the dose rate for that person located outside

 $<sup>^{43}</sup>$  7 days and 1 year are used since those are the exposure periods provided for the corresponding generic criteria in the IAEA Safety Standards Series No. GSR Part 7 [1], as shown in Table 2. 7 days is used to calculate the early response action OIL1<sub>y</sub>, and 1 year is used to calculate the urgent response action OIL2<sub>y</sub>.

<sup>&</sup>lt;sup>44</sup> Due to the lack of dose conversion factors for the equivalent dose to the fetus from ground shine, the dose conversion factor for the equivalent dose to the red marrow from an infinite smooth plane source from Table III.3 of Ref. [69] is used to estimate the dose conversion factors for the equivalent dose to the fetus by using a scaling factor which is representative of the ratio between the dose to the uterus and the red marrow from Table II.6 of Ref. [69] for the photon energies of concern.

the building (shielding factor). It is obtained from Ref. [5] and is consistent with Ref. [70].<sup>45</sup>

- $\circ$  F<sub>of</sub> = 0.6 [unitless] is the fraction of the exposure period that the person is assumed to spend inside a building (occupancy fraction). It is obtained from Ref. [5].
- h<sub>fetus,air-sh(grd-scenario),i</sub>(Δ) [Sv/(Bq/m<sup>2</sup>)] is the external equivalent dose to the fetus, over an exposure period Δ, from air shine resulting from resuspended ground deposition, for the 'ground' scenario, per unit ground surface activity of radionuclide i. It is determined by Eq. (29):

 $h_{fetus,air-sh(grd-scenario),i}(\Delta) =$   $= h_{red marrow,air-sh,i}(adult) \times SF_{(h_{red marrow} \rightarrow h_{fetus}),air-sh} \times TI_{grd \rightarrow air,i}(\Delta)$ (29)

where:

- $\circ$  h<sub>red marrow,air-sh,i</sub>(adult) [(Sv/s)/(Bq/m<sup>3</sup>)] is the equivalent dose rate to the red marrow of the adult from submersion in a semi-infinite cloud of airborne radioactive material, per unit air activity concentration of radionuclide i.<sup>46</sup> The values are given in Table 25 and are applicable to an exposed individual standing at the centre of a semi-infinite, uniformly contaminated atmospheric cloud. This would overestimate the dose from a real (finite) plume of uniform concentration.
- SF<sub>(hredmarrow→hfetus),air-sh</sub> = 0.9 [unitless] is the scaling factor to convert the dose conversion factor for the equivalent dose rate to the red marrow into the equivalent dose rate to the fetus, for air shine. It is based on the ratio between the dose to the uterus and the red marrow from Table II.4 of Ref. [69] for the photon energies of concern.
- $TI_{grd \rightarrow air,i}(\Delta)$  [s/m] is the time integrated transfer factor of radionuclide i from the ground to the air (resuspended into the air by wind or other natural processes), integrated over the exposure period  $\Delta$ . The values are given for an exposure period of 7 d and 1 a in Table 17.
- h<sub>fetus,inh-resusp(grd-scenario),i</sub>(Δ) [Sv/(Bq/m<sup>2</sup>)] is the committed equivalent dose to the fetus, over an exposure period Δ, from inhalation of resuspended ground deposition by a pregnant woman, for the 'ground' scenario, per unit ground surface activity of radionuclide i. It is determined by Eq. (30):

 $h_{\text{fetus,inh-resusp(grd-scenario),i}}(\Delta) = h_{\text{fetus,inh,i}} \times TI_{\text{grd} \to \text{air,i}}(\Delta) \times F_{\text{rf}} \times Q_{\text{air}}$  (30) where:

<sup>&</sup>lt;sup>45</sup> The default of 0.4 is for a wooden framed house with the less effective shielding of the range of structures provided in Ref. [5]. The other structures have shielding factors between 0.2, for a one storey block house, to 0.01 for a large multi-storey structure. The default value is also higher than that estimated for most structures in Ref. [70].

<sup>&</sup>lt;sup>46</sup> Due to the lack of dose conversion factors for the equivalent dose to the fetus from air shine, the dose conversion factor for the equivalent dose to the red marrow from submersion in a semi-infinite cloud of airborne radioactive material from Table III.1. of Ref. [69] is used to estimate the dose conversion factors for the equivalent dose to the fetus by using a scaling factor which is representative of the ratio between the dose to the uterus and the red marrow from Table II.4 of Ref. [69] for the photon energies of concern.

- $\circ$  h<sub>fetus,inh,i</sub> [Sv/Bq] is the committed equivalent dose to the fetus from inhalation by an adult (i.e. pregnant woman), per unit intake of radionuclide i. The values are given in Table 25.
- $\circ$  TI<sub>grd→air,i</sub>( $\Delta$ ) [s/m] is the time integrated transfer factor of radionuclide i from the ground to the air (resuspended into the air by wind or other natural processes), integrated over the exposure period  $\Delta$ . The values are given for an exposure period of 7 d and 1 a in Table 17.
- $\circ$  F<sub>rf</sub> = 1 [unitless] is the fraction of the airborne radioactive material that is deposited in the pulmonary region of the lung (respirable fraction). In reality, F<sub>rf</sub> would be expected to be less than one tenth of this default value in most cases [71].
- $Q_{air} = 1.2 \text{ m}^3/\text{h} = 3.3\text{E-04 m}^3/\text{s}$  is the breathing rate of an adult performing light activity and is obtained from Ref. [72].
- h<sub>fetus,inadv-ing(grd-scenario),i</sub>(Δ) [Sv/(Bq/m<sup>2</sup>)] is the committed equivalent dose to the fetus, over an exposure period Δ, from inadvertent ingestion of ground deposition by an adult (i.e. pregnant woman), for the 'ground' scenario, per unit ground surface activity of radionuclide i. It is determined by Eq. (31):

 $h_{\text{fetus,inadv-ing}(\text{grd-scenario}),i}(\Delta) = h_{\text{fetus,ing},i} \times \text{TI}_{\text{grd}\to\text{GI},i}(\Delta, \text{adult})$ (31)

- h<sub>fetus,ing,i</sub> [Sv/Bq] is the committed equivalent dose to the fetus from ingestion by an adult (i.e. pregnant woman), per unit intake of radionuclide i. The values are given in Table 25.
- $TI_{grd \rightarrow GI,i}(\Delta, adult)$  [m<sup>2</sup>] is the time integrated transfer factor of radionuclide i from the ground to the GI tract by inadvertent ingestion for the adult (considered for fetal exposure), integrated over the exposure period  $\Delta$ . The values for an exposure period of 7 d and 1 a for the adult are given in Table 19.
TABLE 23. DOSE CONVERSION FACTORS FOR THE TOTAL EFFECTIVE DOSE TO THE REPRESENTATIVE PERSON AND FOR THE TOTAL EQUIVALENT DOSE TO THE FETUS FOR THE 'GROUND' SCENARIO

Dadianualida	${\rm E}_{ m grd-scenario,i}(7d)^{ m a}$	$E_{grd-scenario,i}(1a)^{a}$	H <sub>fetus,grd-scenario,i</sub> (7d) <sup>b,c</sup>	H <sub>fetus,grd-scenario,i</sub> (1a) <sup>b,d</sup>
Kaulonuchue	$[Sv/(Bq/m^2)]$	$[Sv/(Bq/m^2)]$	$[Sv/(Bq/m^2)]$	$[Sv/(Bq/m^2)]$
Rb-86	3.5E-11	1.3E-10	2.0E-11	8.3E-11
Sr-89	1.1E-11	2.4E-11	6.6E-11	1.1E-10
Sr-90+	1.6E-10	4.4E-10	1.8E-10	4.6E-10
Sr-91	2.1E-11	2.1E-11	1.3E-11	1.3E-11
Y-91	1.3E-11	4.1E-11	1.3E-12	1.3E-11
Zr-95+	8.7E-10	1.0E-08	5.4E-10	6.3E-09
Zr-97+	8.2E-11	8.2E-11	5.1E-11	5.1E-11
Mo-99+	4.8E-11	5.7E-11	2.9E-11	3.5E-11
Ru-103+	1.7E-10	1.3E-09	1.0E-10	8.0E-10
Ru-105	1.1E-11	1.1E-11	7.0E-12	7.0E-12
Ru-106+	1.4E-10	2.5E-09	5.4E-11	1.5E-09
Rh-105	8.8E-12	9.1E-12	5.2E-12	5.4E-12
Te-127m+	1.8E-11	1.3E-10	2.0E-11	7.2E-11
Te-127	2.2E-13	2.2E-13	9.8E-14	9.8E-14
Te-129m+	3.8E-11	2.0E-10	3.1E-11	1.3E-10
Te-131m	1.3E-10	1.3E-10	1.3E-10	1.3E-10
Te-132+	4.8E-10	6.1E-10	5.0E-10	5.9E-10
I-131	1.4E-10	2.7E-10	8.5E-10	1.1E-09
I-133	4.3E-11	4.4E-11	9.9E-11	9.9E-11
I-134	7.2E-12	7.2E-12	4.6E-12	4.6E-12
I-135	3.2E-11	3.2E-11	2.6E-11	2.6E-11
Cs-134	5.9E-10	1.9E-08	3.6E-10	1.2E-08
Cs-136	6.6E-10	2.1E-09	4.1E-10	1.3E-09
Cs-137+	2.6E-10	8.6E-09	1.4E-10	5.4E-09
Ba-140+	8.8E-10	2.7E-09	5.6E-10	1.7E-09
Ce-141	3.0E-11	1.8E-10	1.5E-11	9.8E-11
Ce-143	3.0E-11	3.1E-11	1.7E-11	1.8E-11
Ce-144+	7.4E-11	7.0E-10	1.8E-11	3.6E-10
Pr-143	3.5E-12	4.9E-12	1.1E-13	3.4E-13
Nd-147	4.5E-11	1.2E-10	2.4E-11	6.6E-11
Np-239	2.7E-11	3.1E-11	1.5E-11	1.7E-11
Pu-238	9.3E-08	2.2E-07	2.6E-09	6.2E-09
Pu-239	1.0E-07	2.4E-07	2.5E-09	5.8E-09
Pu-240	1.0E-07	2.4E-07	2.5E-09	5.8E-09
Pu-241	2.0E-09	4.6E-09	1.9E-12	4.4E-12
Am-241	8.2E-08	1.9E-07	1.4E-09	3.4E-09
Cm-242	5.0E-09	9.9E-09	9.3E-10	1.8E-09
Cm-244	4.8E-08	1.1E-07	1.4E-09	3.2E-09

<sup>a</sup>  $E_{grd-scenario,i}(\Delta)$  calculated based on Eq. (22). <sup>b</sup>  $H_{fetus,grd-scenario,i}(\Delta)$  calculated based on Eq. (27). <sup>c</sup> In this publication, the term 'fetus' encompasses both the embryo and the fetus.

d An exposure period of 1 year instead of the 9 months specified in Table 2 is used for simplicity.

+ The progenies that are in equilibrium with this parent radionuclide (as listed in Table 5) have been considered in the OIL calculations as being in equilibrium with the parent and do not need to be considered separately.

Radionuclida	e <sub>plane-srf,i</sub> (adult) <sup>a</sup>	e <sub>air-sh,i</sub> (adult) <sup>b</sup>	e <sub>inh,i</sub> (adult) <sup>c</sup>	e <sub>ing,i</sub> (infant) <sup>d</sup>
Kaulonuchue	$[(Sv/s)/(Bq/m^2)]$	$[(Sv/s)/(Bq/m^3)]$	[Sv/Bq]	[Sv/Bq]
Rb-86	9.3E-17	4.8E-15	9.3E-10	2.0E-08
Sr-89	2.3E-18	7.7E-17	7.9E-09	1.8E-08
Sr-90+	5.6E-18	2.0E-16	1.6E-07	9.3E-08
Sr-91	6.8E-16	3.5E-14	4.1E-10	4.0E-09
Y-91	5.7E-18	2.6E-16	8.9E-09	1.8E-08
Zr-95+	2.4E-15	1.2E-13	9.9E-09	1.3E-08
Zr-97+	1.5E-15	7.5E-14	9.7E-10	1.4E-08
Mo-99+	2.6E-16	1.3E-14	1.0E-09	3.6E-09
Ru-103+	4.6E-16	2.3E-14	3.0E-09	4.6E-09
Ru-105	7.7E-16	3.8E-14	1.8E-10	1.8E-09
Ru-106+	2.1E-16	1.0E-14	6.6E-08	4.9E-08
Rh-105	7.6E-17	3.7E-15	3.5E-10	2.7E-09
Te-127m+	1.6E-17	3.9E-16	9.9E-09	1.9E-08
Te-127	5.2E-18	2.4E-16	1.4E-10	1.2E-09
Te-129m+	7.7E-17	3.3E-15	7.9E-09	2.4E-08
Te-131m	1.4E-15	7.0E-14	9.4E-10	1.4E-08
Te-132+	2.4E-15	1.2E-13	2.1E-09	3.2E-08
I-131	3.8E-16	1.8E-14	7.4E-09	1.8E-07
I-133	6.0E-16	2.9E-14	1.5E-09	4.4E-08
I-134	2.5E-15	1.3E-13	5.5E-11	7.5E-10
I-135	1.5E-15	8.0E-14	3.2E-10	8.9E-09
Cs-134	1.5E-15	7.6E-14	2.0E-08	1.6E-08
Cs-136	2.1E-15	1.1E-13	2.8E-09	9.5E-09
Cs-137+	5.9E-16	2.9E-14	3.9E-08	1.2E-08
Ba-140+	2.8E-15	1.5E-13	7.1E-09	3.4E-08
Ce-141	7.4E-17	3.4E-15	3.8E-09	5.1E-09
Ce-143	2.8E-16	1.3E-14	8.3E-10	8.0E-09
Ce-144+	5.8E-17	2.8E-15	5.3E-08	3.9E-08
Pr-143	7.0E-19	2.1E-17	2.4E-09	8.7E-09
Nd-147	1.4E-16	6.2E-15	2.4E-09	7.8E-09
Np-239	1.6E-16	7.7E-15	1.0E-09	5.7E-09
Pu-238	8.4E-19	4.9E-18	1.1E-04	4.0E-07
Pu-239	3.7E-19	4.2E-18	1.2E-04	4.2E-07
Pu-240	8.0E-19	4.8E-18	1.2E-04	4.2E-07
Pu-241	1.9E-21	7.3E-20	2.3E-06	5.7E-09
Am-241	2.8E-17	8.2E-16	9.6E-05	3.7E-07
Cm-242	9.6E-19	5.7E-18	5.9E-06	7.6E-08
Cm-244	8.8E-19	4.2E-18	5.7E-05	2.9E-07

TABLE 24. DOSE CONVERSION FACTORS USED TO CALCULATE THE TOTAL EFFECTIVE DOSE TO THE REPRESENTATIVE PERSON FOR THE 'GROUND' SCENARIO

а eplane-srf.(adult) is equal to the dose conversion factors for exposure to contaminated ground surface given in Table III.3 of Ref. [69]. The reference uses the legacy unit 'effective dose equivalent', which is equivalent to the effective dose from external exposure.

b

 $e_{air-sh,i}(adult)$  is equal to the dose factors for air submersion given in Table III.1 of Ref. [69]. The reference uses the legacy unit 'effective dose equivalent', which is equivalent to the effective dose from external exposure.  $e_{inh,i}(adult)$  is the highest committed effective dose per unit intake for all absorption types and for members of the public older than 17 years from Table III.2E of Ref. [14]. c

d eing i(infant) is the value for members of the public with an age between 1 and 2 a from Table III.2D of Ref. [14].

+ The progenies that are in equilibrium with this parent radionuclide (as listed in Table 5) have been considered in the OIL calculations as being in equilibrium with the parent and do not need to be considered separately.

Dadianualida	h <sub>red marrow,plane-srf,i</sub> (adult) <sup>a</sup>	h <sub>red marrow,air-sh,i</sub> (adult) <sup>b</sup>	h <sub>fetus,inh,i</sub> c,d	h <sub>fetus,ing,i</sub> e
Kadionuciide	$[(Sv/s)/(Bq/m^2)]$	$[(Sv/s)/(Bq/m^3)]$	[Sv/Bq]	[Sv/Bq]
Rb-86	9.2E-17	4.6E-15	7.5E-10	2.2E-09
Sr-89	1.9E-18	6.4E-17	6.6E-08	1.3E-07
Sr-90+	4.8E-18	1.7E-16	1.7E-07	3.4E-07
Sr-91	6.6E-16	3.3E-14	3.6E-11	1.3E-10
Y-91	5.3E-18	2.4E-16	5.9E-11	1.7E-12
Zr-95+	2.3E-15	1.1E-13	3.0E-09	1.5E-09
Zr-97+	1.5E-15	7.1E-14	9.7E-11	3.2E-10
Mo-99+	2.5E-16	1.2E-14	8.4E-10	2.5E-09
Ru-103+	4.5E-16	2.1E-14	9.9E-10	3.6E-10
Ru-105	7.5E-16	3.6E-14	5.3E-11	6.0E-11
Ru-106+	2.1E-16	9.8E-15	6.0E-09	6.9E-10
Rh-105	7.3E-17	3.4E-15	1.9E-11	2.7E-11
Te-127m+	9.2E-18	2.9E-16	2.1E-08	7.5E-09
Te-127	4.9E-18	2.2E-16	1.5E-11	4.9E-12
Te-129m+	6.8E-17	3.1E-15	1.9E-08	6.8E-09
Te-131m	1.3E-15	6.7E-14	1.2E-07	5.4E-08
Te-132+	2.4E-15	1.2E-13	3.4E-07	1.4E-07
I-131	3.6E-16	1.7E-14	9.9E-07	1.1E-06
I-133	5.8E-16	2.8E-14	2.3E-07	2.5E-07
I-134	2.5E-15	1.3E-13	1.9E-09	1.9E-09
I-135	1.5E-15	7.8E-14	4.8E-08	5.2E-08
Cs-134	1.5E-15	7.2E-14	3.9E-09	1.1E-08
Cs-136	2.0E-15	1.0E-13	1.2E-09	3.5E-09
Cs-137+	5.7E-16	2.7E-14	2.5E-09	7.2E-09
Ba-140+	2.7E-15	1.4E-13	8.1E-09	1.2E-08
Ce-141	6.5E-17	2.8E-15	3.8E-10	6.0E-11
Ce-143	2.6E-16	1.2E-14	3.7E-11	8.9E-11
Ce-144+	5.3E-17	2.5E-15	5.7E-09	4.7E-11
Pr-143	5.2E-19	1.6E-17	7.5E-13	1.1E-14
Nd-147	1.2E-16	5.4E-15	1.9E-11	7.3E-11
Np-239	1.5E-16	6.5E-15	5.9E-10	9.2E-11
Pu-238	1.9E-19	1.7E-18	3.1E-06	6.3E-09
Pu-239	1.2E-19	2.7E-18	2.9E-06	5.9E-09
Pu-240	1.9E-19	1.7E-18	2.9E-06	5.9E-09
Pu-241	1.4E-21	5.6E-20	2.2E-09	4.5E-12
Am-241	1.7E-17	5.2E-16	1.6E-06	3.2E-09
Cm-242	2.3E-19	1.9E-18	1.1E-06	2.2E-09
Cm-244	2.0E-19	1.5E-18	1.6E-06	3.3E-09

TABLE 25. DOSE CONVERSION FACTORS USED TO CALCULATE THE TOTAL EQUIVALENT DOSE TO THE FETUS FOR THE 'GROUND' SCENARIO

<sup>a</sup> h<sub>red marrow,plane-srf,i</sub>(adult) is equal to the dose conversion factors for exposure to contaminated ground surface given in Table III.3 of Ref. [69]. The reference uses the legacy unit 'red marrow dose equivalent', which is the equivalent dose to the red marrow from external exposure.

 $^{b}$  h<sub>red marrow,air-sh,i</sub>(adult) is equal to the dose conversion factors for air submersion given in Table III.1 of Ref. [69].

<sup>c</sup> h<sub>fetus,inh,i</sub> is the value that gives the highest embryo or fetus organ dose for both acute and chronic inhalation, considering all absorption types and vapour (when applicable), for all times of inhalation (including inhalation prior to the pregnancy) for each radionuclide as given in Ref. [39]. For radionuclides for which dose conversion factors are not given in Ref. [39], the one year committed equivalent dose to the uterus of an adult from inhalation

of particulate aerosol (AMAD 1 micron) for all absorption types as given in Ref. [73] was considered as the committed equivalent dose to the fetus.

- <sup>d</sup> In this publication, the term 'fetus' encompasses both the embryo and the fetus.
- <sup>e</sup>  $h_{\text{fetus,ing,i}}$  is the value that gives the highest embryo or fetus organ dose for both acute and chronic ingestion, for all times of ingestion (including ingestion prior to the pregnancy) for each radionuclide as given in Ref. [39]. For radionuclides for which dose conversion factors are not given in Ref. [39], the one year committed equivalent dose to the uterus of an adult due to ingestion given in Ref. [73] was considered as the committed equivalent dose to the fetus.
- + The progenies that are in equilibrium with this parent radionuclide (as listed in Table 5) have been considered in the OIL calculations as being in equilibrium with the parent and do not need to be considered separately.

## 3.6.2. Dose conversion factors for the 'food pre-analysis' scenario

Table 26 provides the dose conversion factors for the committed effective dose to the representative person and the committed equivalent dose to the fetus for the 'food pre-analysis' scenario (needed in the calculation of  $OIL3_{\gamma}$ ), over an exposure period  $\Delta$  of 1 year. These dose conversion factors are determined as described in Sections 3.6.2.1 and 3.6.2.2.

## 3.6.2.1. Committed effective dose to the representative person for the 'food pre-analysis' scenario

Table 26 provides  $e_{ing,food-pre-analysis-scenario,i}$  [Sv/(Bq/m<sup>2</sup>)], which is the committed effective dose to the representative person from ingestion over 1 year, for the 'food pre-analysis' scenario, per unit ground surface activity of radionuclide i. It is determined by Eq. (32). For ingestion, the representative person is defined by a combination of the adult and infant dose conversion factors and consumption rates resulting in the highest dose as described in Table 13.

$$e_{ing,food-pre-analysis-scenario,i} = [\phi_{1} \times F_{milk} \times U_{cow} \times F_{cow-feed} \times T_{feed \rightarrow cow-milk,i} \times Max \begin{cases} (Q_{milk}(infant) \times e_{ing,i}(infant)), \\ , (Q_{milk}(adult) \times e_{ing,i}(adult)) \end{cases} + Max \begin{cases} (Q_{lv}(infant) \times e_{ing,i}(infant)), \\ , (Q_{lv}(adult) \times e_{ing,i}(adult)) \end{cases} \times (32) \times ($$

- $\Phi_1 = 3 \text{ m}^2/\text{kg}$  is the mass interception factor for pasture grass in dry weight, i.e. ratio between the activity concentration on the plant [Bq/kg] and the unit ground surface activity on the terrestrial surface [Bq/m<sup>2</sup>] (soil plus vegetation). It is described in Section 3.5.4.
- $\Phi_2 = 0.3 \text{ m}^2/\text{kg}$  is the mass interception factor for leafy vegetables (fresh or wet weight), i.e. ratio between the activity concentration on the plant [Bq/kg] and the unit ground surface activity on the terrestrial surface [Bq/m<sup>2</sup>] (soil plus vegetation). It is described in Section 3.5.4.
- $F_{milk} = 0.5$  [unitless] is the fraction of the milk consumed assumed to be affected before actions are taken to control intake.
- $F_{1v} = 0.5$  [unitless] is the fraction of the leafy vegetables consumed assumed to be affected before actions are taken to control intake.
- $U_{cow} = 16 \text{ kg/d} = 1.9\text{E-04 kg/s}$  is the cow consumption rate of feed in dry weight. It is obtained from Table XII of Ref. [64].

- $F_{cow-feed} = 0.7$  [unitless] is the fraction of the cow feed assumed to be affected. It is obtained from Ref. [74]<sup>47</sup>.
- T<sub>feed→cow-milk,i</sub> [s/L] is the transfer factor of radionuclide i from feed to the milk of the cow. The values are given in Table 21.
- Q<sub>milk</sub>(infant) = 120 L/a = 3.8E-06 L/s is the consumption rate of milk by infants (considered for the representative person). It is obtained from Table 13 in annex B of Ref. [20].
- Q<sub>milk</sub>(adult) = 105 L/a = 3.3E-06 L/s is the consumption rate of milk by adults (considered for the fetal exposure). It is obtained from Table 13 in annex B of Ref. [20].
- $Q_{lv}(infant) = 20 \text{ kg/a} = 6.3\text{E-07 kg/s}$  is the consumption rate of leafy vegetables in fresh weight by infants (i.e. the representative person). It is obtained from Table 13 in annex B of Ref. [20].
- $Q_{lv}(adult) = 60 \text{ kg/a} = 1.9\text{E}-06 \text{ kg/s}$  is the consumption rate of leafy vegetables in fresh weight by adults (considered for the fetus). It is obtained from Table 13 in annex B of Ref. [20].
- e<sub>ing,i</sub>(infant) [Sv/Bq] is the committed effective dose to the infant from ingestion, per unit intake of radionuclide i. The values are given in Table 27.
- e<sub>ing,i</sub>(adult) [Sv/Bq] is the committed effective dose to the adult from ingestion, per unit intake of radionuclide i. The values are given in Table 27.
- $\Delta_{\text{eff,OIL3,i}}$  [s] is the effective availability period for radionuclide i for ingestion in the 'food pre-analysis' scenario. The values are given in Table 20.
- F<sub>consumption,i</sub> [unitless] is the fraction of the radionuclide i remaining at the time of human consumption. The values are given in Table 22.

# 3.6.2.2. Committed equivalent dose to the fetus for the 'food pre-analysis' scenario

Table 26 provides  $h_{\text{fetus,ing,food-pre-analysis-scenario,i}}$  [Sv/(Bq/m<sup>2</sup>)], which is the committed equivalent dose to the fetus from ingestion of food, milk and drinking water by the pregnant woman, over the period of in utero development, for the 'food pre-analysis' scenario, per unit ground surface activity of radionuclide i. It is determined by Eq. (33):

$$\begin{split} h_{fetus,ing,food-pre-analysis-scenario,i} = \\ &= [\varphi_1 \times F_{milk} \times U_{cow} \times F_{cow-feed} \times T_{feed \rightarrow cow-milk,i} \times Q_{milk}(adult) \times h_{fetus,ing,i} + (33) \\ &+ \varphi_2 \times F_{lv} \times Q_{lv}(adult) \times h_{fetus,ing,i}] \times \Delta_{eff,OIL3,i} \times F_{consumption,i} \end{split}$$

where:

- h<sub>fetus,ing,i</sub> [Sv/Bq], which is the committed equivalent dose to the fetus from ingestion by an adult (i.e. pregnant woman), per unit intake of radionuclide i. The values are given in Table 27.
- The other factors are described above.

The contribution to the dose from ingestion of affected drinking water is not included in these calculations, because it is very hard to make a reasonable estimation of the dose received. However, since it could be a significant contributor to the dose, whenever  $OIL3_{\gamma}$  is exceeded, as a precaution, the consumption and distribution of directly collected rainwater is restricted as well.

<sup>&</sup>lt;sup>47</sup> Ref. [74] assumes for the average backyard cow an average dry matter intake of 11 kg/d, from which the pasture intake is 8 kg/d (dry mass), i.e. approximately a fraction of 0.7.

Radionuclide	$e_{ing,food-pre-analysis-scenario,i}^{a}$ $[Sv/(Bq/m^2)]$	b,c,d h <sub>fetus,ing,food-pre-analysis-scenario,i</sub> [Sv/(Bq/m <sup>2</sup> )]
Rb-86	1.2E-07	1.2E-08
Sr-89	4.3E-09	6.3E-08
Sr-90+	2.9E-08	2.1E-07
Sr-91	6.2E-12	4.0E-13
Y-91	2.5E-09	6.8E-13
Zr-95+	1.7E-09	6.2E-10
Zr-97+	4.2E-11	2.8E-12
Mo-99+	1.3E-10	1.9E-10
Ru-103+	5.6E-10	1.3E-10
Ru-105	9.3E-14	9.2E-15
Ru-106+	7.9E-09	3.3E-10
Rh-105	3.6E-11	8.8E-13
Te-127m+	3.4E-09	3.5E-09
Te-127	1.1E-12	1.2E-14
Te-129m+	3.4E-09	2.5E-09
Te-131m	1.3E-10	1.4E-09
Te-132+	9.9E-10	1.1E-08
I-131	4.6E-08	3.8E-07
I-133	8.9E-10	6.7E-09
I-134	8.3E-21	2.8E-20
I-135	1.0E-11	8.1E-11
Cs-134	1.8E-08	1.0E-08
Cs-136	3.0E-09	1.5E-09
Cs-137+	1.3E-08	6.8E-09
Ba-140+	2.8E-09	2.7E-09
Ce-141	5.9E-10	2.1E-11
Ce-143	7.3E-11	2.4E-12
Ce-144+	6.3E-09	2.2E-11
Pr-143	6.8E-10	2.6E-15
Nd-147	5.4E-10	1.5E-11
Np-239	1.1E-10	5.0E-12
Pu-238	1.1E-07	3.1E-09
Pu-239	1.2E-07	2.9E-09
Pu-240	1.2E-07	2.9E-09
Pu-241	2.4E-09	2.2E-12
Am-241	1.0E-07	1.6E-09
Cm-242	1.2E-08	1.0E-09
Cm-244	6.0E-08	1.6E-09

# TABLE 26. DOSE CONVERSION FACTORS FOR THE 'FOOD PRE-ANALYSIS' SCENARIO

<sup>a</sup> e<sub>ing,food-pre-analysis-scenario,i</sub> calculated based on Eq. (32).
 <sup>b</sup> h<sub>fetus,ing,food-pre-analysis-scenario,i</sub> calculated based on Eq. (33).
 <sup>c</sup> In this publication, the term 'fetus' encompasses both the embryo and the fetus.
 <sup>d</sup> An exposure period of 1 year instead of the 9 months specified in Table 2 is used for simplicity.

+ The progenies that are in equilibrium with this parent radionuclide (as listed in Table 5) have been considered in the OIL calculations as being in equilibrium with the parent and do not need to be considered separately

TABLE 27. DOSE CONVERSION FACTORS USED TO CALCULATE THE DOSE CONVERSION FACTORS FOR THE 'FOOD PRE-ANALYSIS', 'FOOD POST-ANALYSIS' AND 'SKIN' SCENARIOS

Dadianualida	e <sub>ing,i</sub> (infant) <sup>a</sup>	e <sub>ing,i</sub> (adult) <sup>b</sup>	h <sub>fetus,ing,i</sub> c,d	ad <sub>skin,skin-srf,i</sub> e
Kaulolluchue	[Sv/Bq]	[Sv/Bq]	[Sv/Bq]	$[(Gy/s)/(Bq/m^2)]$
Rb-86	2.0E-08	2.8E-09	2.2E-09	3.3E-14
Sr-89	1.8E-08	2.6E-09	1.3E-07	3.2E-14
Sr-90+	9.3E-08	3.1E-08	3.4E-07	5.0E-14
Sr-91	4.0E-09	6.5E-10	1.3E-10	3.2E-14
Y-91	1.8E-08	2.4E-09	1.7E-12	3.2E-14
Zr-95+	1.3E-08	2.2E-09	1.5E-09	5.9E-15
Zr-97+	1.4E-08	2.2E-09	3.2E-10	6.5E-14
Mo-99+	3.6E-09	6.2E-10	2.5E-09	2.4E-14
Ru-103+	4.6E-09	7.3E-10	3.6E-10	1.1E-15
Ru-105	1.8E-09	2.6E-10	6.0E-11	2.6E-14
Ru-106+	4.9E-08	7.0E-09	6.9E-10	4.5E-14
Rh-105	2.7E-09	3.7E-10	2.7E-11	7.3E-15
Te-127m+	1.9E-08	2.5E-09	7.5E-09	1.4E-14
Te-127	1.2E-09	1.7E-10	4.9E-12	1.4E-14
Te-129m+	2.4E-08	3.0E-09	6.8E-09	3.1E-14
Te-131m	1.4E-08	1.9E-09	5.4E-08	7.1E-15
Te-132+	3.2E-08	4.1E-09	1.4E-07	3.0E-14
I-131	1.8E-07	2.2E-08	1.1E-06	1.1E-14
I-133	4.4E-08	4.3E-09	2.5E-07	2.6E-14
I-134	7.5E-10	1.1E-10	1.9E-09	3.4E-14
I-135	8.9E-09	9.3E-10	5.2E-08	2.2E-14
Cs-134	1.6E-08	1.9E-08	1.1E-08	1.1E-14
Cs-136	9.5E-09	3.0E-09	3.5E-09	6.4E-15
Cs-137+	1.2E-08	1.3E-08	7.2E-09	1.5E-14
Ba-140+	3.4E-08	5.0E-09	1.2E-08	5.7E-14
Ce-141	5.1E-09	7.1E-10	6.0E-11	6.0E-15
Ce-143	8.0E-09	1.1E-09	8.9E-11	2.6E-14
Ce-144+	3.9E-08	5.3E-09	4.7E-11	4.3E-14
Pr-143	8.7E-09	1.2E-09	1.1E-14	2.0E-14
Nd-147	7.8E-09	1.1E-09	7.3E-11	1.4E-14
Np-239	5.7E-09	8.0E-10	9.2E-11	5.0E-15
Pu-238	4.0E-07	2.3E-07	6.3E-09	6.0E-17
Pu-239	4.2E-07	2.5E-07	5.9E-09	2.3E-17
Pu-240	4.2E-07	2.5E-07	5.9E-09	5.7E-17
Pu-241	5.7E-09	4.8E-09	4.5E-12	3.6E-20
Am-241	3.7E-07	2.0E-07	3.2E-09	3.7E-16
Cm-242	7.6E-08	1.2E-08	2.2E-09	5.4E-17
Cm-244	2.9E-07	1.2E-07	3.3E-09	5.1E-17

<sup>a</sup> e<sub>ing,i</sub>(infant) is the value for members of the public with an age between 1 and 2 a from Table III.2D of Ref. [14].
 <sup>b</sup> e<sub>ing,i</sub>(adult) is the value for members of the public with an age > 17 a from Table III.2D of Ref. [14].
 <sup>c</sup> h<sub>fetus,ing,i</sub> is the value that gives the highest embryo or fetus organ dose for both acute and chronic ingestion, for all times of ingestion (including ingestion prior to the pregnancy) for each radionuclide as given in Ref. [39]. For

radionuclides for which dose conversion factors are not given in Ref. [39], the one year committed equivalent dose to the uterus of an adult due to ingestion given in Ref. [73] was considered as the committed equivalent dose to the fetus.

- <sup>d</sup> In this publication the term 'fetus' encompasses both the embryo and the fetus.
- <sup>e</sup> ad<sub>skin,skin-srf,i</sub> is calculated following the method described in Appendix VI.2 of Ref. [75] (this method is used instead of the values given in Table 19 of Ref. [75], because inconsistencies were found in Table 19 of Ref. [75]) using the spectral data from the CD accompanying Ref. [52]. The dose is to 100 cm<sup>2</sup> of the dermis (skin structures at a depth 0.4 mm below the surface). Severe damage to the derma over this area results in severe deterministic effects [19, 75, 76].
- + The progenies that are in equilibrium with this parent radionuclide (as listed in Table 5) have been considered in the OIL calculations as being in equilibrium with the parent and do not need to be considered separately.

## **3.6.3.** Dose conversion factors for the 'skin' scenario

Table 28 provides the dose conversion factors for the 'skin' scenario (needed in the calculation of  $OIL4_{\gamma}$  and  $OIL4_{\beta}$ ), for the committed effective dose to the representative person and the committed equivalent dose to the fetus from inadvertent ingestion, and the RBE weighted absorbed dose from external exposure to the skin derma of the representative person. These dose conversion factors are determined as described in Sections 3.6.3.1, 3.6.3.2 and 3.6.3.3.

# 3.6.3.1. Committed effective dose to the representative person for the 'skin' scenario

Table 28 provides  $e_{ing,skin-scenario,i}$  [Sv/(Bq/m<sup>2</sup>)], which is the committed effective dose to the representative person (i.e. the infant for this exposure scenario and pathway) from inadvertent ingestion, for the 'skin' scenario, per unit skin surface activity of radionuclide i. It is determined by Eq. (34):

$$e_{\text{ing.skin-scenario.i}} = e_{\text{ing.i}}(\text{infant}) \times T_{\text{skin} \to \text{GI}}(\text{infant}) \times \Delta_{\text{eff.OIL4-Urgent.i}}$$
(34)

where:

- e<sub>ing,i</sub>(infant) [Sv/Bq] is the committed effective dose to the infant from ingestion, per unit intake of radionuclide i. The values are given in Table 27.
- $T_{skin \rightarrow GI}(infant) = 6.4E-08 \text{ m}^2/\text{s}$  is the transfer factor from the skin to the GI tract due to inadvertent ingestion for the representative person (i.e. the infant for this exposure scenario and pathway). It is described in Section 3.5.8.
- $\Delta_{\text{eff,OIL4-Urgent,i}}$  [s] is the effective availability period for radionuclide i for inadvertent ingestion from the skin in the 'skin' scenario for the urgent generic criteria. The values are given in Table 20.

## 3.6.3.2. Committed equivalent dose to the fetus for the 'skin' scenario

Table 28 provides  $h_{fetus,ing,skin-scenario,i}$  [Sv/(Bq/m<sup>2</sup>)], which is the committed equivalent dose to the fetus from inadvertent ingestion of radioactive material on the skin, for the 'skin' scenario, per unit skin surface activity of radionuclide i. It is determined by Eq. (35):

$$h_{\text{fetus,ing,skin-scenario,i}} = h_{\text{fetus,ing,i}} \times T_{\text{skin} \to \text{GI}}(\text{adult}) \times \Delta_{\text{eff,OIL4-Urgent,i}}$$
(35)

where:

• h<sub>fetus,ing,i</sub> [Sv/Bq] is the committed equivalent dose to the fetus from ingestion by an adult (i.e. pregnant woman), per unit intake of radionuclide i. The values are given in Table 27.

- $T_{skin \rightarrow GI}(adult) = 3.2E-08 \text{ m}^2/\text{s}$  is the transfer factor from the skin to the GI tract due to inadvertent ingestion by the pregnant woman for fetal exposure. It is described in Section 3.5.8.
- $\Delta_{\text{eff,OIL4-Urgent,i}}$  [s] is the effective availability period for radionuclide i for inadvertent ingestion from the skin in the 'skin' scenario for the urgent generic criteria. The values are given in Table 20.

# 3.6.3.3. RBE weighted absorbed dose to the skin derma of the representative person for the 'skin' scenario

Table 28 provides  $ad_{skin,skin-scenario,i}$  [Gy/(Bq/m<sup>2</sup>)], which is the RBE weighted absorbed dose to the skin derma of the representative person,<sup>48</sup> for the 'skin' scenario, per unit skin surface activity of radionuclide i. It is determined by Eq. (36):

 $ad_{skin,skin-scenario,i} = ad_{skin,skin-srf,i} \times \Delta_{eff,OIL4-Acute,i}$ 

where:

- ad<sub>skin,skin-srf,i</sub> [(Gy/s)/(Bq/m<sup>2</sup>)] is the RBE weighted absorbed dose rate to the skin derma of the representative person from activity on the skin, per unit skin surface activity of radionuclide i. The values are given in Table 27.
- $\Delta_{eff,OIL4-Acute,i}$  [s] is the effective availability period for radionuclide i for inadvertent ingestion from the skin in the 'skin' scenario for the acute generic criteria. The values are given in Table 20.

(36)

<sup>&</sup>lt;sup>48</sup> The RBE weighted absorbed dose rate to the skin derma is considered to be age independent.

Dediamaki	eing,skin-scenario,i	b,c h <sub>fetus,ing,skin-scenario,i</sub>	ad <sub>skin,skin-scenario,i</sub> d
Kadionuciide	$[Sv/(Bq/m^2)]$	$[Sv/(Bq/m^2)]$	$[Gy/(Bq/m^2)]$
Rb-86	9.5E-11	5.2E-12	9.5E-10
Sr-89	8.7E-11	3.2E-10	9.1E-10
Sr-90+	4.6E-10	8.3E-10	1.4E-09
Sr-91	7.8E-12	1.3E-13	6.8E-10
Y-91	8.7E-11	4.1E-15	9.2E-10
Zr-95+	6.1E-11	3.7E-12	1.7E-10
Zr-97+	3.8E-11	4.2E-13	1.6E-09
Mo-99+	1.5E-11	4.9E-12	6.5E-10
Ru-103+	2.2E-11	8.7E-13	3.1E-11
Ru-105	2.0E-12	3.4E-14	4.1E-10
Ru-106+	2.4E-10	1.7E-12	1.3E-09
Rh-105	9.4E-12	4.7E-14	1.9E-10
Te-127m+	9.4E-11	1.8E-11	4.1E-10
Te-127	2.3E-12	4.7E-15	2.8E-10
Te-129m+	1.2E-10	1.6E-11	8.8E-10
Te-131m	4.6E-11	8.9E-11	1.8E-10
Te-132+	1.3E-10	2.9E-10	8.4E-10
I-131	8.2E-10	2.5E-09	3.1E-10
I-133	1.3E-10	3.6E-10	6.4E-10
I-134	2.1E-13	2.6E-13	1.4E-10
I-135	1.3E-11	3.9E-11	4.0E-10
Cs-134	7.8E-11	2.7E-11	3.2E-10
Cs-136	4.5E-11	8.2E-12	1.8E-10
Cs-137+	5.9E-11	1.8E-11	4.3E-10
Ba-140+	1.6E-10	2.8E-11	1.6E-09
Ce-141	2.5E-11	1.4E-13	1.7E-10
Ce-143	2.7E-11	1.5E-13	6.7E-10
Ce-144+	1.9E-10	1.2E-13	1.2E-09
Pr-143	4.1E-11	2.6E-17	5.8E-10
Nd-147	3.6E-11	1.7E-13	4.1E-10
Np-239	2.2E-11	1.8E-13	1.3E-10
Pu-238	2.0E-09	1.5E-11	1.7E-12
Pu-239	2.1E-09	1.4E-11	6.7E-13
Pu-240	2.1E-09	1.4E-11	1.6E-12
Pu-241	2.8E-11	1.1E-14	1.0E-15
Am-241	1.8E-09	7.9E-12	1.1E-11
Cm-242	3.7E-10	5.4E-12	1.5E-12
Cm-244	1.4E-09	8.1E-12	1.5E-12

TABLE 28. DOSE CONVERSION FACTORS FOR THE 'SKIN' SCENARIO

<sup>a</sup> e<sub>ing,skin-scenario,i</sub> calculated based on Eq. (34).
 <sup>b</sup> h<sub>fetus,ing,skin-scenario,i</sub> calculated based on Eq. (35).
 <sup>c</sup> In this publication the term 'fetus' encompasses both the embryo and the fetus.

<sup>d</sup> ad<sub>skin,skin-scenario,i</sub> calculated based on Eq. (36). + The progenies that are in equilibrium with this parent radionuclide (as listed in Table 5) have been considered in the OIL calculations as being in equilibrium with the parent and do not need to be considered separately.

## 3.6.4. Dose conversion factors for the 'food post-analysis' scenario

Table 29 provides the dose conversion factors for the committed effective dose to the representative person and the committed equivalent dose to the fetus for the 'food post-analysis' scenario (needed in the calculation of OIL7), over an exposure period  $\Delta$  of 1 year. These dose conversion factors are determined as described in Sections 3.6.4.1 and 3.6.4.2.

## 3.6.4.1. Committed effective dose to the representative person for the 'food post-analysis' scenario

Table 29 provides  $e_{ing,food-post-analysis-scenario,i}$  [Sv/(Bq/kg)], which is the committed effective dose to the representative person from ingestion of food, milk and drinking water over 1 year, for the 'food post-analysis' scenario, per unit activity concentration of radionuclide i in food, milk or drinking water. It is determined by Eq. (37). For ingestion, the representative person is defined by a combination of the adult and infant dose conversion factors and consumption rates, resulting in the highest dose as described in Table 15.

$$e_{ing,food-post-analysis-scenario,i} = \\ = Max \begin{cases} \left( Q_{diet}(infant) \times e_{ing,i}(infant) \right), \\ , \left( Q_{diet}(adult) \times e_{ing,i}(adult) \right) \end{cases} \times \Delta_{eff,OIL7,i} \times F_{diet,OIL7} \times F_{prep} \end{cases}$$
(37)

where:

- e<sub>ing,i</sub>(infant) [Sv/Bq] is the committed effective dose to the infant from ingestion, per unit intake of radionuclide i. The values are given in Table 27.
- e<sub>ing,i</sub>(adult) [Sv/Bq] is the committed effective dose to the adult from ingestion, per unit intake of radionuclide i. The values are given in Table 27.
- Q<sub>diet</sub>(infant) = 415 kg/a = 1.6E-05 kg/s is the consumption rate of food, milk and drinking water (total diet) by infants (considered for the representative person). It is obtained from Table 13 of annex B of Ref. [20].
- Q<sub>diet</sub>(adult) = 1040 kg/a = 3.2E-05 kg/s is the consumption rate of food, milk and drinking water (total diet) by adults (considered for the fetal exposure). It is obtained from Table 13 of annex B of Ref. [20].
- Δ<sub>eff,OIL7,i</sub> [s] is the effective availability period for radionuclide i for ingestion in the 'food post-analysis' scenario. The values are given in Table 20.
- $F_{diet,OIL7} = 0.5$  [unitless] fraction of the total diet (i.e. all food, milk and drinking water consumed) assumed to be affected for the 'food post-analysis' scenario. In reality, it is expected to be smaller, considering the diverse components and sources of the diet.
- $F_{prep} = 1$  [unitless] is the fraction of the radioactive material remaining for ingestion after preparations. It assumes no reduction due to normal preparations/processing (e.g. washing and cooking), which, in some cases, substantially reduces the concentration of radionuclides in the food [65].

## 3.6.4.2. Committed equivalent dose to the fetus for the 'food post-analysis' scenario

Table 29 provides  $h_{fetus,ing,food-post-analysis-scenario,i}$  [Sv/(Bq/kg)], which is the committed equivalent dose to the fetus from ingestion of food, milk and drinking water by the pregnant woman over 1 year, for the 'food post-analysis' scenario, per unit activity concentration of radionuclide i in food, milk or drinking water. It is determined by Eq. (38):

 $h_{fetus,ing,food-post-analysis-scenario,i} =$   $= [h_{fetus,ing,i} \times Q_{diet}(adult) \times \Delta_{eff,OIL7,i} \times F_{diet,OIL7} \times F_{prep}$ (38)

where:

- h<sub>fetus,ing,i</sub> [Sv/Bq] is the committed equivalent dose to the fetus from ingestion by an adult (i.e. pregnant woman), per unit intake of radionuclide i. The values are given in Table 27.
- The other factors are described above.

## 3.6.5. Dose conversion factors for the 'thyroid' scenario

 $h_{thyroid,I-131-thyroid-burden} = 1.2E-05$  Sv/Bq is the committed equivalent dose to the thyroid of an infant from I-131 in the thyroid (thyroid burden)<sup>49</sup> (needed in the calculation of OIL8). No specific values for  $h_{thyroid,I-131-thyroid-burden}$  could be found, and therefore it was estimated based on the assumption that approximately 1/3 of the I-131 intake is retained in the thyroid, as described in Eq. (39). Although the equivalent dose to the thyroid from ingestion was used in Eq. (39), using dose conversion factors for inhalation would lead to the same result.

 $h_{\text{thyroid},I-131-\text{thyroid}-\text{burden}} = h_{\text{thyroid},\text{ing}-I-131} / F_{I-131-\text{retained}-\text{in}-\text{thyroid}}$ (39)

- F<sub>I-131-retained-in-thyroid</sub> = 0.3 [unitless] is the fraction of I-131 that is retained in the thyroid following intake. The value is based on Table XII-C6-3 of Ref. [77], which provides similar values for different age groups and times after intake.
- $h_{thyroid,ing-I-131} = 3.6E-06$  Sv/Bq is the committed equivalent dose to the thyroid from ingestion of I-131 by the infant, per unit intake of I-131. It is obtained from Ref. [73].

<sup>&</sup>lt;sup>49</sup> The contribution to the thyroid dose from short lived iodine and Te-132 were not included in the calculations, because they are not major contributors to the thyroid dose relative to I-131. The resulting OIL will be conservative, because the measured ambient dose equivalent rate is assumed to only result from I-131 (which contributes a much higher dose to the thyroid relative to its ambient equivalent dose rate).

Radionuclide	e <sub>ing,food-post-analysis-scenario,i</sub> a [Sv/(Ba/kg)]	h <sub>fetus,ing,food-post-analysis-scenario,i</sub> b,c,d [Sv/(Ba/kg)]
Rb-86	3 1E-07	8 4E-08
Sr-89	7 4E-07	1 3E-05
Sr-90+	1 9E-05	1 7E-04
Sr-91	1 3E-09	1 1E-10
Y-91	8.5E-07	2.0E-10
Zr-95+	6.5E-07	2.0E-07
Zr-97+	8.3E-09	4.6E-10
Mo-99+	8.2E-09	1.4E-08
Ru-103+	1.5E-07	2.9E-08
Ru-105	2.7E-10	2.3E-11
Ru-106+	7.4E-06	2.6E-07
Rh-105	3.3E-09	8.2E-11
Te-127m+	1.5E-06	1.5E-06
Te-127	3.8E-10	3.9E-12
Te-129m+	6.7E-07	4.7E-07
Te-131m	1.4E-08	1.4E-07
Te-132+	8.5E-08	9.4E-07
I-131	1.2E-06	1.8E-05
I-133	3.1E-08	4.5E-07
I-134	2.2E-11	1.4E-10
I-135	2.0E-09	2.9E-08
Cs-134	8.4E-06	4.9E-06
Cs-136	1.0E-07	9.5E-08
Cs-137+	6.7E-06	3.7E-06
Ba-140+	3.5E-07	3.1E-07
Ce-141	1.4E-07	4.0E-09
Ce-143	9.0E-09	2.5E-10
Ce-144+	5.4E-06	1.6E-08
Pr-143	9.7E-08	3.1E-13
Nd-147	7.0E-08	1.6E-09
Np-239	1.1E-08	4.5E-10
Pu-238	1.2E-04	3.3E-06
Pu-239	1.3E-04	3.1E-06
Pu-240	1.3E-04	3.1E-06
Pu-241	2.4E-06	2.3E-09
Am-241	1.0E-04	1.7E-06
Cm-242	8.0E-06	5.8E-07
Cm-244	6.1E-05	1.7E-06

# TABLE 29. DOSE CONVERSION FACTORS FOR THE 'FOOD POST-ANALYSIS' SCENARIO

<sup>a</sup> e<sub>ing,food-post-analysis-scenario,i</sub> is calculated based on Eq. (37).
 <sup>b</sup> h<sub>fetus,ing,food-post-analysis-scenario,i</sub> is calculated based on Eq. (38).
 <sup>c</sup> In this publication, the term 'fetus' encompasses both the embryo and the fetus.
 <sup>d</sup> An exposure period of 1 year instead of the 9 months specified in Table 2 is used for simplicity.

+ The progenies that are in equilibrium with this parent radionuclide (as listed in Table 5) have been considered in the OIL calculations as being in equilibrium with the parent and do not need to be considered separately.

#### 3.7. INSTRUMENT RESPONSE

OILs are operational criteria intended to be used with monitoring results provided by instruments. This section describes how the response of the instruments was considered in the OIL calculations.

## 3.7.1. Ambient dose equivalent rate

It is assumed that all ambient dose equivalent rate instruments respond similarly while measuring the ambient dose equivalent rate H\*(10), with the response depending on the monitored surface or object, the monitoring procedures and the conditions encountered. OILs are provided for the ambient dose equivalent rate as measured by a field survey instrument for emissions from radioactive material deposited on the ground (OIL1<sub> $\gamma$ </sub>, OIL2<sub> $\gamma$ </sub>, OIL3<sub> $\gamma$ </sub>), radioactive material deposited on the skin (OIL4<sub> $\gamma$ </sub>) and radioactive material in the thyroid (OIL8<sub> $\gamma$ </sub>). The basis for the ambient dose equivalent rate for these OILs (as measured at the recommended distance) is described below.

## 3.7.1.1. Ground monitoring for $OIL1_{\gamma}$ , $OIL2_{\gamma}$ and $OIL3_{\gamma}$

The ambient dose equivalent rate at 1 m above ground level, per unit ground surface activity of radionuclide i (i.e.  $H_{grd-sh,i}^*[(Sv/s)/(Bq/m^2)]$ ) is given in Table 30 and determined by Eq. (40):

$$H^*_{grd-sh,i} = e_{plane-srf,i}(adult) \times CorF_{grd} \times SF_{e \to H^*}$$
(40)

where:

- e<sub>plane-srf,i</sub>(adult) [(Sv/s)/(Bq/m<sup>2</sup>)] is the effective dose rate to the adult from an infinite smooth plane source, per unit surface activity of radionuclide i. The values are given in Table 24.
- CorF<sub>grd</sub> = 0.7 [unitless] is the ground roughness correction factor to account for the dose rate reduction due to ground roughness, as described in Ref. [69].
- $SF_{e \rightarrow H^*} = 1.4$  [unitless] is the scaling factor to convert the effective dose into ambient dose equivalent for the photon energies of concern. It is obtained from Section 5.3.2 of Ref. [78].

#### 3.7.1.2. Skin monitoring for $OIL4_{\gamma}$

The ambient dose equivalent rate at 10 cm from activity on 100 cm<sup>2</sup> of the skin surface, per unit surface activity of radionuclide i (i.e.  $H_{skin-100cm-srf,i}^2 [(Sv/s)/(Bq/m^2)]$ ) is given in Table 30 and determined by Eq. (41):

$$H^*_{skin-100cm^2-srf,i} = \sum_{j} Y^{\gamma}_{E_{j,i}} \times ConF_{\phi \to H^*(10), E_j} \times GF_{srf} \times UC$$
(41)

- $Y_{E_{j,i}}^{\gamma}$  [Bq<sup>-1</sup> s<sup>-1</sup>] is the absolute yield of discrete photons of energy E<sub>j</sub> emitted per nuclear transformation of the radionuclide i. It is taken from the CD accompanying Ref. [52].
- ConF<sub>φ→H\*(10),Ej</sub> [pSv cm<sup>2</sup>] is the conversion factor for the ambient dose equivalent H\*(10) [pSv] per fluence [cm<sup>-2</sup>] of photons with energy E<sub>j</sub>. It is obtained from Table A.21 of Ref. [78]. For those energies not in Table A.21 of Ref. [78], the conversion factor was derived by interpolation.
- UC =  $10^{-6} (\text{Sv m}^2)/(\text{pSv cm}^2)$  is a unit conversion factor.

• GF<sub>srf</sub> [unitless] is the geometry factor to account for the measurement being taken at 10 cm from the skin with an instrument with a view of 100 cm<sup>2</sup>. It is determined by Eq. (42), which was obtained based on Eq. 30 of Ref. [79]:

$$GF_{srf} = \frac{1}{4} \ln \left[ 1 + \left( \frac{R_{view-area}}{d_{monitoring}} \right)^2 \right]$$
(42)

- $\circ$  d<sub>monitoring</sub> = 0.1 m is the monitoring distance at which the measurements of the skin are recommend to be taken for the default OIL4<sub>y</sub> value (as indicated in Section 2.6.2), and is considered a reasonable distance for skin surface monitoring under emergency conditions.
- $R_{view-area} = 0.0564$  m is the radius of a circle with an area of 100 cm<sup>2</sup>. This area was chosen because it is the typical surface area of a hand [80] and face. The skin of the hand and face are to be monitored in accordance with the procedures for the default OIL4<sub>y</sub> indicated in Section 2.6.2.

	H* <sub>grd-sh,i</sub> <sup>a</sup>	H* <sub>skin-100cm</sub> <sup>2</sup> <sup>b</sup>
Radionuclide	$[(Sv/s)/(Bq/m^2)]$	$[(Sv/s)/(Bq/m^2)]$
Rb-86	9.1E-17	3.3E-18
Sr-89	2.2E-18	3.2E-21
Sr-90+	5.5E-18	4.7E-22
Sr-91	6.6E-16	2.6E-17
Y-91	5.6E-18	1.1E-19
Zr-95+	2.3E-15	9.1E-17
Zr-97+	1.5E-15	5.9E-17
<b>Mo-99</b> +	2.6E-16	1.1E-17
Ru-103+	4.5E-16	2.1E-17
Ru-105	7.5E-16	3.0E-17
Ru-106+	2.1E-16	8.1E-18
Rh-105	7.5E-17	3.2E-18
Te-127m+	1.6E-17	2.5E-18
Te-127	5.1E-18	2.1E-19
Te-129m+	7.5E-17	5.0E-18
Te-131m	1.3E-15	5.4E-17
Te-132+	2.4E-15	9.8E-17
I-131	3.7E-16	1.6E-17
I-133	5.9E-16	2.4E-17
I-134	2.5E-15	9.4E-17
I-135	1.4E-15	5.3E-17
Cs-134	1.5E-15	6.0E-17
Cs-136	2.0E-15	7.8E-17
Cs-137+	5.7E-16	2.3E-17
Ba-140+	2.7E-15	1.0E-16
Ce-141	7.2E-17	3.7E-18
Ce-143	2.7E-16	1.3E-17
Ce-144+	5.7E-17	2.0E-18
Pr-143	6.9E-19	3.4E-25
Nd-147	1.4E-16	7.1E-18
Np-239	1.6E-16	1.1E-17
Pu-238	8.2E-19	6.5E-19
Pu-239	3.6E-19	2.8E-19
Pu-240	7.9E-19	6.1E-19
Pu-241	1.9E-21	1.4E-22
Am-241	2.7E-17	3.7E-18
Cm-242	9.4E-19	6.4E-19
Cm-244	8.6E-19	5.5E-19

<sup>a</sup> H\*<sub>grd-sh,i</sub> is calculated based on Eq. (40).
<sup>b</sup> H\*<sub>skin-100cm<sup>2</sup>-srf,i</sub> is calculated based on Eq. (41).
<sup>+</sup> The progenies that are in equilibrium with this parent radionuclide (as listed in Table 5) have been considered in the OIL calculations as being in equilibrium with the parent and do not need to be considered separately.

## 3.7.1.3. Thyroid monitoring for $OIL8_{\gamma}$

OIL8<sub> $\gamma$ </sub> was initially calculated for a variety of instruments [81, 82, 83, 84], but it was found that not all instruments would result in a default OIL8<sub> $\gamma$ </sub> value above 0.5 µSv/h, which is considered the lowest practical ambient dose equivalent rate for monitoring the thyroid under severe emergency conditions (lower values may not be feasible due to increased variation in the background radiation). Therefore, in order for OIL8<sub> $\gamma$ </sub> to be operational, i.e. usable under field conditions, a baseline instrument for monitoring the thyroid is used in the calculation of the default OIL8<sub> $\gamma$ </sub> value. This baseline instrument for monitoring the thyroid is defined by (a) a calibration factor  $\leq 3.5E+13$  Bq/(Sv/s) and (b) an effective area  $\leq 15$  cm<sup>2</sup> (due to the geometrical considerations). Instruments with higher calibration factors that will result in default OIL8<sub> $\gamma$ </sub> values below 0.5 µSv/h may be used if the measurement is taken in a low background or the monitoring technique is improved in any other way.

The ambient dose equivalent rate ( $H^*_{I-131-thyroid-burden}$ (baseline-inst)) in front of the thyroid as measured with the baseline instrument for monitoring the thyroid (baseline-inst) against the neck of the infant, per unit activity of I-131 (burden) in the thyroid, is equal to 2.9E-14 (Sv/s)/Bq (i.e. 0.1 ( $\mu$ Sv/h)/kBq, as used in the OIL chart for thyroid monitoring given in Section 2.6.4) and determined by Eq. (43):

$$H_{I-131-thyroid-burden}^{*}(baseline-inst) = \frac{1}{CalF_{R,I-131}(baseline-inst)}$$
(43)

where:

•  $CalF_{R,I-131}$ (baseline-inst) = 3.5E+13 Bq/(Sv/s) is the calibration factor of the baseline monitoring instrument for the activity of I-131 in the thyroid relative to the ambient dose rate measured in front of the thyroid in contact with the skin of an infant.

Examples of possible methods for determining the calibration factor are described in Refs [81, 84]. Providing detailed guidance on how to derive the calibration factor for specific instruments lies beyond the scope of this publication.

## 3.7.2. Beta monitoring instrument response

The response of beta monitoring instruments (e.g. cps) is needed for the calculation of the default and instrument specific  $OIL4_{\beta}$  values. Under emergency conditions, for the same surface activity, the response of different instruments can be considerably different depending on factors such as: (a) the efficiency of the instrument for the radionuclides being measured; (b) the window area of the instrument; (c) the area being monitored relative to the window area of the instrument; (d) surface conditions (e.g. irregular, presence of dirt); and (e) monitoring techniques (e.g. duration of measurement or distance of the detector from the surface). In addition, not all surface contamination monitors are suitable for monitoring skin contamination during an emergency (e.g. the window area may be too large to give representative results for monitoring the hands and face). These factors were accounted for in the default OIL4<sub>β</sub> value by:

• Using the response of a baseline monitoring instrument (for all the radionuclides of concern) in calculating the default  $OIL4_{\beta}$  value, as described in Section 3.7.2.1. This baseline monitoring instrument has a detector area and an efficiency such that its response would be conservative or within a factor of two to the wide range of instruments for monitoring surface contamination typically used for monitoring beta activity on the skin during an

emergency. This does not include instruments designed for detection (e.g. those used by first responders such as HAZMAT teams).

- Including correction factors that account for reductions in count rate due to the distance of the detector from the surface and surface conditions encountered during the emergency.
- Stating the measurement distance and providing a description of a suitable instrument in the beta chart for skin monitoring in Section 2.6.2.

This is done to allow for the use of the default  $OIL4_{\beta}$  value without revision for most suitable instruments. However, if the specific properties (e.g. efficiency and detector area) of the beta monitoring instruments to be used in the response to the emergency are known, either (a) the suitability of the default  $OIL4_{\beta}$  value needs to be confirmed or (b) the default  $OIL4_{\beta}$  value needs to be calculated for this specific instrument, following the instructions provided in Section 5.2. This needs to be done in advance as part of the preparedness, since calculation and effective application of instrument specific OILs early during an emergency may be very difficult, if not impossible.

Section 3.7.2.1 describes the baseline beta monitoring instrument, Section 3.7.2.2 the impact of the field conditions and Section 3.7.2.3 the suitability criteria for beta monitoring instruments.

# 3.7.2.1. Baseline beta monitoring instrument

The baseline beta monitoring instrument is characterized by its effective window area and its ideal response factors, which will be required in calculating the default  $OIL4_{\beta}$  value and determining the coefficients and calibration factors of the baseline beta monitoring instrument. These are needed in the calculation of an instrument specific  $OIL4_{\beta}$  value (as discussed in Section 5.2).

# 3.7.2.1.1. Effective window area of the baseline beta monitoring instrument

For the baseline beta monitoring instrument (used in the calculation of the default  $OIL4_{\beta}$  value provided in Section 2.6.2), an effective window area  $(a_{eff,\beta}(baseline-inst) [cm<sup>2</sup>])$  of 15 cm<sup>2</sup> is used, because it is at the low end of the effective window areas of the commonly available beta monitoring instruments.

# 3.7.2.1.2. Ideal response factor of the baseline beta monitoring instrument

The 'ideal response factor' is equivalent to the efficiency of the baseline monitoring instrument. The term 'ideal response factor' is used instead of 'efficiency' to avoid any confusion with efficiency values obtained based on empirical data (i.e. by using a reference source). This was necessary considering that the efficiencies of the baseline beta monitoring instrument had to be calculated for all radionuclides, since vendors typically only provide the efficiencies for a few radionuclides and not for all those of concern in a reactor emergency.

The ideal response factor of the baseline beta monitoring instrument  $(R_{4\pi,\beta,i}(\text{baseline-inst}) \text{ [cps/Bq]})$  reflects the response [cps] for each nuclear transformation per second [Bq] of radionuclide i on the

surface, for  $4\pi$  geometry<sup>50</sup>, assuming ideal conditions (e.g. no self-absorption or air absorption) and the instrument window being very close to the surface. It is given in Table 32 and determined by Eq. (44):<sup>51</sup>

 $R_{4\pi,\beta,i}(\text{baseline} - \text{inst}) = Y_{\beta,E,i} \times F_{\text{ideal-surf-emission}} \times R_{\beta}(\text{baseline} - \text{inst})$ (44) where:

- $Y_{\beta,E,i}$  [Bq<sup>-1</sup> s<sup>-1</sup>] is the number of particles of interest emitted with energies equal to or above 75 keV (i.e. detectable by the baseline monitor) per nuclear transformation of radionuclide i (yield); the particles of interest include beta +, beta and internal conversion electrons. The values are given in Table 32.
- F<sub>ideal-surf-emission</sub> = 0.5 [unitless] is the ideal surface emission factor, i.e. the ratio between (a) the number of particles that emerge from the front face of a surface and (b) the number of particles emitted from a radionuclide on the surface. This accounts for the fact that half of the emissions escape towards the detector under ideal conditions (e.g. no backscatter, self-absorption or air absorption).
- $R_{\beta}$ (baseline-inst) = 0.3 [counts] is the ratio between (a) the number of counts by the baseline instrument and (b) the number of particles of interest with an energy above 75 keV (and thus assumed to be detected) that reach the outside face of the detector. This value is chosen to ensure that, for the relevant range of energies, the  $R_{4\pi,\beta,i}$ (baseline-inst) values are lower than (i.e. conservative)<sup>52</sup> the  $4\pi$  instrument efficiencies stated by instrument manufacturers for most of the contamination monitoring instruments.

The  $R_{4\pi,\beta,i}$ (baseline-inst) values given in Table 32 were compared to the instrument efficiencies for medium and high energy beta emitters (e.g. C-14, Cl-36, C-60, Cs-137, I-129, Sr-90 and Tc-99) stated by instrument manufacturers for a range of beta surface contamination monitoring instruments that could be used in the field. This does not include instruments designed for detection (e.g. those used by first responders such as HAZMAT teams). It was found that the  $R_{4\pi,\beta,i}$ (baseline-inst) values were lower (i.e. more conservative) than the  $4\pi$  instrument efficiencies stated by instrument manufacturers for most of the instruments, and within a factor of two of those stated by virtually all the instrument vendors.

3.7.2.1.3. Instrument coefficient and calibration factors of the baseline beta monitoring instrument

The calibration factors and instrument coefficients of the baseline beta monitoring instrument are only used for the calculation of an instrument specific OIL4 $_\beta$  value (as discussed in Section 5.2), and therefore only calculated for a limited number of radionuclides, i.e. those often provided by vendors, as described in Table 31.

<sup>&</sup>lt;sup>50</sup> The numerical value of  $R_{4\pi,\beta,i}$  (baseline-inst) corresponds to (or is often referred to) the  $4\pi$  efficiency stated by many manufacturers. It is assumed that the  $2\pi$  efficiency is equal to 2 times the  $4\pi$  efficiency.

<sup>&</sup>lt;sup>51</sup> This approach does not consider backscatter; which is 'conservative', since the count rate with backscatter will be higher than calculated here.

higher than calculated here. <sup>52</sup> 'Conservative' meaning that using the default  $OIL4_{\beta}$  value for the baseline monitor will result in actions being taken at projected dose below the generic criteria under real conditions with virtually all beta monitoring instruments. The conservatism considered in this publication are discussed in Section 3.10.

The baseline beta monitoring instrument coefficient for radionuclide i (i.e. C<sub>β,i</sub>(baseline-inst) [cps/(Bq/cm<sup>2</sup>)]) is given in Table 31 and determined by Eq. (45):

 $C_{\beta,i}(\text{baseline} - \text{inst}) \cong a_{\text{eff},\beta}(\text{baseline} - \text{inst}) \times R_{4\pi,\beta,i}(\text{baseline} - \text{inst})$  (45) where:

- $\circ$  a<sub>eff,β</sub>(baseline-inst) = 15 cm<sup>2</sup> is the effective window area (e.g. sensitive or radiation inlet area) of the baseline beta monitoring instrument and is described in Section 3.7.2.1.1.
- $R_{4\pi,\beta,i}$ (baseline-inst) [cps/Bq] is the ideal response factor of the baseline beta monitoring instrument, which reflects the response [cps] for each nuclear transformation per second [Bq] of radionuclide i on the surface, for  $4\pi$  geometry, assuming ideal conditions (e.g. no self-absorption or air absorption) and the instrument window being very close to the surface. It is given in Table 32.<sup>53</sup>
- The calibration factor of the baseline beta monitoring instrument for radionuclide i (CalF<sub>β,i</sub>(baseline-inst) [(Bq/cm<sup>2</sup>)/cps]) is given in Table 31 and determined by Eq. (46):

$$\operatorname{CalF}_{\beta,i}(\operatorname{baseline} - \operatorname{inst}) = 1/C_{\beta,i}(\operatorname{baseline} - \operatorname{inst})$$
 (46)

## 3.7.2.2. Field beta correction factor

Field monitoring conditions may result in the beta monitoring instruments providing considerably different responses (in cps) to the same levels of deposition of radioactive material. To account for the reduction in the beta count rate for measurements made under field (i.e. non-ideal) conditions, the field beta correction factor of  $\text{CorF}_{\text{field},\beta} = 0.25$  [unitless] is used, which accounts for absorption by the air (OIL4<sub>β</sub> was calculated for measurements at 2 cm from the bare skin) and reductions due to surface conditions. It is determined by Eq. (47):

$$CorF_{field,\beta} = CorF_{distance,\beta} \times CorF_{surface,\beta}$$
(47)

- $\text{CorF}_{\text{distance},\beta} = 0.5$  [unitless] is the distance correction factor for the beta monitoring instrument that accounts for the reduction due to absorption in air at the recommended distance of 2 cm for skin monitoring. It is based on Table 4.5 of Ref. [85] for the medium or high energy beta emitters of concern during a reactor emergency.
- $\text{CorF}_{\text{surface},\beta} = 0.5$  [unitless] is the surface correction factor for the beta monitoring instrument that accounts for the absorption due to non-ideal surface conditions encountered during an emergency. It is based on Table 5.15 of Ref. [85] for the medium or high energy beta emitters of concern during a reactor emergency.

<sup>&</sup>lt;sup>53</sup> The numerical value of  $R_{4\pi,\beta,i}$  (baseline-inst) corresponds to (or is often referred to as) the  $4\pi$  efficiency stated by many manufacturers. It is assumed that the  $2\pi$  efficiency is equal to 2 times the  $4\pi$  efficiency.

## 3.7.2.3. Suitability criteria for the beta monitoring instrument

This section provides criteria to determine if a beta monitoring instrument is suitable for the default  $OIL4_{\beta}$  for monitoring the activity on the skin. The suitability criteria can be divided into (a) general criteria applicable to all beta monitoring instruments to be used for skin monitoring and (b) suitability criteria for use with the default  $OIL4_{\beta}$  value provided in this publication.

# 3.7.2.3.1. General suitability criteria for all beta monitoring instruments

As a minimum criterion, a beta monitoring instrument is considered suitable to be used for monitoring the skin activity if:

- The instrument can display cps (or equivalent cpm) over the relevant ranges of the  $\text{OIL4}_\beta$  values.
- The effective window area of the instrument is less than or equal to 50 cm<sup>2</sup>. This maximum window area is specified to prevent significantly non-conservative responses in terms of counts per second relative to the actual level of surface activity on the hands and face. These non-conservative responses would result from (a) the detector area being significantly larger than the area of the face and hands being monitored (the OIL4<sub>β</sub> value assumes that the window is smaller or approximately equal to the hand or face area being monitored) and (b) significant portions of the monitor window being located at distances from hand and face greater than the 2 cm assumed in the calculation of OIL4<sub>β</sub>.<sup>54</sup>

# 3.7.2.3.2. Suitability criteria for use with the default OIL4 $_{\beta}$ value

A beta monitoring instrument is considered suitable to be used directly with the default  $OIL4_{\beta}$  value provided in this publication, without having to adapt the value to the specific instrument, if:

• Either the calibration factors of the beta monitoring instrument (CalF<sub> $\beta$ ,i</sub>(inst) [(Bq/cm<sup>2</sup>)/cps]), as determined by Eq. (48), are smaller than the calibration factor of the suitable instrument given in Table 31.

$$\operatorname{CalF}_{\beta,i}(\operatorname{inst}) \cong 1/a_{\operatorname{eff},\beta}(\operatorname{inst}) \times R_{4\pi,\beta,i}(\operatorname{inst})$$
 (48)

- $a_{eff,\beta}(inst)$  [cm<sup>2</sup>] is the effective window area (e.g. sensitive or radiation inlet area) of the beta monitoring instrument (inst).
- $R_{4\pi,\beta,i}(inst)$  [cps/Bq] is the response factor of the beta monitoring instrument, which reflects the response [cps] for each nuclear transformation per second [Bq] of radionuclide i on the surface, for  $4\pi$  geometry, assuming ideal conditions (e.g. no selfabsorption or air absorption) and the instrument window being very close to the surface. It is often provided by the manufacturer as the  $4\pi$  efficiency of the instrument. If the

<sup>&</sup>lt;sup>54</sup> This is applicable to any other irregular (i.e. non-plane) surface because of the significant reduction due to absorption in air, as described in Section 3.7.2.2.

manufacturer only provides a  $2\pi$  geometry efficiency, divide this by 2 to get the  $4\pi$ efficiency.55

Or the instrument coefficients of the beta monitoring instrument ( $C_{\beta,i}(inst) [cps/(Bq/cm^2)]$ ), • as determined by Eq. (49), are greater than the instrument coefficients of the suitable instrument given in Table 31.

$$C_{\beta,i}(\text{inst}) \cong a_{\text{eff},\beta}(\text{inst}) \times R_{4\pi,\beta,i}(\text{inst})$$
(49)

These conditions ensure that the instrument has a response within a factor of 2 or higher (i.e. conservative) than that assumed in the development of the  $OIL4_{\beta}$  value.

## TABLE 31. CALIBRATION FACTORS AND INSTRUMENT COEFFICIENTS FOR THE BASELINE AND SUITABLE BETA MONITORING INSTRUMENTS

<b>D</b> 144	Baseline i (baseli	nstrument ne-inst)	Suitable instrument (suitable-inst)	
Emitter	$CalF_{\beta,i}{}^{a}$ [(Bq/cm <sup>2</sup> )/cps]	$C_{\beta,i}^{\ b}$ [cps/(Bq/cm <sup>2</sup> )]	$CalF_{\beta,i}^{c}$ [(Bq/cm <sup>2</sup> )/cps]	$C_{\beta,i}^{\ \ d}$ [cps/(Bq/cm <sup>2</sup> )]
P-32, Cl-36, Cs-137 or other medium or high energy beta emitters $(E_{\beta,max} > 400 \text{ keV})$	0.5	2	< 1	>1
Sr-90 / Y-90 (Sr-90 in equilibrium with progeny Y-90)	0.25	4	< 0.5	> 2

<sup>a</sup> CalF<sub> $\beta$ ,i</sub>(baseline-inst) is determined by Eq. (46).

<sup>b</sup>  $C_{\beta,i}$ (baseline-inst) determined by Eq. (45). <sup>c</sup>  $CalF_{\beta,i}$ (suitable-inst) is 2 times the  $CalF_{\beta,i}$ (baseline-inst).

<sup>d</sup>  $C_{\beta,i}$ (suitable-inst) is 1/2 of the  $C_{\beta,i}$ (baseline-inst).

 $<sup>^{55}</sup>$  4 $\pi$  efficiency applies to emissions in all directions, and  $2\pi$  efficiency applies to emissions toward the detector.

TABLE 32.	ENERGY	RELATED	YIELD	AND	IDEAL	RESPONSE	FACTOR	OF	THE
BASELINE I	BETA MON	NITORING I	NSTRU	MENT					

Dadionualida	$Y_{\beta,E,i}^{a}$	$\mathbf{R}_{4\pi,\beta,i}(baseline-inst)^{b}$
Kaulonuchue	$[Bq^{-1} s^{-1}]$	[cps/Bq]
P-32 °	9.7E-01	1.5E-01
Cl-36 °	8.6E-01	1.3E-01
Rb-86	9.5E-01	1.4E-01
Sr-89	9.4E-01	1.4E-01
Sr-90+	1.8E+00	2.6E-01
Sr-91	9.4E-01	1.4E-01
Y-91	9.5E-01	1.4E-01
Zr-95+	1.0E+00	1.5E-01
Zr-97+	1.9E+00	2.9E-01
Mo-99+	1.0E+00	1.5E-01
Ru-103+	3.5E-01	5.2E-02
Ru-105	1.1E+00	1.7E-01
Ru-106+	9.9E-01	1.5E-01
Rh-105	6.8E-01	1.0E-01
Te-127m+	1.4E+00	2.1E-01
Te-127	8.1E-01	1.2E-01
Te-129m+	1.3E+00	1.9E-01
Te-131m	8.0E-01	1.2E-01
Te-132+	1.4E+00	2.1E-01
I-131	7.8E-01	1.2E-01
I-133	9.1E-01	1.4E-01
I-134	9.7E-01	1.5E-01
I-135	8.5E-01	1.3E-01
Cs-134	5.9E-01	8.8E-02
Cs-136	7.3E-01	1.1E-01
Cs-137+	8.6E-01	1.3E-01
Ba-140+	2.0E+00	3.0E-01
Ce-141	9.1E-01	1.4E-01
Ce-143	9.3E-01	1.4E-01
Ce-144+	1.5E+00	2.3E-01
Pr-143	8.8E-01	1.3E-01
Nd-147	8.9E-01	1.3E-01
Np-239	1.3E+00	2.0E-01
Pu-238	1.0E-03	1.5E-04
Pu-239	3.9E-04	5.8E-05
Pu-240	8.0E-04	1.2E-04
Pu-241	7.9E-06	1.2E-06
Am-241	3.3E-03	5.0E-04
Cm-242	4.0E-04	6.1E-05
Cm-244	3.1E-04	4.6E-05

<sup>a</sup> Y<sub>β,E,i</sub> [Bq<sup>-1</sup> s<sup>-1</sup>] is obtained from the CD accompanying Ref. [52].
<sup>b</sup> R<sub>4π,β,i</sub>(baseline-inst) is calculated based on Eq. (44).
<sup>c</sup> P-32 and Cl-36 are provided in this table (and not in the other tables) because they are often used for the calibration of instruments, but are not relevant to the dose received by the public following a severe release of radioactive material from an LWR or its spent fuel.

+ The progenies that are in equilibrium with this parent radionuclide (as listed in Table 5) have been considered in the OIL calculations as being in equilibrium with the parent and do not need to be considered separately.

#### 3.8. OIL(t,mix) FUNCTIONS AND DEFAULT OIL VALUES

This section provides the calculations to determine the time and mix dependent OIL(t,mix) functions, which are used as a basis for selecting the default OIL values. For each of the OILs, two sections are provided describing (a) the calculation of the OIL(t,mix) functions and (b) the analysis of the OIL(t,mix) functions to choose default OIL values.

Although the use of prefixed SI units and the use of non-SI units has been avoided as far as reasonable throughout this publication, due to the common use of  $\mu$ Sv/h when monitoring, the OIL(t,mix) functions and default OIL values are given in  $\mu$ Sv/h.

## **3.8.1. ΟΙL1**<sub>γ</sub>

#### 3.8.1.1. $OIL1_{\gamma}(t,mix)$ calculation

Calculation of the default  $OIL1_{\gamma}$  value is based on evaluation of the  $OIL1_{\gamma}(t,mix)$  functions for each mix given in Table 3.  $OIL1_{\gamma}(t,mix)$  [ $\mu$ Sv/h] is the ambient dose equivalent rate measured at 1 m above ground level at time t after reactor shutdown, which, for the 'ground' scenario, results in any of the generic criteria specified for  $OIL1_{\gamma}$  (in Table 2) being met, for a specific radionuclide mix. The values are given in Fig. 9 for all considered mixes and are determined by Eq. (50):

$$\text{DIL1}_{\gamma}(t, \text{mix}) = \left(\sum_{i} \left( \text{RA}_{i}(t, \text{mix}) \times \text{H}_{\text{grd}-\text{sh}, i}^{*} \right) \right) \times \text{WF}_{\text{OIL1}_{\gamma}} \times \text{UC} \times \text{DA}_{\text{OIL1}}(t, \text{mix}) \quad (50)$$

where:

- t [s] is the time after shutdown of the reactor (i.e. after the fuel was last irradiated in a reactor).
- 'mix' refers to the mix number as provided in Table 3, Table 7, Table 8 and Table 9.
- RA<sub>i</sub>(t,mix) [unitless] is the relative activity of radionuclide i at time t after reactor shutdown for a specific radionuclide mix. It is determined by Eq. (1).
- H\*<sub>grd-sh,i</sub> [(Sv/s)/(Bq/m<sup>2</sup>)] is the ambient dose equivalent rate at 1 m above ground level, per unit ground surface activity of radionuclide i. The values are given in Table 30.
- $WF_{OIL1\gamma} = 3$  [unitless] is a weighting factor used to avoid the implementation of unwarranted response actions based on the default OIL1<sub> $\gamma$ </sub> value. It is described in Section 3.9.
- UC =  $3.6E+09 (\mu Sv/h)/(Sv/s)$  is the unit conversion factor from Sv/s to  $\mu Sv/h$ .
- DA<sub>OIL1</sub>(t,mix) [Bq/m<sup>2</sup>] is the derived gross activity on the ground at time t after reactor shutdown, which, for the 'ground', scenario results in any of the generic criteria specified for OIL1<sub>γ</sub> (in Table 2) being met, for a specific radionuclide mix. It is determined by Eq. (51):

$$DA_{OIL1}(t, mix) = min\{A_{OIL1,E}(t, mix), A_{OIL1,H_{fetus}}(t, mix)\}$$
(51)

where:

•  $A_{OIL1,E}(t,mix)$  [Bq/m<sup>2</sup>] is the gross activity on the ground at time t after reactor shutdown, which, for the 'ground' scenario, results in a total effective dose to the representative person meeting the GC(Urgent,E,7d), for a specific radionuclide mix. It is determined by Eq. (52).

$$A_{OIL1,E}(t, mix) = \frac{GC(Urgent, E, 7d)}{\sum_{i} \left( E_{grd-scenario,i}(7d) \times RA_{i}(t, mix) \right)}$$
(52)

where:

- GC(Urgent,E,7d) = 0.1 Sv is the generic criterion used to take urgent response actions based on the total effective dose to the representative person in the first 7 days (given in Table 2).
- $E_{grd-scenario,i}(7d) [Sv/(Bq/m^2)]$  is the total effective dose to the representative person over an exposure period  $\Delta$  of 7 days for the 'ground' scenario, per unit ground surface activity of radionuclide i. The values are given in Table 23.
- RA<sub>i</sub>(t,mix) [unitless] is the relative activity of radionuclide i at time t after reactor shutdown for a specific radionuclide mix. It is determined by Eq. (1).
- $\circ$  A<sub>OIL1,Hfetus</sub>(t,mix) [Bq/m<sup>2</sup>] is the gross activity on the ground at time t after reactor shutdown, which, for the 'ground' scenario results in a total equivalent dose to the fetus meeting the GC(Urgent,H<sub>fetus</sub>,7d), for a specific radionuclide mix. It is determined by Eq. (53):

$$A_{OIL1,H_{fetus}}(t,mix) = \frac{GC(Urgent, H_{fetus}, 7d)}{\sum_{i} \left( H_{fetus,grd-scenario,i}(7d) \times RA_{i}(t,mix) \right)}$$
(53)

where:

- GC(Urgent,H<sub>fetus</sub>,7d) = 0.1 Sv is the generic criterion used to take urgent response actions based on the total equivalent dose to the fetus in the first 7 days (given in Table 2).
- $H_{fetus,grd-scenario,i}(7d)$  [Sv/(Bq/m<sup>2</sup>)] is the total equivalent dose to the fetus over an exposure period  $\Delta$  of 7 days for the 'ground' scenario, per unit ground surface activity of radionuclide i. The values are given in Table 23.
- RA<sub>i</sub>(t,mix) [unitless] is the relative activity of radionuclide i at time t after reactor shutdown for a specific radionuclide mix. It is determined by Eq. (1).

## 3.8.1.2. Analysis

The OIL1<sub> $\gamma$ </sub>(t,mix) functions calculated in this way are displayed in Fig. 9.<sup>56</sup> The red line shows the chosen default OIL1<sub> $\gamma$ </sub> value of 1000  $\mu$ Sv/h, which is lower than the OIL1<sub> $\gamma$ </sub>(t,mix) functions for almost all times after shutdown and mixes.

<sup>&</sup>lt;sup>56</sup> The reason that some of the OIL(t,mix) functions are not smooth is the 'min' function used in the calculation of the DA(t,mix).



FIG. 9.  $OIL1_{\gamma}(t,mix)$  [ $\mu$ Sv/h] functions and default  $OIL1_{\gamma}$  value (in red).



FIG. 10.  $OIL2_{\gamma}(t,mix)$  [ $\mu$ Sv/h] functions and default  $OIL2_{\gamma}$  value (in red).

## **3.8.2. ΟΙL2**<sub>γ</sub>

## 3.8.2.1. $OIL2_{\gamma}(t,mix)$ calculation

Calculation of the default  $OIL2_{\gamma}$  value is based on evaluation of the  $OIL2_{\gamma}(t,mix)$  functions for each mix given in Table 3.  $OIL2_{\gamma}(t,mix)$  [ $\mu$ Sv/h] is the ambient dose equivalent rate measured at 1 m above ground level at time t after reactor shutdown, that for the 'ground' scenario results in any of the generic criteria specified for  $OIL2_{\gamma}$  (in Table 2) being met, for a specific radionuclide mix. The values are given in Fig. 10 for all considered mixes and determined by Eq. (54):

$$OIL2_{\gamma}(t, mix) = \left(\sum_{i} (RA_{i}(t, mix) \times H^{*}_{grd-sh,i})\right) \times WF_{OIL2_{\gamma}} \times UC \times DA_{OIL2}(t, mix) \quad (54)$$

where:

- t [s] is the time after shutdown of the reactor (i.e. after the fuel was last irradiated in a reactor).
- 'mix' refers to the mix number as provided in Table 3, Table 7, Table 8 and Table 9.
- RA<sub>i</sub>(t,mix) [unitless] is the relative activity of radionuclide i at time t after reactor shutdown for a specific radionuclide mix. It is determined by Eq. (1).
- H\*<sub>grd-sh,i</sub> [(Sv/s)/(Bq/m<sup>2</sup>)] is the ambient dose equivalent rate at 1 m above ground level, per unit ground surface activity of radionuclide i. The values are given in Table 30.
- $WF_{OIL2\gamma} = 1$  [unitless] is a weighting factor used to avoid the implementation of unwarranted response actions based on the default  $OIL2_{\gamma}$  value. It is described in Section 3.9.
- UC =  $3.6E+09 (\mu Sv/h)/(Sv/s)$  is the unit conversion factor from Sv/s to  $\mu Sv/h$ .
- DA<sub>OIL2</sub>(t,mix) [Bq/m<sup>2</sup>] is the derived gross activity on the ground at time t after reactor shutdown, which, for the 'ground' scenario, results in any of the generic criteria specified for OIL2<sub>γ</sub> (in Table 2) being met, for a specific radionuclide mix. It is determined by Eq. (55):

$$DA_{OIL2}(t, mix) = min\{A_{OIL2,E}(t, mix), A_{OIL2,H_{fetus}}(t, mix)\}$$
(55)

where:

 $\circ$  A<sub>OIL2,E</sub>(t,mix) [Bq/m<sup>2</sup>] is the gross activity on the ground at time t after reactor shutdown, which, for the 'ground' scenario, results in a total effective dose to the representative person meeting the GC(Early,E,1a), for a specific radionuclide mix. It is determined by Eq. (56):

$$A_{OIL2,E}(t, mix) = \frac{GC(Early, E, 1a)}{\sum_{i} \left( E_{grd-scenario,i}(1a) \times RA_{i}(t, mix) \right)}$$
(56)

- GC(Early,E,1a) = 0.1 Sv is the generic criterion used to take early response actions based on the total effective dose to the representative person in the first year (given in Table 2).
- E<sub>grd-scenario,i</sub>(1a) [Sv/(Bq/m<sup>2</sup>)] is the total effective dose to the representative person over an exposure period Δ of 1 year for the 'ground' scenario, per unit ground surface activity of radionuclide i. The values are given in Table 23.

- RA<sub>i</sub>(t,mix) [unitless] is the relative activity of radionuclide i at time t after reactor shutdown for a specific radionuclide mix. It is determined by Eq. (1).
- $\circ$  A<sub>OIL2,Hfetus</sub>(t,mix) [Bq/m<sup>2</sup>] is the gross activity on the ground at time t after reactor shutdown, which, for the 'ground' scenario, results in a total equivalent dose to the fetus meeting the GC(Early,H<sub>fetus</sub>,9mo), for a specific radionuclide mix. It is determined by Eq. (57):

$$A_{OIL2,H_{fetus}}(t,mix) = \frac{GC(Early,H_{fetus},9mo)}{\sum_{i} \left(H_{fetus,grd-scenario,i}(1a) \times RA_{i}(t,mix)\right)}$$
(57)

where:

- GC (Early,H<sub>fetus</sub>,9mo) = 0.1 Sv is the generic criterion used to take early response actions based on the total equivalent dose to the fetus in the full period of in utero development (given in Table 2).
- H<sub>fetus,grd-scenario,i</sub>(1a) [Sv/(Bq/m<sup>2</sup>)] is the total equivalent dose to the fetus over an exposure period Δ of 1 year for the 'ground' scenario, per unit ground surface activity of radionuclide i. The values are given in Table 23.
- RA<sub>i</sub>(t,mix) [unitless] is the relative activity of radionuclide i at time t after reactor shutdown for a specific radionuclide mix. It is determined by Eq. (1).

## 3.8.2.2. Analysis

The OIL2<sub> $\gamma$ </sub>(t,mix) functions calculated in this way are displayed in Fig. 10. The red line shows the two chosen default values for OIL2<sub> $\gamma$ </sub>: (a) 100  $\mu$ Sv/h for measurements that are taken less than 10 days after the shutdown of a reactor and (b) 25  $\mu$ Sv/h for measurements that are taken later than 10 days after reactor shutdown. Two values are chosen to account for the high OIL2<sub> $\gamma$ </sub> values during the first 10 days after reactor shutdown caused by the short lived radionuclides.

Despite being greater than the  $OIL2_{\gamma}(t,mix)$  for two mixes during short times after shutdown, the default  $OIL2_{\gamma}$  is acceptable, because this OIL triggers the implementation of early response actions in the first days after an emergency has been declared (as described in Ref. [4]), which is only a fraction of the exposure period of 1 year assumed in the calculations. Therefore, the resulting dose will be well below the generic criterion of 100 mSv in all cases.

# **3.8.3. ΟΙL3**<sub>γ</sub>

## 3.8.3.1. $OIL3_{\gamma}(t,mix)$ calculation

Calculation of default  $OIL3_{\gamma}$  is based on evaluation of the  $OIL3_{\gamma}(t,mix)$  functions for each mix given in Table 3.  $OIL3_{\gamma}(t,mix)$  [ $\mu$ Sv/h] is the ambient dose equivalent rate measured at 1 m above ground level at time t after reactor shutdown, which, for the 'food pre-analysis' scenario, results in any of the generic criteria specified for  $OIL3_{\gamma}$  (in Table 2) being met, for a specific radionuclide mix. The values are given in Fig. 11 for all considered mixes and determined by Eq. (58):

$$OIL3_{\gamma}(t, mix) = \left(\sum_{i} \left( RA_{i}(t, mix) \times H_{grd-sh,i}^{*} \right) \right) \times WF_{OIL3_{\gamma}} \times UC \times DA_{OIL3}(t, mix)$$
(58)

where:

- t [s] is the time after shutdown of the reactor (i.e. after the fuel was last irradiated in a reactor).
- 'mix' refers to the mix number as provided in Table 3, Table 7, Table 8 and Table 9.
- RA<sub>i</sub>(t,mix) [unitless] is the relative activity of radionuclide i at time t after reactor shutdown for a specific radionuclide mix. It is determined by Eq. (1).
- H\*<sub>grd-sh,i</sub> [(Sv/s)/(Bq/m<sup>2</sup>)] is the ambient dose equivalent rate at 1 m above ground level, per unit ground surface activity of radionuclide i. The values are given in Table 30.
- $WF_{OIL3\gamma} = 5$  [unitless] is a weighting factor used to avoid the implementation of unwarranted response actions based on the default OIL3<sub> $\gamma$ </sub> value. It is described in Section 3.9.
- UC =  $3.6E+09 (\mu Sv/h)/(Sv/s)$  is the unit conversion factor from Sv/s to  $\mu Sv/h$ .
- DA<sub>OIL3</sub>(t,mix) [Bq/m<sup>2</sup>] is the derived gross activity on the ground at time t after reactor shutdown, which, for the 'food pre-analysis' scenario, results in any of the generic criteria specified for OIL3<sub>γ</sub> (in Table 2) being met, for a specific radionuclide mix. It is determined by Eq. (59):

$$DA_{OIL3}(t, mix) = \min \left\{ A_{OIL3, e_{ing}}(t, mix), A_{OIL3, h_{fetus, ing}}(t, mix) \right\}$$
(59)

where:

•  $A_{OIL3,eing}(t,mix)$  [Bq/m<sup>2</sup>] is the gross activity on the ground at time t after reactor shutdown, which, for the 'food pre-analysis' scenario, results in a committed effective dose from ingestion to the representative person meeting the GC(Ingestion,e<sub>ing</sub>,1a), for a specific radionuclide mix. It is determined by Eq. (60):

$$A_{OIL3,e_{ing}}(t,mix) = \frac{GC(Ingestion, e_{ing}, 1a)}{\sum_{i} \left( e_{ing,food-pre-analysis-scenario,i} \times RA_{i}(t,mix) \right)}$$
(60)

- GC(Ingestion,e<sub>ing</sub>,1a) = 0.01 Sv is the generic criterion used to restrict the ingestion of non-essential food, milk and drinking water based on the committed effective dose from ingestion to the representative person over the first year (given in Table 2).
- e<sub>ing,food-pre-analysis-scenario,i</sub> [Sv/(Bq/m<sup>2</sup>)] is the committed effective dose to the representative person (defined by a combination of the adult and infant dose conversion factors and consumption rates) from ingestion over 1 year, for the 'food pre-analysis' scenario, per unit ground surface activity of radionuclide i. The values are given in Table 26.
- RA<sub>i</sub>(t,mix) [unitless] is the relative activity of radionuclide i at time t after reactor shutdown for a specific radionuclide mix. It is determined by Eq. (1).
- $\circ$  A<sub>OIL3,hfetus,ing</sub>(t,mix) [Bq/m<sup>2</sup>] is the gross activity on the ground at time t after reactor shutdown, which, for the 'food pre-analysis' scenario, results in a committed

equivalent dose to the fetus from ingestion meeting the GC(Ingestion, $h_{fetus,ing}$ ,9mo), for a specific radionuclide mix. It is determined by Eq. (61):

$$A_{OIL3,h_{fetus,ing}}(t, mix) = = \frac{GC(Ingestion, h_{fetus,ing}, 9mo)}{\sum_{i} (h_{fetus,ing,food-pre-analysis-scenario,i} \times RA_{i}(t, mix))}$$
(61)

where:

- GC(Ingestion,h<sub>fetus,ing</sub>,9mo) = 0.01 Sv is the generic criterion used to restrict the ingestion of non-essential food, milk and drinking water based on the committed equivalent dose to the fetus from ingestion over the full period of in utero development (given in Table 2).
- h<sub>fetus,ing,food-pre-analysis-scenario,i</sub> [Sv/(Bq/m<sup>2</sup>)] is the committed equivalent dose to the fetus from ingestion of food, milk and drinking water by the pregnant woman, over the period of in utero development, or the 'food pre-analysis' scenario, per unit ground surface activity of radionuclide i. The values are given in Table 26.
- RA<sub>i</sub>(t,mix) [unitless] is the relative activity of radionuclide i at time t after reactor shutdown for a specific radionuclide mix. It is determined by Eq. (1).

# 3.8.3.2. Analysis

The OIL3<sub> $\gamma$ </sub>(t,mix) functions calculated in this way are displayed in Fig. 11. The red line shows the chosen default OIL3<sub> $\gamma$ </sub> value. As the default OIL3<sub> $\gamma$ </sub> value, 1 µSv/h was chosen, despite the fact that it is greater than the OIL3<sub> $\gamma$ </sub>(t,mix) [µSv/h] by as much as a factor of 3 for certain mixes and times after shutdown. This is considered acceptable because of the very conservative assumption that as much as 50% of all the food, milk and water ingested is affected. In addition, 1µSv/h is considered the lowest practical ambient dose equivalent rate to be used for ground monitoring under severe emergency conditions (lower values may not be feasible owing to increased variation in the background radiation).



FIG. 11.  $OIL3_{\gamma}(t,mix)$  [ $\mu$ Sv/h] functions and default  $OIL3_{\gamma}$  value (in red).

# 3.8.4. OIL4<sub>γ</sub>

#### 3.8.4.1. $OIL4_{\gamma}(t, mix)$ calculation

Calculation of default OIL4<sub> $\gamma$ </sub> is based on evaluation of the OIL4<sub> $\gamma$ </sub>(t,mix) functions for each mix given in Table 3. OIL4<sub> $\gamma$ </sub>(t,mix) [ $\mu$ Sv/h] is the ambient dose equivalent rate measured at 10 cm from bare skin at time t after reactor shutdown, which, for the 'skin' scenario, results in any of the generic criteria specified for OIL4<sub> $\gamma$ </sub> (in Table 2) being met, for a specific radionuclide mix. The values are given in Fig. 12 for all considered mixes and determined by Eq. (62):

$$OIL4_{\gamma}(t, mix) = \left(\sum_{i} \left(RA_{i}(t, mix) \times H_{skin-100cm^{2}-srf,i}^{*}\right)\right) \times WF_{OIL4_{\gamma}} \times UC \times DA_{OIL4}(t, mix)$$
<sup>(62)</sup>

- t [s] is the time after shutdown of the reactor (i.e. after the fuel was last irradiated in a reactor).
- 'mix' refers to the mix number as provided in Table 3, Table 7, Table 8 and Table 9.
- RA<sub>i</sub>(t,mix) [unitless] is the relative activity of radionuclide i at time t after reactor shutdown for a specific radionuclide mix. It is determined by Eq. (1).
- H\*<sub>skin-100cm<sup>2</sup>-srf,i</sub> [(Sv/s)/(Bq/m<sup>2</sup>)] is the ambient dose equivalent rate at 10 cm from activity on 100 cm<sup>2</sup> of the skin surface, per unit surface activity of radionuclide i. The values are given in Table 30.

- $WF_{OIL4\gamma} = 0.5$  [unitless] is a weighting factor used to avoid the implementation of unwarranted response actions based on the default OIL4<sub> $\gamma$ </sub> value. It is described in Section 3.9.
- UC =  $3.6E+09 (\mu Sv/h)/(Sv/s)$  is the unit conversion factor from Sv/s to  $\mu Sv/h$ .
- DA<sub>OIL4</sub>(t,mix) [Bq/m<sup>2</sup>] is the derived gross activity on the skin surface at time t after reactor shutdown, which, for the 'skin' scenario, results in any of the generic criteria specified for OIL4<sub>y</sub> (in Table 2) being met, for a specific radionuclide mix. It is determined by Eq. (63):

 $DA_{OIL4}(t, mix) = \min\{A_{OIL4,e}(t, mix), A_{OIL4,h_{fetus}}(t, mix), A_{OIL4,AD_{skin}}(t, mix)\}$ (63)

where:

A<sub>OIL4,e</sub>(t,mix) [Bq/m<sup>2</sup>] is the gross activity on the skin surface at time t after reactor shutdown, which, for the 'skin' scenario, results in an effective dose to the representative person from inadvertent ingestion meeting the GC(Urgent,E,7d), for a specific radionuclide mix. It is determined by Eq. (64):

$$A_{OIL4,e}(t, mix) = \frac{GC(Urgent, E, 7d)}{\sum_{i} \left( e_{ing,skin-scenario,i} \times RA_{i}(t, mix) \right)}$$
(64)

where:

- GC(Urgent,E,7d) = 0.1 Sv is the generic criterion used to take urgent response actions based on the total effective dose to the representative person in the first 7 days (given in Table 2).
- e<sub>ing,skin-scenario,i</sub> [Sv/(Bq/m<sup>2</sup>)] is the committed effective dose to the representative person (i.e. the infant for this exposure scenario and pathway) from inadvertent ingestion, for the 'skin' scenario, per unit skin surface activity of radionuclide i. The values are given in Table 28.
- RA<sub>i</sub>(t,mix) [unitless] is the relative activity of radionuclide i at time t after reactor shutdown for a specific radionuclide mix. It is determined by Eq. (1).
- A<sub>OIL4,hfetus</sub>(t,mix) [Bq/m<sup>2</sup>] is the gross activity on the skin surface at time t after reactor shutdown, that for the 'skin' scenario results in an equivalent dose to the fetus from inadvertent ingestion meeting the GC(Urgent,H<sub>fetus</sub>,7d), for a specific radionuclide mix. It is determined by Eq. (65):

$$A_{OIL4,h_{fetus}}(t,mix) = \frac{GC(Urgent, H_{fetus}, 7d)}{\sum_{i} \left( h_{fetus,ing,skin-scenario,i} \times RA_{i}(t,mix) \right)}$$
(65)

- GC(Urgent,H<sub>fetus</sub>,7d) = 0.1 Sv is the generic criterion used to take urgent response actions based on the total equivalent dose to the fetus in the first 7 days (given in Table 2).
- h<sub>fetus,ing,skin-scenario,i</sub> [Sv/(Bq/m<sup>2</sup>)] is the committed equivalent dose to the fetus from inadvertent ingestion of radioactive material on the skin, for the 'skin' scenario, per unit skin surface activity of radionuclide i. The values are given in Table 28.

- RA<sub>i</sub>(t,mix) [unitless] is the relative activity of radionuclide i at time t after reactor shutdown for a specific radionuclide mix. It is determined by Eq. (1).
- $\circ$  A<sub>OIL4,ADskin</sub>(t,mix) [Bq/m<sup>2</sup>] is the gross activity on the skin surface at time t after reactor shutdown, which, for the 'skin' scenario, results in an RBE weighted absorbed dose to the skin meeting the GC(Acute,AD<sub>skin-ext</sub>,10h), for a specific radionuclide mix. It is determined by Eq. (66):

$$A_{OIL4,AD_{skin}}(t, mix) = \frac{GC(Acute, AD_{skin-ext}, 10h)}{\sum_{i} (ad_{skin, skin-scenario, i} \times RA_{i}(t, mix))}$$
(66)

where:

- GC(Acute,AD<sub>skin-ext</sub>,10h) = 10 Gy to 100 cm<sup>2</sup> is the generic criterion used to take response actions under any circumstance based on the RBE weighted absorbed dose to the skin over 10 hours (given in Table 2).
- ad<sub>skin,skin-scenario,i</sub> [Gy/(Bq/m<sup>2</sup>)] is the RBE weighted absorbed dose to the skin derma of the representative person,<sup>57</sup> for the 'skin' scenario, per unit skin surface activity of radionuclide i. The values are given in Table 28.
- RA<sub>i</sub>(t,mix) [unitless] is the relative activity of radionuclide i at time t after reactor shutdown for a specific radionuclide mix. It is determined by Eq. (1).

# 3.8.4.2. Analysis

The OIL4<sub> $\gamma$ </sub>(t,mix) functions calculated in this way are displayed in Fig. 12. The red line shows the chosen default OIL4<sub> $\gamma$ </sub> value of 1 µSv/h, which is conservative for all times after shutdown and mixes. In fact, default OIL4<sub> $\gamma$ </sub> values as high as 4 µSv/h would be conservative as described in Section 3.9.

<sup>&</sup>lt;sup>57</sup> The RBE weighted absorbed dose rate to the skin derma is considered to be age independent.



FIG. 12.  $OIL4_{\gamma}(t,mix)$  [ $\mu$ Sv/h] functions and default  $OIL4_{\gamma}$  value (in red).



FIG. 13.  $OIL4_{\beta}(t,mix)$  [cps] functions and default  $OIL4_{\beta}$  value (in red).

# 3.8.5. OIL4<sub>β</sub>

## 3.8.5.1. $OIL4_{\beta}(t,mix)$ calculation

Calculation of default OIL4<sub> $\beta$ </sub> is based on evaluation of the OIL4<sub> $\beta$ </sub>(t,mix) functions for each mix given in Table 3. OIL4<sub> $\beta$ </sub>(t,mix) [cps] is the beta count rate measured with the baseline beta monitoring instrument at 2 cm from bare skin at time t after reactor shutdown, which, for the 'skin' scenario, results in any of the generic criteria specified for OIL4<sub> $\beta$ </sub> (in Table 2) being met, for a specific radionuclide mix. The values are given in Fig. 13 for all considered mixes and determined by Eq. (67):

$$OIL4_{\beta}(t, mix) = \left(\sum_{i} \left( RA_{i}(t, mix) \times R_{4\pi,\beta,i}(baseline - inst) \right) \right) \times WF_{OIL4_{\beta}} \times DA_{OIL4}(t, mix) \times a_{eff,\beta}(baseline - inst) \times CorF_{field,\beta}$$
(67)

where:

- t [s] is the time after shutdown of the reactor (i.e. after the fuel was last irradiated in a reactor).
- 'mix' refers to the mix number as provided in Table 3, Table 7, Table 8 and Table 9.
- $R_{4\pi,\beta,i}$ (baseline-inst) [cps/Bq] is the ideal response factor of the baseline beta monitoring instrument, which reflects the response [cps] for each nuclear transformation per second [Bq] of radionuclide i on the surface, for  $4\pi$  geometry, assuming ideal conditions (e.g. no self-absorption or air absorption) and the instrument window being very close to the surface.<sup>58</sup> It is given in Table 32.
- RA<sub>i</sub>(t,mix) [unitless] is the relative activity of radionuclide i at time t after reactor shutdown for a specific radionuclide mix. It is determined by Eq. (1).
- $WF_{OIL4\beta} = 0.5$  [unitless] is a weighting factor used to avoid the implementation of unwarranted response actions based on the default OIL4<sub> $\beta$ </sub> value. It is described in Section 3.9.
- DA<sub>OIL4</sub>(t,mix) [Bq/m<sup>2</sup>] is the derived gross activity on the skin surface at time t after reactor shutdown, that for the 'skin' scenario results in any of the generic criteria specified for OIL4<sub>β</sub> (Table 2) being met, for a specific radionuclide mix. It is determined by Eq. (63).
- $a_{eff,\beta}(baseline-inst) = 15 \text{ cm}^2 = 1.5\text{E-03 m}^2$  is the effective window area (e.g. sensitive or radiation inlet area) of the baseline beta monitoring instrument and is described in Section 3.7.2.1.1.
- $\text{CorF}_{\text{field},\beta} = 0.25$  [unitless] is the field beta correction factor for the beta monitoring instrument that accounts for the reduction in the beta count rate for measurements made under field (i.e. non-ideal) conditions. It is determined by Eq. (47).

# 3.8.5.2. Analysis

The OIL4<sub> $\beta$ </sub>(t,mix) functions calculated in this way are displayed in Fig. 13. The red line shows the default OIL4<sub> $\beta$ </sub> value of 1000 cps, which is conservative for all times after shutdown and mixes. In fact, default OIL4<sub> $\beta$ </sub> values as high as 2000 cps would be conservative as described in Section 3.9.

<sup>&</sup>lt;sup>58</sup> The numerical value of  $R_{4\pi,\beta,i}$  (baseline-inst) corresponds to (or is often referred to) the  $4\pi$  efficiency stated by many manufacturers.  $2\pi$  efficiency is equal to 2 times the  $4\pi$  efficiency.

## 3.8.6. OIL7

#### 3.8.6.1. OIL7(t,mix) calculation

OIL7 values are expressed as activity concentrations [Bq/kg] of the two marker radionuclides I-131 (OIL7<sub>I-131</sub>) and Cs-137 (OIL7<sub>Cs-137</sub>) in food, milk or drinking water,<sup>59</sup> which, for the 'food post-analysis' scenario, result in any of the generic criteria specified for OIL7 (in Table 2) being met, thus indicating if the food, milk and water are safe for consumption.

Calculation of the default  $OIL7_{I-131}$  and  $OIL7_{Cs-137}$  values is based on evaluation of the  $OIL7_{I-131}(t,mix)$  and  $OIL7_{Cs-137}(t,mix)$  functions for each mix given in Table 3, as described below:

• OIL7<sub>I-131</sub>(t,mix) [Bq/kg] is the activity concentration of the marker radionuclide I-131 in food, milk or drinking water at time t after reactor shutdown, which, for the 'food post-analysis' scenario, results in any of the generic criteria specified for OIL7 (in Table 2) being met, for a specific radionuclide mix (it needs to be used together with OIL7<sub>Cs-137</sub>). The values are given in Fig. 14 for all considered mixes and determined by Eq. (68):

$$OIL7_{I-131}(t, mix) = RA_{I-131}(t, mix) \times WF_{OIL7} \times DA_{OIL7}(t, mix)$$
(68)

where:

- t [s] is the time after shutdown of the reactor (i.e. after the fuel was last irradiated in a reactor).
- o 'mix' refers to the mix number as provided in Table 3, Table 7, Table 8 and Table 9.
- RA<sub>I-131</sub>(t,mix) [unitless] is the relative activity of I-131 at time t after reactor shutdown for a specific radionuclide mix. It is determined by Eq. (1).
- $\circ$  WF<sub>OIL7</sub> = 5 [unitless] is a weighting factor used to avoid the implementation of unwarranted response actions based on the default OIL7 value. It is described in Section 3.9.
- DA<sub>OIL7</sub>(t,mix) [Bq/kg] is the activity concentration in food, milk or drinking water at time t after reactor shutdown, which, for the 'food post-analysis' scenario, results in any of the generic criteria specified for OIL7 (in Table 2) being met, for a specific radionuclide mix. It is determined by Eq. (69):

$$DA_{OIL7}(t, mix) = \min \left\{ A_{OIL7,e_{ing}}(t, mix), A_{OIL7,h_{fetus,ing}}(t, mix) \right\}$$
(69)

where:

A<sub>OIL7,eing</sub>(t,mix) [Bq/kg] is the activity concentration in food, milk or drinking water, which, for the 'food post-analysis' scenario, results in a committed effective dose to the representative person from ingestion meeting the GC(Ingestion,eing,1a), for a specific radionuclide mix. It is determined by Eq. (70):

<sup>&</sup>lt;sup>59</sup> OIL7 applies to food, milk and water destined for human consumption (i.e. it is not applicable for 'not ready to eat' products such as dried food or concentrated food).
$$A_{OIL7,e_{ing}}(t, mix) =$$

$$= \frac{GC(Ingestion, e_{ing}, 1a)}{\sum_{i} \left( e_{ing,food-post-analysis-scenario,i} \times RA_{i}(t, mix) \right)}$$
(70)

where:

- GC(Ingestion,e<sub>ing</sub>,1a) = 0.01 Sv is the generic criterion used to restrict the ingestion of non-essential food, milk and drinking water based on the committed effective dose from ingestion to the representative person over the first year (as given in Table 2).
- eing,food-post-analysis-scenario,i [Sv/(Bq/kg)] is the committed effective dose to the representative person (defined by a combination of the adult and infant dose conversion factors) from ingestion of food, milk and drinking water over 1 year, for the 'food post-analysis' scenario, per unit activity concentration of radionuclide i in food, milk or drinking water. The values are given in Table 29.
- RA<sub>i</sub>(t,mix) [unitless] is the relative activity of radionuclide i at time t after reactor shutdown for a specific radionuclide mix. It is determined by Eq. (1).
- A<sub>OIL7,hfetus,ing</sub>(t,mix) [Bq/kg] is the activity concentration in food, milk or drinking water, that for the 'food post-analysis' scenario results in a committed equivalent dose to the fetus from ingestion meeting the GC(Ingestion,h<sub>fetus,ing</sub>,1a), for a specific radionuclide mix. It is determined by Eq. (71):

$$A_{OIL7,h_{fetus,ing}}(t, mix) = = \frac{GC(Ingestion, h_{fetus,ing}, 9mo)}{\sum_{i} \left( h_{fetus,ing,food-post-analysis-scenario,i} \times RA_{i}(t, mix) \right)}$$
(71)

where:

- GC(Ingestion, h<sub>fetus,ing</sub>, 9mo) = 0.01 Sv is the generic criterion used to restrict the ingestion of non-essential food, milk and drinking water based on the committed equivalent dose to the fetus from ingestion over the full period of in utero development (as given in Table 2).
- h<sub>fetus,ing,food-post-analysis-scenario,i</sub> [Sv/(Bq/kg)] is the committed equivalent dose to the fetus from ingestion of food, milk and drinking water by the pregnant woman, over 1 year, for the 'food post-analysis' scenario, per unit activity concentration of radionuclide i in food, milk or drinking water. The values are given in Table 29.
- RA<sub>i</sub>(t,mix) [unitless] is the relative activity of radionuclide i at time t after reactor shutdown for a specific radionuclide mix. It is determined by Eq. (1).

• OIL7<sub>Cs-137</sub> (t,mix) [Bq/kg] is the activity concentration of the marker radionuclide Cs-137 in food, milk or drinking water at time t after reactor shutdown, which, for the 'food post-analysis' scenario, results in any of the generic criteria specified for OIL7 (in Table 2) being met, for a specific radionuclide mix (it needs to be used together with OIL7<sub>I-131</sub>). The values are given in Fig. 15 for all considered mixes and determined by Eq. (72):

$$OIL7_{Cs-137}(t, mix) = RA_{Cs-137}(t, mix) \times WF_{OIL7} \times DA_{OIL7}(t, mix)$$
(72)

where:

- $\circ$  RA<sub>Cs-137</sub>(t,mix) [unitless] is the relative activity of Cs-137 at time t after reactor shutdown for a specific radionuclide mix. It is determined by Eq. (1).
- The other factors are described above.

#### 3.8.6.2. Analysis

The OIL7<sub>I-131</sub>(t,mix) and OIL7<sub>Cs-137</sub> (t,mix) functions calculated in this way are displayed in Fig. 14 and Fig. 15. The red lines show the default OIL7<sub>I-131</sub> value of 1000 Bq/kg (in Fig. 14) and the default OIL7<sub>Cs-137</sub> value of 200 Bq/kg (in Fig. 15). As shown in Fig. 14, the OIL7<sub>I-131</sub>(t,mix) functions fall drastically after the 10th day, due to the short half-life of I-131, hence the need to use Cs-137 as a second marker radionuclide representative of longer lived radionuclides (Fig. 15).

In combination, the default  $OIL7_{I-131}$  and  $OIL7_{Cs-137}$  values are conservative for nearly all times after shutdown and mixes. In order to examine the combined use of  $OIL7_{I-131}$  and  $OIL7_{Cs-137}$ , Fig. 16 displays ROIL7(t,mix) [unitless], which is the maximum value of  $OIL7_{I-131}(t,mix)$  and  $OIL7_{Cs-137}(t,mix)$  relative to their corresponding default OIL value, as described in Eq. (73):

$$ROIL7(t, mix) = max \left\{ \frac{OIL7_{I-131}(t, mix)}{OIL7_{I-131}}, \frac{OIL7_{Cs-137}(t, mix)}{OIL7_{Cs-137}} \right\}$$
(73)



FIG. 14. OIL7<sub>1-131</sub>(t,mix) [Bq/kg] functions and default OIL7<sub>1-131</sub> value (in red).



FIG. 15. OIL7 <sub>Cs-137</sub>(t,mix) [Bq/kg] functions and default OIL7 <sub>Cs-137</sub> value (in red).



FIG. 16. ROIL7(t,mix) [unitless] function, displaying the combined use of OIL7<sub>1-131</sub> and OIL7<sub>Cs-137</sub>.

### **3.8.7. ΟΙL8**<sub>γ</sub>

#### 3.8.7.1. $OIL8_{\gamma}(t_m, baseline-inst)$ calculation

Calculation of the default  $OIL8_{\gamma}$  value is based on evaluation of  $OIL8_{\gamma}(t_m, baseline-inst)$  [ $\mu$ Sv/h], which is the ambient dose equivalent rate measured with the baseline monitoring instrument (baseline-inst) in front of the thyroid in contact with the skin of an infant, at time  $t_m$  after intake, for which the activity concentration of I-131 in the thyroid results in the generic criterion specified for  $OIL8_{\gamma}$  (in Table 2) being met. The values are given in Fig. 17 and determined by Eq. (74):

$$OIL8_{\gamma}(t_{m}, baseline - inst) = H^{*}_{I-131-thyroid-burden}(baseline - inst) \times \times CorF_{I-131-thyroid-burden}(t_{m}) \times WF_{OIL8_{\gamma}} \times UC \times A_{OIL8,h_{thyroid,I-131-burden}}$$
(74)

where:

- t<sub>m</sub> [s] is the time between intake and measurement of the thyroid.
- H\*<sub>I-131-thyroid-burden</sub>(baseline-inst) = 2.9E-14 (Sv/s)/Bq is the ambient dose equivalent rate in front of the thyroid as measured with the baseline instrument for monitoring the thyroid (baseline-inst) against the neck of an infant, per unit activity of I-131 (burden) in the thyroid.
- CorF<sub>I-131-thyroid-burden</sub>(t<sub>m</sub>) [unitless] is the correction factor to account for the reduction of the activity in the thyroid due to the radioactive decay and biological removal of I-131. It is displayed in Fig. 8 and determined by Eq. (20).
- $WF_{OIL8\gamma} = 1$  [unitless] is a weighting factor used to avoid the implementation of unwarranted response actions based on the default  $OIL8_{\gamma}$  value. It is described in Section 3.9.

- UC =  $3.6E+09 (\mu Sv/h)/(Sv/s)$  is the unit conversion factor from Sv/s to  $\mu Sv/h$ .
- A<sub>OIL8,hthyroid,1-131-burden</sub> [Bq] is the activity of I-131 in the thyroid (thyroid burden), which, for the 'thyroid' scenario, results in a committed equivalent dose to the thyroid of an infant meeting the GC(Urgent,h<sub>thyroid,thy-burden</sub>). It is determined by Eq. (75):

$$A_{OIL8,h_{thyroid,I-131-burden}} = \frac{GC(Urgent, h_{thyroid,thy-burden})}{h_{thyroid,I-131-thyroid-burden}}$$
(75)

where:

- $\circ$  GC(Urgent,h<sub>thyroid,thy-burden</sub>) = 0.1 Sv is the generic criterion used to take urgent response actions based on the committed equivalent dose to the thyroid from radioiodine in the thyroid (thyroid burden) (given in Table 2).
- $h_{thyroid,I-131-thyroid-burden} = 1.2E-05 \text{ Sv/Bq}$  is the committed equivalent dose to the thyroid of an infant from I-131 in the thyroid (thyroid burden). It is determined by Eq. (39).

#### 3.8.7.2. Analysis

The OIL8<sub> $\gamma$ </sub>(t<sub>m</sub>,baseline-inst) values calculated in this way are displayed as a dashed curve in Fig. 17. The bold continuous line shows the chosen default OIL8<sub> $\gamma$ </sub> value of 0.5  $\mu$ Sv/h, which is conservative for the first week after intake of the radioiodine and is considered the lowest practical ambient dose equivalent rate for monitoring the thyroid under severe emergency conditions (lower values may not be feasible due to increased variation in the background radiation).



FIG. 17.  $OIL8_{\gamma}(t_m, baseline-inst)$  [ $\mu Sv/h$ ] function (dashed curve) and default  $OIL8_{\gamma}$  [ $\mu Sv/h$ ]value (bold continuous line).

#### 3.9. WEIGHTING FACTORS

The weighting factors used in the default OIL calculations (in Section 3.8) are intended to ensure that all members of the public are effectively protected and to avoid the implementation of unwarranted actions that will do more harm than good. The weighting factors may have to be reviewed and adapted to national arrangements, conditions and/or other considerations as deemed appropriate by specialists who are familiar with dose assessment as well as with emergency preparedness and response to nuclear and radiological emergencies.

• **OIL1**<sub> $\gamma$ </sub> and **OIL2**<sub> $\gamma$ </sub>: As described in Ref. [4], OIL1<sub> $\gamma$ </sub> is used in combination with OIL2<sub> $\gamma$ </sub> to allow for the prioritization of response actions for those at highest risk (i.e. at risk of severe radiation induced health effects) and to avoid diverting limited resources from the highest priority actions during an emergency.<sup>60</sup> OIL1<sub> $\gamma$ </sub> is intended to trigger the implementation of urgent response actions in the first hours after an emergency has been declared, and, therefore, the actual exposure period is expected to be a fraction of the 7 days assumed in the calculations. Since the dose actually received during the response, if the response actions are implemented within the first day, would be 1/3 of the dose projected for an exposure period of 7 days, a weighting factor WF<sub>OIL1 $\gamma$ </sub> = 3 [unitless] is used in the calculation of OIL1<sub> $\gamma$ </sub> to ensure the best use of limited resources to protect those at highest risk.

For OIL2<sub> $\gamma$ </sub> more time is available (i.e. weeks) to implement the response actions. The weighting factor is set to WF<sub>OIL2 $\gamma$ </sub> = 1 [unitless], because the default OIL2<sub> $\gamma$ </sub> value is considered a reasonable criterion for implementing effective response actions.

OIL3, and OIL7: The generic criteria for food, milk and drinking water restrictions provided in Table II.3 of Ref. [1] were established at 1/10 of the generic criteria for taking early response actions given in Table II.2 of Ref. [1], to ensure that the dose via all exposure pathways, including ingestion, will not exceed the generic criteria for early response actions mentioned above. Careful consideration needs to be given when selecting default OIL values, taking into account the concept of operations, which assumes that those areas where ingestion of food, milk or drinking water may contribute to exceeding the generic criteria for early response actions (from all exposure pathways, including ingestion) will be within the ICPD outlined in Refs [1, 4]. In the ICPD, restrictions are initially implemented on the basis of the emergency classification system and not on the basis of OILs. It is therefore assumed that, if the concept of operations outlined in Ref. [4] is followed, the doses within the ICPD will be mainly attributable to the exposure pathways covered in the 'ground' scenario, since the appropriate restrictions on food, milk and drinking water will have already been implemented based on the declaration of the emergency. In the areas beyond the ICPD, the contribution to the total dose by all exposure pathways except ingestion of food, milk and drinking water will be significantly smaller, allowing an increase in the contribution from ingestion of food, milk and drinking water to the total dose by making use of a weighting factor of  $WF_{OIL3} = 5$  and  $WF_{OIL7} = 5$ 

<sup>&</sup>lt;sup>60</sup> While  $OIL1_{\gamma}$  is intended to protect members of the public from severe radiation induced health effects,  $OIL2_{\gamma}$  is intended to keep the doses to the members of the public below the international generic criteria given in Ref. [1] that call for taking response actions (to include medical screening) in order to further assess: (a) the small possible risk to pregnant women (i.e. the fetus); and (b) the small possible increase in the risk of radiation induced cancers.

[unitless]. The total dose received from all exposure pathways is still expected to be well below the generic criteria for early response actions, if the concept of operations outlined in Ref. [4] is implemented.

- **OIL4**<sub> $\gamma$ </sub> and **OIL4**<sub> $\beta$ </sub>: Default OIL4<sub> $\gamma$ </sub> values as high as 4  $\mu$ Sv/h (and default OIL4<sub> $\beta$ </sub> values as high as 2000 cps) would be conservative (i.e. lead to a projected dose below the urgent generic criteria). However, a weighting factor of WF<sub>OIL4 $\gamma$ </sub> = 0.5 (and WF<sub>OIL4 $\beta$ </sub> = 0.5) is used to drive the value down to about 1  $\mu$ Sv/h (and 1000 cps) to account for the following:
  - $\circ$  Activity on the skin exceeding the default OIL4<sub> $\gamma$ </sub> or OIL4<sub> $\beta$ </sub> values may indicate that the individual has inhaled or ingested sufficient radioactive iodine prior to monitoring to result in doses to the thyroid that are 2 or more times above the generic criteria, calling for registration and later medical follow-up.<sup>61</sup>
  - ο Implementing response actions at the default  $OIL4_{\gamma}$  and  $OIL4_{\beta}$  values is only expected for a limited number of members of the public following a release of radioactive material due to severe conditions at the NPP (only few individuals are expected to have sufficient radioactive material on the skin that would warrant response actions).<sup>62</sup>
  - The default OIL4 $_{\gamma}$  value (i.e. 1  $\mu$ Sv/h) is considered to be the lowest practical value usable under field conditions following a severe release of radioactive material from an LWR or its spent fuel.
- **OIL7:** See the bullet 'OIL3 $_{\gamma}$  and OIL7' above.
- **OIL8**<sub> $\gamma$ </sub>: For OIL8<sub> $\gamma$ </sub> the weighting factor was set to WF<sub>OIL8 $\gamma$ </sub> = 1 because the default OIL8<sub> $\gamma$ </sub> value is considered a reasonable criterion for implementing effective response actions and is already as low as considered practical for field conditions.

#### 3.10. CONSERVATISM

This section describes the conservatism considered in the calculation of the OILs. In this publication, 'conservative' is understood as the assumption of conditions that will result in a projected dose higher than the dose actually expected to be received under real conditions.

A variety of conservative assumptions are made whenever it is considered necessary, for example when: (a) there is insufficient information available (e.g. for the dose conversion factors for the fetus); (b) generalizations are required (e.g. for the baseline monitoring instrument or exposure scenarios); (c) required by operational considerations (e.g. limited availability of time and resources during the response); or (d) considered necessary based on the experience from past nuclear and radiological emergencies (e.g. OILs need to apply to all members of the public and not only to subgroups). The list below provides the main<sup>63</sup> conservatism considered in the calculation of the OILs with regard to their impact on the possible radiation induced health effects. The list follows the

<sup>&</sup>lt;sup>61</sup> This will most likely not be the case if urgent response actions are implemented on the basis of an emergency classification system as described in Refs [1, 2, 4, 5, 11].

<sup>&</sup>lt;sup>62</sup> Assuming that urgent response actions are implemented on the basis of an emergency classification system as described in Refs [1, 2, 4, 5, 11].

<sup>&</sup>lt;sup>63</sup> 'Main' regarding their impact on the possible radiation induced health effects. This list is not complete.

outline given in Section 3.1 and is intended to provide an overview (not a comprehensive description):

- Generic criteria (Section 3.2):
  - The generic criteria are established at doses below those at which radiation induced health effects would be expected to be observed, even in a very large exposed group composed of the most sensitive members of the public (e.g. children and pregnant women).
  - These health effects would probably not be seen at doses up to doses two or more times higher than the generic criteria of Ref. [1], in part due to the lower dose rates that will occur off the site of an NPP following a release of radioactive material [4, 19, 38, 86].
  - At the generic criteria for which response actions are expected to be taken under any circumstances to avoid or minimize the severe deterministic effects of Ref. [1], severe deterministic effects will be seen in only 5% of exposed individuals [19].
- **Radionuclide mixes (Section 3.3):** From all radionuclide mixes expected during the exposure, those resulting in the highest dose to the public are driving the default OIL values.
- **Representative person (Section 3.4):** The OILs are developed based on the dose projected or received by the representative person. In a real event, no member of the public is expected to receive a dose during an emergency approaching the one calculated for the representative person for the conditions described in the scenarios.
- Exposure scenarios and exposure pathways (Section 3.4): The details of the exposure scenarios provided in Section 3.4 are generic, because conditions influencing the exposure among individual members of the public may vary significantly. To ensure the dose received by any member of the public will not exceed the dose calculated for the exposure scenarios (i.e. most of the projected dose will be averted under actual emergency conditions), the following conservative assumptions are made:
  - For the 'ground' scenario, it is assumed that (a) apart from food restrictions, no protective actions of any type are taken and people live normally during the full period of exposure (which includes children playing on the ground and people working outside); and (b) the public is living in wooden framed houses with the least effective shielding of the range of structures provided in Ref. [70].
  - For the food related scenarios, it is assumed that (a) the affected food, milk and drinking water is consumed over 1 year; (b) 50% of the total diet (i.e. all food, milk and drinking water consumed) is assumed to be affected; (c) the most restrictive age dependent dose conversion factors and consumption rates (i.e. those for infants) are used; and (d) there will be no reduction in the activity concentrations due to normal preparations/processing of the food (e.g. peeling, washing or cooking), which in some cases substantially reduces the concentration of radionuclides in the food [65].
  - For the 'skin' scenario, it is assumed that no actions are taken to remove the material from the skin (e.g. washing) or to reduce inadvertent ingestion (e.g. keeping the hands away from the mouth).
- **Behaviour of the radionuclides (Section 3.5):** The radionuclide behaviour may vary significantly depending on specific circumstances. To ensure the factors provided in Section 3.5 cover a wide spectrum of possible cases the following conservative assumptions are made:

- The transfer factor from the ground to the air (i.e. resuspension factor) is initially conservative by around an order of magnitude when compared with those normally encountered by the public while living normally in an area.
- o It is assumed that all the deposited material is available for inadvertent ingestion.
- It is assumed that all airborne radioactive material is deposited in the pulmonary region of the lung (i.e. the respirable fraction is 1, which, in reality, would be expected to be less than one tenth of this default value in most cases [71]).
- To estimate the amount of soil loaded on the hand for normal behaviour, the values for the maximally exposed individual from Ref. [62] is used, which is roughly twice as high as for the average.
- **Dose conversion factors (Section 3.6):** The most restrictive combinations of different factors such as age, public behaviour, exposure pathways, chemical forms of the radionuclides, radionuclide behaviour, times relative to conception, acute or chronic exposure and types of absorption are considered in the calculation of the dose conversion factors. Especially the dose conversion factors or intake were approached in a conservative way:
  - The equivalent dose to the fetus from intake (inhalation and ingestion) is established at the highest dose to any fetal organ. Furthermore, the dose conversion factors for the organs from intake were the highest in Ref. [39] for both acute and chronic inhalation, considering all absorption types and vapour (when applicable), for all times of intake (including intake prior to the pregnancy). For radionuclides for which dose conversion factors are not given in Ref. [39], the one year committed equivalent dose to the uterus of an adult from intake as given in Ref. [73] was considered as the committed equivalent dose to the fetus.
  - For the effective dose from inhalation, the highest committed effective dose per unit intake for all absorption types and for members of the public older than 17 years from Table III-2E of Ref. [14] are used. This, combined with the breathing rate for an adult performing light activity, ensured the dose from inhalation would be the highest among the public.
  - For the effective dose from ingestion for each radionuclide, the highest dose is used of either (a) infant consumption rates combined with the committed effective dose to the infant, or (b) adult consumption rates combined with the committed effective dose to the adult.
- Monitoring instruments (Section 3.7): The response of the baseline beta monitoring instrument was chosen so that it is lower than the  $4\pi$  instrument efficiencies stated by instrument manufacturers for most of the surface contamination monitoring instruments, and within a factor of two of those stated by virtually all the instrument vendors. Notable exceptions are instruments designed for detection (e.g. those used by first responders or HAZMAT teams).
- **OIL(t,mix) functions and default OIL value (Section 3.8):** The default OIL values are below the OIL(t,mix) functions for most mixes and times after shutdown (Fig. 9 to Fig. 17).

# 4. SPREADSHEETS

The calculations to derive the default OIL values described in Section 3 were performed with Microsoft Excel 2010 spreadsheets, available on the attached CD. These spreadsheets are provided to support Member States in adjusting the default OIL values to their specific needs. The spreadsheets follow the general structure of this publication. Most of the calculations are provided in the 'OIL Calculation' spreadsheet. Those calculations that would considerably slow down the execution of the macros have been included in the 'Spectra' spreadsheet and serve as input for the 'OIL Calculation' spreadsheet. For any details regarding the basis of the calculations, please refer to Section 3. Before revising the default OIL values, Sections 5 needs to be considered.

#### 4.1. MACROS

The spreadsheets contain macros to automate the calculation process, and **these need to be enabled for the spreadsheet to work correctly.** Since macros pose a potential security concern, they are normally disabled by default and need to be enabled by the user.

Security Warning	g Macros have been disabled.	Enable Content Enable the macros			
X4	• (* fr				
A	В			C	

The code of the macros was developed in Microsoft Visual Basic for Applications 2010 (VBA7) and is described step by step within the code itself.

2 Microsoft Visual Basic for Appli	lications - OlL Calculation.xlsm - [Module1 (Code)]		8
Eile Edit View Insert	Format Debug Bun Tools Add-Ins Window Help	Type a question for help	• _ Ø ×
Elle Edit View Inset      Elle Edit View Inset      Project VSAProject (All Calc      VSAProject (Oll Calc      VSAProject (Oll Calc      VSAProject (Oll Calc      VSAProject (Calcable      Sheets (Control)      Sheets (Control)      Sheets (Control)      Module I      VSAProject (Calcaber      Sheets	Figmat Debug Bun Iook Add-hs Window Help [General] [General] Define the name of the sub Sub OIL_Calculation() '	Type a question for help rough I (1+4) (1.e. I4,I5,I	· - 8 ×
	'Get the OlL value as calculated for the specified time and put it in J4 (or J5, J6) Range("J" 4 1).Value = Range("C81").Value 'add +1 to i, i.e. jump through I4, I5 i = 1 + 1 m m 4		ľ

### 4.2. 'OIL CALCULATION' SPREADSHEET

This spreadsheet contains all the calculations described in Section 3, except those related to energy spectra (which are provided in the 'Spectra' spreadsheet described in Section 4.3). It is divided into four tabs, each with a specific function as described in the following sections.

# 4.2.1. Welcome tab

The 'welcome' tab provides a short description of the purpose and the scope of the spreadsheet.



# 4.2.2. Control tab

The 'control' tab allows the user to calculate and display the OILs. It is structured as follows: (a) on the left is the control area; (b) in the centre is the chart displaying the calculated OIL(t,mix) functions for OIL1<sub> $\gamma$ </sub>, OIL2<sub> $\gamma$ </sub>, OIL3<sub> $\gamma$ </sub>, OIL4<sub> $\gamma$ </sub>, OIL4<sub> $\beta$ </sub> and OIL7 (OIL8<sub> $\gamma$ </sub> is provided in the 'OIL8' tab); and (c) on the right are the results provided in tabular form.



To control the spreadsheet, perform the calculations and display the results, the user needs to follow four simple steps:

- **STEP 1** Select the OIL(t,mix) to be calculated from the dropdown menu, which provides OIL1<sub> $\gamma$ </sub>, OIL2<sub> $\gamma$ </sub>, OIL3<sub> $\gamma$ </sub>, OIL4<sub> $\gamma$ </sub>, OIL4<sub> $\beta$ </sub> and OIL7 (OIL8<sub> $\gamma$ </sub> is provided in the 'OIL8' tab).
- **STEP 2** Select the fuel type (further details on the fuel type are provided in Section 3.3). Three options are available in the dropdown menu:

**Default**: This option automatically selects the fuel type according to the mix as described in Table 3.

**Standard:** This option forces the use of the standard fuel irrespective of the selected mix.

**High burnup:** This option forces the use of the high burnup fuel irrespective of the selected mix.

**STEP 3** Select the mix (further details on the mixes are provided in Table 3). Three options are available:

**OPTION 1 - Single mix:** This option allows the selections of a single mix from the dropdown menu. It is necessary to press the button to calculate and display the OIL(t,mix) function.

**OPTION 2 - All mixes:** This option calculates and displays the OIL(t,mix) functions for all the mixes provided in Table 3.

**OPTION 3 - Your mix:** This option allows calculating and displaying the OIL(t,mix) function for your own mix. It is necessary to introduce the release fraction for each radionuclide in the table indicated by the silver arrow and to press the button.



**STEP 4** Consider saving the file with a different name and exiting without saving (to avoid modifying the master file).

### 4.2.3. Calculation tab

The 'calculation' tab contains most of the calculations described in this publication, except for the OIL8<sub> $\gamma$ </sub> calculations (provided in the 'OIL8' tab) and spectra related calculations (provided in the 'Spectra' spreadsheet). The 'calculation' tab is organized following the general structure and order of Section 3, and its fields are colour coded as described in Table 33.



# 4.2.4. OIL8 tab

The  $OIL8_{\gamma}$  calculation follows a different structure than that of the other OILs and is therefore provided in a separate tab. Since  $OIL8_{\gamma}$  is mix independent, there is no need for the user to control this tab. The colour coding is described in Table 33.



#### TABLE 33. COLOUR CODING

Colour of the field	Content of the field
Purple	Data as given in the original references (as indicated throughout this publication)
Orange	Calculations (as described throughout this publication)
Blue	Exceptions to general rules requiring special attention
Green	Results (i.e. the OIL(t,mix) values for a specific mix, fuel type and time, as applicable)
Red	Reserved for the operation of the macros (do not modify)
Yellow	Fields that cannot simply be modified (i.e. the modification will not be automatically considered in the calculations) <sup>a</sup>

<sup>a</sup> These fields are either associated with exposure periods (which are defined in the generic criteria) or with the calculation of  $TI_{grd \rightarrow air,i}(\Delta)$  and  $TI_{grd \rightarrow GI,i}(\Delta,g)$ , which, as described in Section 3.5.2 and 3.5.3, were not solved in the Excel file but obtained via numerical integration (by using a common mathematical computer program).

#### 4.3. 'SPECTRA' SPREADSHEET

This spreadsheet serves as input for the 'OIL Calculation' spreadsheet by providing those calculations that would considerably slow down the execution of the macros of the 'OIL Calculation' spreadsheet. It is divided into four tabs, each with a specific function as described in the following sections. The colour coding is described in Table 33.

#### 4.3.1. Welcome tab

The 'welcome' tab provides a short description of the purpose and scope of the spreadsheet.

#### 4.3.2. Calculation tab

The 'calculation' tab provides the calculation of the quantities to be used as input for the 'OIL Calculation' spreadsheet. The colour coding is described in Table 33.

K80	• (* _ fr																	
8	v	w	× .	¥	1	AA	AB	AC.	AD	AE	M2	AG	A/4	Ai	A,	AA.	AL.	AM
							Republic the OL Calculators'					Pputituthe CB. Celosenuri						Inputtion OR
							mather	AMBENT DOSE EQUIVALENT RATE FROM THE SKIN				IDEAL RESPONSE FACTOR OF THE BASELINE BETA						
rogeny 2	1 (V. + Cas (L.) (-)	Marta Carla	Parent	Progeny 1	Progeny 2	Mal. 40 4. 10(1)	Mananar [+]	France	Parent	Progeny t	Progeny 2	Passant	Yaar	Parent	Progeny 1	Progeny 2	Y.a. [4]	R <sub>m.L</sub> (baseline-inst)
a freeday	IOva vitaxefit	Silversitherer*1	10-mailtonett	DOverstower's	10yauthourf1	SOuth House's	Debuller's	15vautents	T-mailton's	STANKIN PL	ST-AUBOW'S	Internation 1	100.01	100'1'	101'1'2		acres	Insetting
	1.92-14	3.38-10	And the second	England and a		3.38-10	3.32-14	3.32-18		and the second		3.56-18	9.52-01				9.58-21	1.42-01
	3.62-56	3.25-10		1000		3.26-10	3.22-14	3.26-21		1 10 10		526-21	9.46-01	10000			3.42-01	1.46-01
	3.66.13	316.10	1.10.10	3.98-59		3.02.10	3.27.14	248.17	4.06-00	4 78-22		2.00.17	847.01	7.00.01	100.01		3.47.51	148.21
	5.32-16	3.26-10				5.26-10	3.22-14	1.15-13				1.16-19	9.52-01				8.52-01	1.46-01
	2.06-12	ND	3.05-11	7.06-14		3.06-11	5.96-15	ND	2.86-17	2:96-17		\$.12-17	10	0.26-01	1.86-81		1.06+00	1.56-01
136-13	9.56-12	ND	3.45-10	0.06-00	2.86-10	8.36-10	6.56-14	ND	3.36-17	0.0E-00	2.66-17	5.96-17	10	9.86-01	0.06-00	836-01	1.96-00	2.96-01
	1.06-12	80	2.00-10	0.05+00		2.36-10	118.18	10	2.00-10	5.00-18		3.48.47	100	3.02-01	0.05-01		1.66-00	8.16.41
	2.06-12	256-10	a			2.58-10	2.66-14	3.06-17	e			3.06-17	115-00	and out			1.16-00	1.76-01
	6.16-13	ND	0.05-00	4.58-10		4.56-10	4.52-54	ND	0.05-00	8.12-18		8.18-18	ND	0.05+00	10-38.0		196-01	1.58-01
	1.18-12	7.05-11				7.06-11	7.36-15	3.25-15				3.28-18	8.55-01	1	1.1.1		8.02-01	1.06-01
	0.00-14	100	3.96-12	1.45-10		1.46-18	1.46-14	10	2.38-18	2.16-19		2.06-18	100	5.96-01	8.16-61		1.42-00	216-41
	4.10-13	10	1.15.10	3.05.10		2.18.10	3.16.14	NO	2.86-18	2.28-18		5.05.18	80	6.65-01	3.45-21		1.56-00	1.96-01
	3.36-12	5.28-11				5.26.11	7.16.15	5.42-17				3.45-17	8.06-01				0.55-01	1.25-01
	8.46-12	ND	6.25-13	2.46-10		2.76-10	3.0E-14	ND	1.36-17	4.52-17		9.86-17	80	8.46-41	9,25-01		1.4E+00	2.16-01
	6.56-12	0.76-11				9.76-11	1.16-14	1.66-17				1.02-17	7.66-01				7.86-81	1.22-01
	4 18.12	105-10				3.05.10	345.14	348-17				9 45.17	\$77.01				\$75.41	1.55.01
	1.58-12	2.06-10				2.05-10	2.28.14	5.38-17				5.32-17	6.52-01				6.52-01	1.36-01
	4.45-12	@.0E-11				8.86-11	1.16-14	6.08-17				6.06-17	5.96-01				5.96-01	8.86-82
	5.8E-12	3.4E-11	111201021	10000		3.48-11	6.4E-15	7.86-17	10000			7.86-17	7.36-01				7.38-01	1.18-01
	4 12-11	80	1.05-10	0.05-00		1.05-10	1.55-14	10	7.96-23	2.36-17		2.56-17	10	7.65-01	1.05-01		3.62-01	1.36-01
	0.05+00	596.11	1.16/10			5.96-11	6.02-15	3.75-18	0.0010	e raetr		3.75.18	9.16-01	w.eb-01	100.01		\$ 18-41	1.45.01
	5.96-12	2.56-10				2.58-10	2.65-14	1.36-17				1.36-17	9.38-01				9.26-01	1.45-01
196-14	3.06-14	ND	7.26-12	8.82-14	4.36-18	4.36-10	4.32-14	ND	1.16-18	1.42-18	3.42-19	2.06-18	10	5.36-01	145-04	\$ 22-01	1.56-00	2.36-01
	2.26-20	2.05-10				2.06-10	2.05-14	3.40-25				141.25	8.86-01				8.05-01	13641
	145-11	3.05-11				3.05.11	5.05-15	1.18-17				1.16-17	1.35+00				1.35+00	2.05-01
	1.78-17	3.45.18				3.46-18	6.06-17	6.56-19				6.56-19	1.05-03				1.05-00	1.56-04
	6.7E-15	0.06+00				0.06+00	2.36-17	2.86-19				2.06-19	3.9E-04				3.96-04	5.86-05
	1.66-17	1.36-16				1.36-16	5.72-17	6.12-19				6.12-19	8.08-04				6-DE-24	128-04
	0.00+00	0.05+00				0.05+00	3.66-20	1 41-22				1.45-22	1 16-06				1 10.00	1,00-04
	7.42-17	1.02-16				1.05-16	5.46-17	6.42-19				6.42-19	4.05-04				4.02-04	6.16-05
	\$.76-17	238-15				2.06-15	5.16-17	5.58-19				1.56-19	3.18-04				3.10-04	4.62-05
													9.75-01 0.66-01				878-01	1.52-01 1.32-01
						4		-										
							Roman	3.68-02	E E									
							G.,.	8.96-02	(unitees)									
							E/handing inst)	0.30	Innunte									

#### 4.3.3. RAD tab

The 'RAD' tab provides the absolute yields (or intensities) and mean or discrete energies of the radiations emitted by the radionuclides as required in the calculation of the OILs. The interpolation of data (to evaluate  $ConF_{\phi \to H^*(10), E_j}$  from Eq. (41)) was performed by a polynomial of fifth order as described in the charts.



# 4.3.4. BET tab

The 'BET' tab provides the beta spectra as required in the calculation of the OILs.

1				1 10	-				-	Spectra.alum - Microsoft Escal									k	ole n
The Hone	inset 8	Page Lapour	Farmin	i Data	Entra	View Developer													9	0000
W42	+ (*)	f.																		
A	В	C	D	E	F	G	н	1	1	ĸ	L	M	N	0	P	Q	R	5	T	U .
2																				
3			BET	A SPEC	TRA															
5			ID	RN	E	ng	$n_{\beta}(E_e;E_e+dE_e)$	log10 E.	d <sup>e</sup> st,v (E <sub>e</sub> )	$n_{g}(E_{e};E_{e}+dE_{e}) * d^{e}_{SE,V}(E_{e})$										
6			-	-	[MeV]	[(MeV <sup>-1</sup> Bq <sup>-1</sup> s <sup>-3</sup> )]	[(Bq <sup>-1</sup> s <sup>-1</sup> )]	100	[Gy cm <sup>2</sup> ]	[(Gy/s)/(Bq/cm2)]										
7			1	P.32	0	2.8E-01	2.85.05		0.0E+00	0.0E+00										
8			1	P-32	0.0001	2.8E-01	2.8E-06		0.0E+00	0.0E+00										
9			1	P-32	0.00011	2 8E-01	2.8E-06	1 i i i i i i i i i i i i i i i i i i i	0.0E+00	0.0E+00										
10			1	P-32	0.00012	2.8E-01	2.8E-06		0.0E+00	0.0E+00										
11			1	P-32	0.00013	2.8E-01	2.8E-06		0.0E+00	0.0E+00										
12			1	P.32	0.00014	2.8E-01	2.8E-06		0.0E+00	0.0E+00										
13			1	P-32	0.00015	2.8E-01	2.8E-06		0.0E+00	0.0E+00										
14			1	P-32	0.00016	2.8E-01	5.5E-06	114	0.0E+00	0.0E+00										
15			1	P-32	0.00018	2.8E-01	5.5E-06	-	0.0E+00	0.0E+00										
16			1	P-32	0.0002	2.8E-01	5.5E-06		0.0E+00	0.0E+00										
17			1	P-32	0 00022	2.8E-01	5.5E-06		0.0E+00	0.0E+00										
18			1	P-32	0.00024	2.8E-01	5.5E-06	-	0.0E+00	0.0E+00										
19			1	P-32	0.00026	2.8E-01	5.5E-06		0.0E+00	0.0E+00										
20			1	P-32	0.00028	2.8E-01	5.5E-06		0.0E+00	0.0E+00										
21			1	P-32	0.0003	2.8E-01	5.5E-06	1.1	0.0E+00	0.0E+00										
22			1	P-32	0.00032	2.8E-01	1.1E-05		0.0E+00	0.0E+00										
23			1	P-32	0.00036	2.8E-01	1.1E-05		0.0E+00	0.0E+00										
24			1	P-32	0.0004	2.8E-01	1.4E-05		0.0E+00	0.0E+00										
25			1	P-32	0.00045	2.8E-01	1.4E-05		0.0E+00	0.0E+00										
26			1	P-32	0.0005	2.8E-01	1.4E-05		0.0E+00	0.0E+00										
27			1	P-32	0.00055	2.8E-01	1.4E-05		0.0E+00	0.0E+00										
28			1	P-32	0.0006	2.8E-01	1.4E-05		0.0E+00	0.0E+00										
29			1	P-32	0.00065	2.8E-01	1.4E-05		0.0E+00	0.0E+00										
30			1	P-32	0.0007	2.8E-01	1.4E-05		0.0E+00	0.0E+00										
31			1	P-32	0.00075	2.8E-01	1.4E-05	- 1a	0.0E+00	0.0E+00										
32			11	P-32	0.0008	2.8E-01	1.4E-05		0.0E+00	0.0E+00										
33			1	P-32	0.00085	2.8E-01	1.4E-05		0.0E+00	0.0E+00										
34			1	P-32	0.0009	2.8E-01	2.8E-05	1.0	0.0E+00	0.0E+00										
35			1	P-32	0.001	2.8E-01	2.8E-05		0.0E+00	0.0E+00										
36			11	P-32	0.0011	2.8E-01	2.8E-05		0.0E+00	0.0E+00										
37			1	P.32	0.0012	2 8E-01	2.8E-05		0.0E+00	0.0E+00										
* * * * Welcon	ne Crister	en KAD	BUT /E	1																1 10
Ready																		비교교 대	P5 (=)	

# 4.4. REVISING THE SPREADSHEETS

Before modifying the spreadsheets, it is strongly recommended to acquire a complete understanding of (a) the derivation of the OILs as described in Section 3; (b) the structure of the spreadsheet (including its content, its dependencies and the macros); and (c) the recommendations given in Section 5. It is easy for users to introduce errors, and they may be difficult to audit. The most frequent mistakes when revising the spreadsheet are:

- Not paying attention to fields with exceptions to general rules;
- Modifying fields needed by the macros or not updating the macros according to the changes in the spreadsheet;
- Introducing rows or columns, or modifying their order, without attention to the dependencies and macros;
- Not having a thorough understanding for the derivation of the OILs.

# 5. REVISING THE DEFAULT OIL VALUES

This section provides (a) general considerations concerning the revision of the default OIL values (in Section 5.1) and (b) instructions for adapting the default  $OIL4_{\beta}$  value to a specific instrument (in Section 5.2).

### 5.1. GENERAL CONSIDERATIONS

Member States may want to revise the default OIL values to consider different underlying assumptions or another methodological approach than the ones used by the IAEA in this publication. These changes need to be built on a defensible basis and clearly explained to all relevant interested parties during the preparedness stage. Introducing changes to the default OIL values given in this publication requires experts with experience in dose assessment and emergency preparedness and response, as well as a clear understanding of the methodology for deriving the OILs. Section 3 provides the methodology used by the IAEA, which can be employed as a basis for revision.

Member States may also want to adapt the default OIL values to the specific conditions of the accident once sufficient information becomes available. In this case, arrangements need to be in place to explain these changes to the public in a plain and understandable form, as described in para. 5.69 and 5.70 of the IAEA Safety Standards Series No. GSR Part 7 [1]. Operational criteria (like the OILs) are only to be changed during the emergency if there is clear evidence that the revised criteria will do more good than harm, considering both radiological and non-radiological consequences, and when the situation is clearly understood (e.g. exposure scenarios, public behaviour and radionuclide mixes are well characterized). Failure to do so may lead to confusion and scepticism on the part of decision makers and the public, potentially resulting in unwarranted actions being taken.

In order to support Member States, the key elements that need to be considered to ensure that the revised OIL values are built on a defensible basis have been highlighted in Table 34 and Table 35. Table 34 addresses general considerations for increasing or decreasing the default OIL values, while Table 35 addresses generic considerations applicable to all or multiple OILs.

# TABLE 34. GENERAL CONSIDERATIONS FOR INCREASING OR DECREASING THE DEFAULT OIL VALUES

OIL	Change	Considerations	Example when the revision may be justified
ΟIL1 <sub>γ</sub>	Increase	The use of higher default $OIL1_{\gamma}$ values (e.g. twice the $OIL1_{\gamma}$ ) needs to be considered with great care to ensure that all members of the public are effectively protected against severe radiation induced health effects. Living in an area above the current default $OIL1_{\gamma}$ value for more than a day could result in radiation induced health effects.	Evacuation is not possible for all areas exceeding the default value, and other response actions are effectively implemented that will protect the public.
	Decrease	The use of lower default $OIL1_{\gamma}$ values (e.g. one half of $OIL1_{\gamma}$ ) could result in the diversion of limited resources from the highest priority actions, which may cause more harm than good. However, it may be justified if sufficient resources are available to cope with the higher number of people affected by the response actions.	Lowering the default value does not result in a significant increase in the number of affected people.
	Increase	The use of higher default $OIL2_{\gamma}$ values (e.g. twice the $OIL2_{\gamma}$ ) may be considered if the exposure conditions are well understood early in the emergency, and the generic criteria for reducing the risk for stochastic effects will not be exceeded during the anticipated exposure period (i.e. before response actions are fully implemented for all members of the public).	The behaviour of the public is well characterized and is estimated to result in lower doses than those estimated in this publication.
OIL2γ	Decrease	The use of lower default $OIL2_{\gamma}$ values (e.g. one half of $OIL2_{\gamma}$ ) may be considered if it is clearly demonstrated to be justified and optimized with account taken of non-radiological effects (e.g. social or economic effects). A reduction of the default $OIL2_{\gamma}$ value may be justified if actual conditions during the emergency would result in higher doses than those assumed in the calculation of the OILs.	The impacted area has a very low population density.
OIL3 <sub>7</sub>	Increase	The use of higher default $OIL3_{\gamma}$ values (e.g. twice the $OIL3_{\gamma}$ ) may be justified to avoid other adverse effects caused by the restriction of food, milk or drinking water. The values may also be increased if the exposure conditions are well understood early in the emergency and the generic criteria for reducing the risk for stochastic effects will not be exceeded as a result of the consumption of these products, taking all exposure pathways into account.	If only certain crops (for example due to the time of the year) are affected, it may be justified to increase the default value.
	Decrease	The default $OIL3_{\gamma}$ value is as low as considered to be reasonably measurable and practical under field and emergency conditions.	The areas affected by the release of radioactive material are relatively small.

OIL	Change	Considerations	Example when the revision may be justified
OIL4 <sub>γ</sub>	Increase	The use of higher default $OILA_{\gamma}$ values (e.g. twice the $OILA_{\gamma}$ ) may be considered if it is difficult to apply the default OIL value under emergency conditions.	The resources available for monitoring are limited and there are higher priorities.
	Decrease	The default $OIL4_{\gamma}$ value is as low as considered to be reasonably measurable and practical under field and emergency conditions. Lower values may be used if demonstrated to be justified and feasible for the effective implementation of response actions, considering the diversion of limited resources and unwarranted anxiety among those being monitored.	The location where the monitoring is performed has a very low background.
	Increase	The use of higher default $OIL4_{\beta}$ values (e.g. twice the $OIL4_{\beta}$ ) may be considered if it is difficult to apply the default OIL value under emergency conditions.	The number of affected people exceeds the capabilities of the response.
OIL4 <sub>β</sub>	Decrease	Lower $OIL4_{\beta}$ values (e.g. one half of $OIL4_{\beta}$ ) may be used if demonstrated to be justified and feasible for the effective implementation of response actions, considering the diversion of limited resources and unwarranted anxiety among those being monitored.	The instrument response is higher than assumed in the calculations.
OIL7	Increase	The use of higher default OIL7 values (e.g. twice the OIL7) may be considered if the exposure conditions are well understood (e.g. if enough samples are taken and the behaviour of the public, exposure pathways and scenarios are well characterized) and the generic criteria for reducing the risk for stochastic effects will not be exceeded as a result of the consumption of these products, taking all exposure pathways into account.	The percentage of affected food, milk and drinking water and the consumption behaviour of the public is well characterized and results in lower doses than expected.
	Decrease	The use of lower default OIL7 values (e.g. one half of OIL7) may be considered if the exposure conditions are well understood (e.g. if enough samples have been taken and the behaviour of the public, exposure pathways and scenarios are well characterized). It needs to be considered with great care to ensure that it causes more good than harm, with account taken of non-radiological effects.	A specific affected product is consumed in larger quantities than expected.
OIL8 <sub>7</sub>	Increase	The use of higher default $OIL8_{\gamma}$ values (e.g. twice the $OIL8_{\gamma}$ ) may be considered, especially if the properties (e.g. efficiency) of the monitoring instruments that will be used are known, but it needs to be ensured that these values allow for all members of the public to be effectively identified for registration and medical follow-up.	The instrument efficiency is higher than the one assumed in the calculations.
	Decrease	The default $OIL8_{\gamma}$ value is as low as considered to be reasonably measurable and practical under field and emergency conditions.	An improved monitoring technique allows using lower values.

# TABLE 35. GENERAL CONSIDERATIONS FOR REVISING THE DEFAULT OIL VALUES (*The answer to all questions needs to be 'yes' for the revised OILs to be defensible*)

Changes in the:	Considerations
Objective of the OILs	Does the new objective allow the prompt and effective implementation of emergency response actions on the basis of monitoring results readily available during a nuclear or radiological emergency? Does the new objective allow avoiding a discernible increase in the incidence of radiation induced health effects? Does the new objective allow avoiding the delay of decision making due to limited availability of information (as expected early on in an emergency)? Does the new objective allow avoiding response actions that would result in more harm than good because of an overly conservative approach?
Generic criteria	<ul><li>Have dosimetric criteria (as described in Ref. [1]) been used that provide a solid foundation for the implementation of response actions?</li><li>Do the new dosimetric criteria ensure that there will be no discernible increase in radiation induced health effects?</li><li>Are the new dosimetric criteria reasonable, i.e. do they provide a solid basis for a justified and optimized protection strategy?</li><li>Do the new dosimetric criteria consider individual organ doses and resulting health effects, and not only the effective dose?</li></ul>
Radionuclide mix	Have all radionuclide mixes which are expected to be released or have been released, and which may be of significant impact, been considered? Have all possible releases of radioactive material that may result from a severe emergency been considered, including those resulting from beyond design accidents? Have you considered the fact that, following a release of radioactive material, the measured radionuclide mix will change depending on the location and time (as described in Section 2.2)?
Individuals being exposed	Have all members of the public that may be exposed (including those most sensitive to radiation) been considered? Has it been clearly stated that all members of the public (including those most sensitive to radiation) have been considered?
Exposure scenarios and pathways	Have all relevant exposure scenarios been considered? Have all relevant exposure pathways been considered? Have the behaviour of the public and other factors affecting public exposure been considered for the different scenarios and pathways?
Behaviour of the • radionuclides	Have all relevant aspects of radionuclide behaviour been considered that may have a significant impact on the dose or the OILs?
Dose conversion factors	Have dose conversion factors been developed (or considered) for each radionuclide, dose quantity, exposure pathway, exposure scenario and for all exposed individuals? Have both effective and equivalent dose conversion factors been considered?
Instrument response	Are the OILs applicable to the instruments being used? Are the instruments appropriate for the intended measurement?

Changes in the:	Considerations
OIL(t,mix) functions and default OIL value	Have time and mix dependent OIL(t,mix) functions been developed for all considered radionuclide mixes? Has a reasonably conservative and justified default OIL value been chosen considering the overall protection strategy and operational requirements?
Response actions	Do the response actions take into account the overall protection strategy, the contribution from the different exposure pathways for the specific exposure scenario, how the response action would contribute to reducing the dose and whether the response action is feasible and justified?
Communicating the response actions to decision makers	Have preparations been made for communicating the response actions to decision makers and public information officers?
Changes during the emergency	Is there enough evidence that justifies changing the default OIL values?
Other	Is the approach reasonably conservative, i.e. does it ensure that the response actions protect the public effectively from the radiological health hazard? Are the response actions justified and optimized within the overall protection strategy?

#### 5.2. REVISING OIL4 $_{\beta}$ FOR A SPECIFIC BETA MONITORING INSTRUMENT

Although the ambient dose equivalent rate  $OIL4_{\gamma}$  is sufficient to assess the levels of radioactive material on the skin for a release of radioactive material from an LWR or its spent fuel, in some cases, beta monitoring instruments may be used instead of ambient dose equivalent rate instruments. Therefore, a default  $OIL4_{\beta}$  value is also provided in this publication.

The default  $OIL4_{\beta}$  value given in Section 2 can be used, without revision, with any instrument that meets the suitability criteria described in Section 3.7.2.3. However, beta monitoring instruments that are suitable could have considerably different responses, in cps, to the same levels of deposition, primarily due to differences in the efficiency and effective window area. The default  $OIL4_{\beta}$  value needs to be adapted to specific beta monitoring instruments during the preparedness stage for the response teams' instruments by following two steps:

Step 1: Calculate δ<sub>β</sub>(inst) [unitless], which is the ratio factor of the beta monitoring instrument by using either (a) its calibration factor [(Bq/cm<sup>2</sup>)/cps] (if provided for one of the radionuclides listed in Table 31); (b) its instrument coefficient [cps/(Bq/cm<sup>2</sup>)]; or (c) the effective window area [cm<sup>2</sup>] in combination with the response factor [cps/Bq], as described in Eq. (76):

$$\delta_{\beta}(\text{inst}) = \frac{\text{CalF}_{\beta,i}(\text{baseline} - \text{inst})}{\text{CalF}_{\beta,i}(\text{inst})} = \frac{C_{\beta,i}(\text{inst})}{C_{\beta,i}(\text{baseline} - \text{inst})} \cong$$

$$\cong \frac{a_{\text{eff},\beta}(\text{inst}) \times R_{4\pi,\beta,i}(\text{inst})}{a_{\text{eff},\beta}(\text{baseline} - \text{inst}) \times R_{4\pi,\beta,i}(\text{baseline} - \text{inst})}$$
(76)

where:

- $\circ$  CalF<sub> $\beta$ ,i</sub>(baseline-inst) [(Bq/cm<sup>2</sup>)/cps] is the calibration factor of the baseline beta monitoring instrument for radionuclide i. The values are given in Table 31.
- $\circ$  CalF<sub> $\beta$ ,i</sub>(inst) [(Bq/cm<sup>2</sup>)/cps] is the calibration factor of the beta monitoring instrument (inst) for one of the radionuclides listed in Table 31.
- $C_{\beta,i}$ (baseline-inst) [cps/(Bq/cm<sup>2</sup>)] is the beta baseline monitoring instrument coefficient for radionuclide i. It is given in Table 31.
- $\circ$  C<sub> $\beta,i</sub>(inst) [cps/(Bq/cm<sup>2</sup>)] beta monitoring instrument coefficient for radionuclide i.</sub>$
- $a_{eff,\beta}$ (baseline-inst) = 15 cm<sup>2</sup> is the effective window area (e.g. sensitive or radiation inlet area) of the baseline beta monitoring instrument and is described in Section 3.7.2.1.1.
- $\circ$  a<sub>eff, $\beta$ </sub>(inst) [cm<sup>2</sup>] is the effective window area (e.g. sensitive or radiation inlet area) of the beta monitoring instrument (inst).
- $\circ$  R<sub>4 $\pi,\beta,i$ </sub>(baseline-inst) [cps/Bq] is the ideal response factor of the baseline beta monitoring instrument, which reflects the response [cps] for each nuclear transformation per second [Bq] of radionuclide i on the surface, for 4 $\pi$  geometry, assuming ideal conditions (e.g. no self-absorption or air absorption) and the instrument window being very close to the surface.<sup>64</sup> It is given in Table 32.

<sup>&</sup>lt;sup>64</sup> The numerical value of  $R_{4\pi,\beta,i}$  (baseline-inst) corresponds to (or is often referred to) the  $4\pi$  efficiency stated by many manufacturers.  $2\pi$  efficiency is equal to 2 times the  $4\pi$  efficiency.

- $\circ$  R<sub>4π,β,i</sub>(inst) [cps/Bq] is the response factor of the beta monitoring instrument, which reflects the response [cps] for each nuclear transformation per second [Bq] of radionuclide i on the surface, for 4π geometry, assuming ideal conditions (e.g. no self-absorption or air absorption) and the instrument window being very close to the surface. It is often provided by the manufacturer as the 4π efficiency of the instrument. If the manufacturer only provides a 2π geometry efficiency, divide this by 2 to get the 4π efficiency.<sup>65</sup>
- Step 2: Calculate the OIL4<sub>β</sub>(inst) [cps] for a specific beta monitoring instrument (inst) by using Eq. (77):

$$OIL4_{\beta}(inst) = \delta_{\beta}(inst) \times OIL4_{\beta}$$
(77)

where:

- $\circ$  δ<sub>β</sub>(inst) [unitless] is the ratio factor of the beta monitoring instrument (inst), as determined in the previous step in Eq. (76).
- $\circ~ OIL4_{\beta}$  = 1000 cps is the default OIL4\_{\beta} value for baseline instrument, as given in Section 2.6.2.

 $<sup>^{65}</sup>$  4 $\pi$  efficiency applies to emissions in all directions and 2 $\pi$  efficiency applies to emissions toward the detector.

#### APPENDIX

#### **RELATIONSHIP WITH OTHER IAEA PUBLICATIONS**

This publication (EPR-NPP-OILs) is part of the IAEA Emergency Preparedness and Response Series and is related to the following IAEA publications:

### **IAEA Safety Standards Series:**

- General Safety Requirements No. GSR Part 7 (2015) [1] provides the requirements for establishing an adequate level of preparedness and response for a nuclear or radiological emergency. It is a revised and updated version of the IAEA Safety Standards Series No. GS-R-2 issued in 2002.<sup>66</sup> It establishes in para. 4.28(4) the requirement to develop operational criteria (including OILs) for initiating the different parts of an emergency plan and taking response actions as part of the overall protection strategy. It also provides in its Appendix II generic criteria in terms of projected or received dose at which response actions need to be taken. These generic criteria are the basis for the default OIL values derived in this (EPR-NPP-OILs) publication (as describe in Section 3.2).
- General Safety Guide No. GSG-2 (2011) [2] elaborates the framework of generic and operational criteria for taking effective response actions in a nuclear or radiological emergency, including numerical values of these criteria. The default operational criteria of GSG-2 are applicable to any radiological or nuclear emergency, i.e. they are not specific to any particular type of emergency. While the generic criteria and OILs provided in GSG-2 are consistent with GSR Part 7 [1], their scope has been broadened in GSR Part 7. The OILs for LWR emergencies and guidance provided in this (EPR-NPP-OILs) publication considers the latest guidance given in GSR Part 7 as well as the lessons learned since the publication of GSG-2 (e.g. from the accident at the Fukushima Daiichi nuclear power plant).
- Safety Guide No. GS-G-2.1 (2007) [3] provides additional guidance on meeting specific EPR safety requirements, describes appropriate responses to a range of nuclear and radiological emergencies, and gives background information on past experience. It also discusses the concept of OILs and provides further technical basis on their development.

# IAEA EPR Series publications and IAEA TECDOCs:

• IAEA EPR-NPP Public Protective Actions (2013) [4] provides practical guidance on the actions necessary to be taken to protect the public in the event of an emergency involving actual or projected severe damage to the fuel or spent fuel of an LWR. It provides a basis for developing the tools that would be needed in taking actions in response to such an emergency. EPR-NPP Public Protective Actions can be used as a basis for developing

<sup>&</sup>lt;sup>66</sup> The IAEA Safety Standards Series No. GS-R-2 (2002) was superseded by the IAEA Safety Standards Series No. GSR Part 7 [1]. GS-R-2 provided generic intervention levels and action levels for taking an intervention in a nuclear or radiological emergency. However, they have been superseded by the concept of generic and operational criteria and the protection strategy addressed in GSG-2 [2] and GSR Part 7 [1].

protection strategies for this type of emergency consistently with GSR Part 7 [1]. However, the OILs, associated response actions and methodology provided in this publication (EPR-NPP-OILs) update those in the EPR-NPP Public Protective Actions publication (in a minor way) in consideration of the latest safety requirements (GSR Part 7).

- IAEA-TECDOC-955 (1997) [5] provides technical procedures for determining response actions for the public and controlling doses to emergency workers for accidents at a nuclear reactor. These include: procedures for classifying an accident, projecting consequences, coordinating and interpreting environmental monitoring, determining public protective actions and controlling emergency worker doses. The technical guidance for response actions and OILs contained in IAEA-TECDOC-955 are superseded by Refs [2, 3, 4] and by this publication (EPR-NPP-OILs).
- IAEA EPR-MEDICAL (2005) [77] serves as a practical resource for planning the medical response to a nuclear or radiological emergency. It provides practical tools and generic procedures for use by emergency medical personnel and guidance for the development of medical response capabilities. The OILs it provides for skin contamination are consistent with the OIL4 provided in this publication; however, they are intended for any type of radiation emergency and are not specific to LWR emergencies.

Further details of the principles and arrangements to respond effectively to a severe nuclear or radiological emergency can be found in the IAEA Safety Fundamentals [87], Safety Requirements [1], Safety Guides [2, 3], EPR Series publications, TECDOCs and Accident Reports provided at http://www-pub.iaea.org/MTCD/publications/publications.asp. The IAEA Safety Glossary Ref. [88] defines and explains technical terms used in IAEA safety standards and other safety related IAEA publications, and provides information on their usage

#### REFERENCES

- [1] FOOD AND AGRICULTURE ORGANIZATION OF THE UNITED NATIONS, INTERNATIONAL ATOMIC ENERGY AGENCY, INTERNATIONAL CIVIL AVIATION ORGANIZATION, INTERNATIONAL LABOUR ORGANIZATION, INTERNATIONAL MARITIME ORGANIZATION, INTERPOL, OECD NUCLEAR ENERGY AGENCY, PAN AMERICAN HEALTH ORGANIZATION, PREPARATORY COMMISSION FOR THE COMPREHENSIVE NUCLEAR-TEST-BAN TREATY ORGANIZATION, UNITED NATIONS ENVIRONMENT PROGRAMME, UNITED NATIONS OFFICE FOR THE COORDINATION OF HUMANITARIAN AFFAIRS, WORLD HEALTH ORGANIZATION, WORLD METEOROLOGICAL ORGANIZATION, Preparedness and Response for a Nuclear or Radiological Emergency, IAEA Safety Standards Series No. GSR Part 7, IAEA, Vienna (2015).
- [2] FOOD AND AGRICULTURE ORGANIZATION OF THE UNITED NATIONS, INTERNATIONAL ATOMIC ENERGY AGENCY, INTERNATIONAL LABOUR OFFICE, PAN AMERICAN HEALTH ORGANIZATION, WORLD HEALTH ORGANIZATION, Criteria for Use in Preparedness and Response for a Nuclear or Radiological Emergency, IAEA Safety Standards Series No. GSG-2, IAEA, Vienna (2011).
- [3] FOOD AND AGRICULTURE ORGANIZATION OF THE UNITED NATIONS, INTERNATIONAL ATOMIC ENERGY AGENCY, INTERNATIONAL LABOUR OFFICE, PAN AMERICAN HEALTH ORGANIZATION, UNITED NATIONS OFFICE FOR THE COORDINATION OF HUMANITARIAN AFFAIRS, WORLD HEALTH ORGANIZATION, Arrangements for Preparedness for a Nuclear or Radiological Emergency, IAEA Safety Standards Series No. GS-G-2.1, IAEA, Vienna (2007).
- [4] INTERNATIONAL ATOMIC ENERGY AGENCY, Actions to Protect the Public in an Emergency due to Severe Conditions at a Light Water Reactor, EPR-NPP Public Protective Actions 2013, IAEA, Vienna (2013).
- [5] INTERNATIONAL ATOMIC ENERGY AGENCY, Generic Assessment Procedures for Determining Protective Actions during a Reactor Accident, IAEA-TECDOC-955, IAEA, Vienna (1997).
- [6] INTERNATIONAL ATOMIC ENERGY AGENCY, Lessons Learned from the Response to Radiation Emergencies (1945–2010), EPR-Lessons Learned 2012, IAEA, Vienna (2012).
- [7] INTERNATIONAL ADVISORY COMMITTEE, The International Chernobyl Project: Technical Report, IAEA, Vienna (1991).
- [8] INTERNATIONAL ATOMIC ENERGY AGENCY, The Fukushima Daiichi Accident, 6 vols, IAEA, Vienna (2015).
- [9] INVESTIGATION COMMITTEE ON THE ACCIDENT AT THE FUKUSHIMA NUCLEAR POWER STATIONS OF TOKYO ELECTRIC POWER COMPANY, Final Report, Cabinet Secretariat of the Government of Japan, Tokyo (2012).
- [10] THE NATIONAL DIET OF JAPAN, FUKUSHIMA NUCLEAR ACCIDENT INDEPENDENT INVESTIGATION COMMISSION, The Official Report of the Fukushima Nuclear Accident Independent Investigation Commission, National Diet of Japan, Tokyo (2012).

- [11] INTERNATIONAL ATOMIC ENERGY AGENCY, Method for Developing Arrangements for Response to a Nuclear or Radiological Emergency, EPR-METHOD 2003, IAEA, Vienna (2003).
- [12] INTERNATIONAL ATOMIC ENERGY AGENCY, Safety Assessment for Facilities and Activities, IAEA Safety Standards Series No. GSR Part 4 (Rev. 1), IAEA, Vienna (2016).
- [13] INTERNATIONAL NUCLEAR SAFETY GROUP, A Framework for an Integrated Risk Informed Decision Making Process, INSAG-25, IAEA, Vienna (2011).
- [14] EUROPEAN COMMISSION, FOOD AND AGRICULTURE ORGANIZATION OF THE UNITED NATIONS, INTERNATIONAL ATOMIC ENERGY AGENCY, INTERNATIONAL LABOUR ORGANIZATION, OECD NUCLEAR ENERGY AGENCY, PAN AMERICAN HEALTH ORGANIZATION, UNITED NATIONS ENVIRONMENT PROGRAMME, WORLD HEALTH ORGANIZATION, Radiation Protection and Safety of Radiation Sources: International Basic Safety Standards, IAEA Safety Standards Series No. GSR Part 3, IAEA, Vienna (2014).
- [15] JOINT FAO/WHO FOOD STANDARDS PROGRAMME, CODEX ALIMENTARIUS COMMISSION, Codex General Standard for Contaminants and Toxins in Food and Feed, Schedule 1 - Radionuclides, CODEX STAN 193e-1995, CAC, Rome (2006).
- [16] McKENNA, T., BUGLOVA, E., KUTKOV, V., Lessons learned from Chernobyl and other emergencies: Establishing international requirements and guidance, Health Phys. 93 5 (2007) 527–537.
- [17] GONZÁLEZ, A. J., "The radiation health consequences of Chernobyl: the dilemma of causation", Proc. Helsinki Symp. on Nuclear Accidents, Liabilities and Guarantees, OECD Nuclear Energy Agency, Paris (1993) 25–55.
- [18] INSTITUT DE RADIOPROTECTION ET DE SÛRETÉ NUCLÉAIRE, Éléments de sûreté nucléaire, IRSN, Paris (1996).
- [19] INTERNATIONAL ATOMIC ENERGY AGENCY, WORLD HEALTH ORGANIZATION, Development of an Extended Framework for Emergency Response Criteria: Interim Report for Comments, IAEA-TECDOC-1432, IAEA, Vienna (2005).
- [20] UNITED NATIONS, Sources and Effects of Ionizing Radiation (Report to the General Assembly), Vol. I, Scientific Committee on the Effects of Atomic Radiation (UNSCEAR), UN, New York (2000).
- [21] UNITED NATIONS, Sources and Effects of Ionizing Radiation, Vol. II, Scientific Annexes C, D and E, Scientific Committee on the Effects of Atomic Radiation (UNSCEAR), UN, New York (2011).
- [22] INTERNATIONAL ATOMIC ENERGY AGENCY, WORLD HEALTH ORGANIZATION, UNITED NATIONS DEVELOPMENT PROGRAMME, FOOD AND AGRICULTURE ORGANIZATION OF THE UNITED NATIONS, UNITED NATIONS ENVIRONMENT PROGRAMME, UNITED NATIONS OFFICE FOR THE COORDINATION OF HUMANITARIAN AFFAIRS, UNITED NATIONS SCIENTIFIC COMMITTEE ON THE EFFECTS OF ATOMIC RADIATION, WORLD BANK, Chernobyl's Legacy: Health, Environmental and Socio-Economic Impacts and Recommendations to the Governments of Belarus, the Russian Federation and Ukraine, Second revised version, IAEA, Vienna (2006).

- [23] BUGLOVA, E., KENIGSBERG, J., McKENNA, T., "Reactor accidents and thyroid cancer risk: Use of the Chernobyl experience for emergency response", Proc. Int. Symp. on Radiation and Thyroid Cancer (THOMAS, A., KARAOGLOU, A., WILLIAMS, E.D., Eds), World Scientific Publishing, London (1999) 449–453.
- [24] INTERNATIONAL ATOMIC ENERGY AGENCY, Placing the Radiological Health Hazard in Perspective in an Emergency due to Severe Conditions at a Light Water Reactor, IAEA, Vienna (2013).
- [25] McKENNA, T., et al., Tools for placing the radiological health hazard in perspective following a severe emergency at a light water reactor (LWR) or its spent fuel pool, Health Phys. **108** 1 (2015) 15–31.
- [26] INTERNATIONAL ATOMIC ENERGY AGENCY, Communication with the Public in a Nuclear or Radiological Emergency, EPR-Public Communications 2012, IAEA, Vienna (2012).
- [27] INTERNATIONAL ATOMIC ENERGY AGENCY, Method for Developing a Communication Strategy and Plan for a Nuclear or Radiological Emergency, EPR-Public Communication Plan 2015, IAEA, Vienna (2015).
- [28] TRICHOPOULOS, D., et al., The victims of Chernobyl in Greece: Induced abortions after the accident, BMJ **295** (1987) 1100.
- [29] INTERNATIONAL ATOMIC ENERGY AGENCY, The Radiological Accident in Goiânia, IAEA, Vienna (1988).
- [30] INTERNATIONAL COMMISSION ON RADIOLOGICAL PROTECTION, Report of ICRP Task Group 84 on Initial Lessons Learned from the Nuclear Power Plant Accident in Japan vis-à-vis the ICRP System of Radiological Protection, ICRP, Ottawa (2012).
- [31] VANO, E., OHNO, K., COUSINS, C., NIWA, O., BOICE, J., Radiation risks and radiation protection training for healthcare professionals: ICRP and the Fukushima experience, J. Radiol. Prot. 31 3 (2011) 285–287.
- [32] GONZÁLEZ, A., The recommendations of the ICRP vis-à-vis the Fukushima Dai-ichi NPP accident aftermath, J. Radiol. Prot. **32** 1 (2012) 1–7.
- [33] KNUDSEN, L. B., Legally-induced abortions in Denmark after Chernobyl, Biomed. Pharmacother. 45 6 (1991) 229–231.
- [34] SPINELLI, A., OSBORN, J.F., The effects of the Chernobyl explosion on induced abortion in Italy, Biomed. Pharmacother. **45** 6 (1991) 243–247.
- [35] CHRISTENSEN, G., The Impact of the Chernobyl Accident on Norway, Institute for Energy Technology, Kjeller (1989).
- [36] KINUYA, S., A nuclear power plant accident in Fukushima: what should we do? Ann. Nucl. Med. 26 2 (2012) 113–114.
- [37] WORLD HEALTH ORGANIZATION, Use of Potassium Iodide for Thyroid Protection During Nuclear or Radiological Emergencies, Technical Briefings, WHO, Geneva (2011).

- [38] INTERNATIONAL COMMISSION ON RADIOLOGICAL PROTECTION, The 2007 Recommendations of the International Commission on Radiological Protection, ICRP Publication 103, Elsevier, Oxford (2007).
- [39] INTERNATIONAL COMMISSION ON RADIOLOGICAL PROTECTION, Doses to the Embryo and Fetus from Intakes of Radionuclides by the Mother (Corrected version, May 2002) ICRP Publication 88, Pergamon Press, Oxford (2000).
- [40] LIKHTAROV, I., et al., Estimation of the thyroid doses for Ukrainian children exposed in utero after the Chernobyl accident, Health Phys. **100** 6 (2011) 583–593.
- [41] NUCLEAR REGULATORY COMMISSION, Reactor Safety Study: An Assessment of the Accident Risks in U.S. Commercial Nuclear Power Plants, NUREG-75/014 (WASH-1400), NRC, Washington, DC (1975).
- [42] NUCLEAR REGULATORY COMMISSION, Accident Source Terms for Light-Water Nuclear Power Plants, Final Report, NUREG-1465, NRC, Washington, DC (1995).
- [43] SANDIA NATIONAL LABORATORIES, Accident Source Terms for Light-Water Nuclear Power Plants Using High-Burnup or MOX Fuel, SAND2011-0128, Albuquerque, NM, and Livermore, CA (2011).
- [44] NUCLEAR REGULATORY COMMISSION, State-of-the-Art Reactor Consequence Analyses (SOARCA), Vol. 1 (Rev. 1): Peach Bottom Integrated Analyses, and Vol. 2 (Rev. 1): Surry Integrated Analyses, NUREG/CR-7110, NRC, Washington, DC (2013).
- [45] NUCLEAR REGULATORY COMMISSION, Technical Study of Spent Fuel Pool Accident Risk at Decommissioning Nuclear Power Plants, NUREG-1738, NRC, Washington, DC (2001).
- [46] OECD NUCLEAR ENERGY AGENCY, The Chernobyl Reactor Accident Source Term, NEA/CSNI/R(95)24, OECD NEA, Paris (1995).
- [47] MINISTRY OF EDUCATION, CULTURE, SPORTS, SCIENCE AND TECHNOLOGY IN JAPAN, Analysis Results Concerning (i) Gamma-emitting Nuclides and (ii) Sr-89 and Sr-90 (Second Distribution Survey) by MEXT, MEXT, Tokyo (2012).
- [48] EMERGENCY OPERATION CENTER (MINISTRY OF EDUCATION, CULTURE, SPORTS, SCIENCE AND TECHNOLOGY), AGRICULTURE, FORESTRY AND FISHERIES RESEARCH COUNCIL (MINISTRY OF AGRICULTURE, FORESTRY AND FISHERIES), Summarized Version of the "Results of the Research on Distribution of Radioactive Substances Discharged by the Accident at TEPCO's Fukushima Dai-ichi NPP", NRA, Tokyo (2012).
- [49] KINOSHITA, N., et al., Assessment of individual radionuclide distributions from the Fukushima nuclear accident covering central-east Japan, Proc. Natl. Acad. Sci. USA 108 49 (2011) 19526–19529.
- [50] CHAISAN, K., SMITH, J.T., BOSSEW, P., KIRCHNER, G., LAPTEV, G.V., Worldwide isotope ratios of the Fukushima release and early-phase external dose reconstruction, Sci. Rep. **3** 2520 (2013).

- [51] KATATA, G., et al., Detailed source term estimation of the atmospheric release for the Fukushima Daiichi Nuclear Power Station accident by coupling simulations of atmospheric dispersion model with improved deposition scheme and oceanic dispersion model, Atmos. Chem. Phys. 15 (2015) 1029–1070.
- [52] INTERNATIONAL COMMISSION ON RADIOLOGICAL PROTECTION, Nuclear Decay Data for Dosimetric Calculations, ICRP Publication 107, Elsevier Science, Oxford and New York (2008).
- [53] INTERNATIONAL COMMISSION ON RADIOLOGICAL PROTECTION, Assessing Dose of the Representative Person for the Purpose of the Radiation Protection of the Public and the Optimisation of Radiological Protection: Broadening the Process, ICRP Publication 101, Elsevier, Oxford (2006).
- [54] EUROPEAN COMMISSION, Deposition of radionuclides, their subsequent relocation in the environment and resulting implications, Report EUR 16604 EN, European Commission, Luxembourg (1995).
- [55] ANSPAUGH, L. R., et al., Movement of radionuclides in terrestrial ecosystems by physical processes, Health Phys. 82 5 (2002) 669–679.
- [56] MAXWELL, R. M., ANSPAUGH, L. R., An improved model for prediction of resuspension, Health Phys. **101** 6 (2011) 722–730.
- [57] NATIONAL RADIOLOGICAL PROTECTION BOARD, Calculation of Resuspension Doses for Emergency Response, NRPB-W1, NRPB, Didcot (2002).
- [58] NATIONAL COUNCIL ON RADIATION PROTECTION AND MEASUREMENTS, Recommended Screening Limits for Contaminated Surface Soil and Review of Factors Relevant to Site Specific Studies, NCRP Report No. 129, NCRP, Bethesda, MD (1999).
- [59] INTERNATIONAL ATOMIC ENERGY AGENCY, Modelling of Resuspension, Seasonality and Losses during Food Processing, IAEA-TECDOC-647, IAEA, Vienna (1992).
- [60] DEPARTMENT OF HEALTH AND HUMAN SERVICES, CENTERS FOR DISEASE CONTROL AND PREVENTION, NATIONAL CANCER INSTITUTE, Feasibility Study of Weapons Test Fallout; Report on the Health Consequences to the American Population from Nuclear Weapons Tests Conducted by the United States and Other Nations, Appendix E, HHS-CDC-NCI, Washington, DC (2005).
- [61] ROED, J., Dry Deposition of Urban Surfaces, RISØ-R-515, Risø National Laboratory, Roskilde, Denmark (1985).
- [62] UNITED STATES ENVIRONMENTAL PROTECTION AGENCY, Evaluation of Skin and Ingestion Exposure Pathways, EPA 520/1 89-016, EPA, Washington, DC (1989).
- [63] UNITED STATES ENVIRONMENTAL PROTECTION AGENCY, Exposure Factors Handbook: 2011 Edition (Final), EPA/600/R-09/052F, EPA, Washington, DC (2011).

- [64] INTERNATIONAL ATOMIC ENERGY AGENCY, Generic Models for Use in Assessing the Impact of Discharges of Radioactive Substances to the Environment, Safety Report Series No. 19, IAEA, Vienna (2001).
- [65] INTERNATIONAL ATOMIC ENERGY AGENCY, Handbook of Parameter Values for the Prediction of Radionuclide Transfer in Terrestrial and Freshwater Environment, Technical Reports Series No. 472, IAEA, Vienna (2010).
- [66] NUCLEAR REGULATORY COMMISSION, Calculation of Annual Doses to Man from Routine Releases of Reactor Effluents for the Purpose of Evaluating Compliance with 10 CFR Part 50, Appendix I, NRC Regulatory Guide 1.109, Rev. 1, NRC, Washington, DC (1977).
- [67] HESSION, H., BYRNE, M., CLEARY, S., ANDERSSON, K. G., ROED, J., Measurement of contaminant removal from skin using a portable fluorescence scanning system, J. Environ. Radioact. 85 (2006) 196-204.
- [68] INTERNATIONAL COMMISSION ON RADIOLOGICAL PROTECTION, Age-dependent Doses to Members of the Public from Intake of Radionuclides - Part 1, ICRP Publication 56, Pergamon Press, Oxford (1990).
- [69] UNITED STATES ENVIRONMENTAL PROTECTION AGENCY, Federal Guidance Report No. 12: External Exposure to Radionuclides in Air, Water, and Soil, EPA-402-R-93-081, EPA, Washington, DC (1993).
- [70] NUCLEAR REGULATORY COMMISSION, COMISSION OF EUROPEAN COMMUNITIES, Probabilistic Accident Consequence Uncertainty Analysis, Uncertainty Assessment for Deposited Material and External Doses, Vol. 2: Appendices, NUREG/CR-6526 Vol. 2, NRC, Washington, DC (1997).
- [71] UNITED STATES DEPARTMENT OF ENERGY, DOE Handbook: Airborne Release Fractions/Rates and Respirable Fractions for Nonreactor Nuclear Facilities, 2 vols., DOE-HDBK-3010-94, DOE, Washington, DC (1994; 2013).
- [72] INTERNATIONAL COMMISSION ON RADIOLOGICAL PROTECTION, Report on the Task Group on Reference Man, ICRP Publication 23, ICRP, Oxford (1975).
- [73] INTERNATIONAL COMMISSION ON RADIOLOGICAL PROTECTION, ICRP Database of Dose Coefficients: Workers and Members of the Public, Version 3.0., Pergamon Press, Oxford (2012).
- [74] NATIONAL CANCER INSTITUTE, Estimated Exposures and Thyroid Doses Received by the American People from Iodine-131 in Fallout Following Nevada Atmospheric Nuclear Bomb Tests: A Report from the National Cancer Institute, Department of Health and Human Services, Washington, DC (1997).
- [75] INTERNATIONAL ATOMIC ENERGY AGENCY, Dangerous Quantities of Radioactive Material (D-values), EPR-D-VALUES 2006, IAEA, Vienna (2006).
- [76] KUTKOV, V., BUGLOVA, E., McKENNA, T., Severe deterministic effects of external exposure and intake of radioactive material: basis for emergency response criteria, J. Radiol. Prot. **31** 2 (2011) 237–253.

- [77] INTERNATIONAL ATOMIC ENERGY AGENCY, WORLD HEALTH ORGANIZATION, Generic Procedures for Medical Response during a Nuclear or Radiological Emergency, EPR-MEDICAL 2005, IAEA, Vienna (2005).
- [78] INTERNATIONAL COMMISSION ON RADIOLOGICAL PROTECTION, Conversion Coefficients for Use in Radiological Protection against External Radiation, ICRP Publication 74, Pergamon Press, Oxford (1996).
- [79] ISAKSSON, M., "Environmental Dosimetry—Measurements and Calculations", Radioisotopes—Applications in Physical Sciences (SINGH, N., Ed.), InTech, Rijeka (2011), 175–196.
- [80] AGARWAL, P., SAHU, S., Determination of hand and palm area as a ratio of body surface area in Indian population, Indian J. Plast. Surg. **43** 1 (2010) 49–53.
- [81] NORDIC NUCLEAR SAFETY RESEARCH, Assessment of Accidental Uptake of Iodine-131 in Emergency Situations, NKS-298, NKS, Roskilde, Denmark (2014).
- [82] SUZUKI, T., "The internal and external dosimetry challenges from past experience: Fukushima Daiichi accident", paper presented at IAEA Int. Conf. on Occupational Radiation Protection: Enhancing the Protection of Workers—Gaps, Challenges and Developments, CN-223, Vienna (2014).
- [83] BUNDESMINISTERIUM FÜR UMWELT, NATURSCHUTZ UND REAKTOR-SICHERHEIT, Thyroid Exposure in Belorussian and Ukrainian Children after the Chernobyl Accident and Resulting Risk of Thyroid Cancer, Appendix 2, BMU-2005-668, BMU, Bonn (2005).
- [84] ULANOVSKY, A., DROZDOVITCH, V., BOUVILLE, A., Influence of radionuclides distributed in the whole body on the thyroid dose estimates obtained from direct thyroid measurements made in Belarus after the Chernobyl accident, Radiat. Prot. Dosim. 112 3 (2004) 405–418.
- [85] NUCLEAR REGULATORY COMMISSION, Minimum Detectable Concentration / Activities for Typical Radiation Survey Instruments for Various Contaminants and Field Conditions, NUREG-1507, NRC, Washington, DC (1998).
- [86] INTERNATIONAL COMMISSION ON RADIOLOGICAL PROTECTION, Pregnancy and Medical Radiation, ICRP Publication 84, Pergamon Press, Oxford (2000).
- [87] EUROPEAN ATOMIC ENERGY COMMUNITY, FOOD AND AGRICULTURE ORGANIZATION OF THE UNITED NATIONS, INTERNATIONAL ATOMIC ENERGY AGENCY, INTERNATIONAL LABOUR ORGANIZATION, INTERNATIONAL MARITIME ORGANIZATION, OECD NUCLEAR ENERGY AGENCY, PAN AMERICAN HEALTH ORGANIZATION, UNITED NATIONS ENVIRONMENT PROGRAMME, WORLD HEALTH ORGANIZATION, Fundamental Safety Principles, IAEA Safety Standards Series No. SF-1, IAEA, Vienna (2006).
- [88] INTERNATIONAL ATOMIC ENERGY AGENCY, IAEA Safety Glossary: Terminology Used in Nuclear Safety and Radiation Protection, 2007 Edition, IAEA, Vienna (2007).
# NOTATION

Notation	Unit	Definition	Value
ad <sub>skin,skin-scenario,i</sub>	[Gy/ (Bq/m <sup>2</sup> )]	RBE weighted absorbed dose to the skin derma of the representative person, for the 'skin' scenario, per unit skin surface activity of radionuclide i	Table 28
ad <sub>skin,skin-srf,i</sub>	[(Gy/s)/ (Bq/m <sup>2</sup> )]	RBE weighted absorbed dose rate to the skin derma of the representative person, from activity on the skin, per unit skin surface activity of radionuclide i	Table 27
$a_{eff,\beta}$ (baseline-inst)	$[cm^2]$	Effective window area (e.g. sensitive or radiation inlet area) of the baseline beta monitoring instrument	15
a <sub>eff,β</sub> (inst)	[cm <sup>2</sup> ]	Effective window area (e.g. sensitive or radiation inlet area) of the beta monitoring instrument (inst)	External input
A <sub>i</sub> (t,mix)	[Bq]	Activity of radionuclide i at time t after reactor shutdown, for a specific radionuclide mix	Eq. (2)
A <sub>OIL1,E</sub> (t,mix)	[Bq/m <sup>2</sup> ]	Gross activity on the ground at time t after reactor shutdown, that for the 'ground' scenario results in a total effective dose to the representative person meeting the GC(Urgent,E,7d), for a specific radionuclide mix	Eq. (52)
$A_{OIL1,Hietus}(t,mix)$	[Bq/m <sup>2</sup> ]	Gross activity on the ground at time t after reactor shutdown, which, for the 'ground' scenario, results in a total equivalent dose to the fetus meeting the $GC(Urgent, H_{fetus}, 7d)$ , for a specific radionuclide mix	Eq. (53)
A <sub>OIL2,E</sub> (t,mix)	[Bq/m <sup>2</sup> ]	Gross activity on the ground at time t after reactor shutdown, which, for the 'ground' scenario, results in a total effective dose to the representative person meeting the GC(Early,E,1a), for a specific radionuclide mix	Eq. (56)
A <sub>OIL2,Hietus</sub> (t,mix)	[Bq/m <sup>2</sup> ]	Gross activity on the ground at time t after reactor shutdown, which, for the 'ground' scenario, results in a total equivalent dose to the fetus meeting the $GC(Early, H_{fetus}, 9mo)$ , for a specific radionuclide mix	Eq. (57)
A <sub>OIL3,eing</sub> (t,mix)	[Bq/m <sup>2</sup> ]	Gross activity on the ground at time t after reactor shutdown, which, for the 'food pre-analysis' scenario, results in a committed effective dose from ingestion to the representative person meeting the GC(Ingestion, $e_{ing}$ ,1a), for a specific radionuclide mix	Eq. (60)
$A_{OIL3,hfetus,ing}(t,mix)$	[Bq/m <sup>2</sup> ]	Gross activity on the ground at time t after reactor shutdown, which, for the 'food pre-analysis' scenario, results in a committed equivalent dose to the fetus from ingestion meeting the GC(Ingestion, $h_{fetus,ing}$ ,9mo), for a specific radionuclide mix	Eq. (61)
A <sub>OIL4,ADskin</sub> (t,mix)	[Bq/m <sup>2</sup> ]	Gross activity on the skin surface at time t after reactor shutdown, which, for the 'skin' scenario, results in an RBE weighted absorbed dose to the skin meeting the GC(Acute,AD <sub>skin-ext</sub> ,10h), for a specific radionuclide mix	Eq. (66)
A <sub>OIL4,e</sub> (t,mix)	[Bq/m <sup>2</sup> ]	Gross activity on the skin surface at time t after reactor shutdown, which, for the 'skin' scenario, results in an effective dose to the representative person from inadvertent ingestion meeting the GC(Urgent,E,7d), for a specific radionuclide mix	Eq. (64)

Notation	Unit	Definition	Value
A <sub>OIL4,hfetus</sub> (t,mix)	[Bq/m <sup>2</sup> ]	Gross activity on the skin surface at time t after reactor shutdown, which, for the 'skin' scenario, results in an equivalent dose to the fetus from inadvertent ingestion meeting the GC(Urgent, $H_{fetus}$ ,7d), for a specific radionuclide mix	Eq. (65)
A <sub>OIL7,eing</sub> (t,mix)	[Bq/kg]	Activity concentration in food, milk or drinking water, which, for the 'food post-analysis' scenario, results in a committed effective dose to the representative person from ingestion meeting the GC(Ingestion,e <sub>ing</sub> ,1a), for a specific radionuclide mix	Eq. (70)
$A_{OIL7,hfetus,ing}(t,mix)$	[Bq/kg]	Activity concentration in food, milk or drinking water, which, for the 'food post-analysis' scenario, results in a committed equivalent dose to the fetus from ingestion meeting the GC(Ingestion, $h_{fetus,ing}$ , 1a), for a specific radionuclide mix	Eq. (71)
A <sub>OIL8,h</sub> thyroid,I-131-burden	[Bq]	Activity of I-131 in the thyroid (thyroid burden), which, for the 'thyroid' scenario, results in a committed equivalent dose to an infant meeting the $GC(Urgent,h_{thyroid,thy-burden})$	Eq. (75)
$CalF_{\beta,i}$ (baseline-inst)	[(Bq/cm <sup>2</sup> ) /cps]	Calibration factor of the baseline beta monitoring instrument for radionuclide i	Table 31
$CalF_{\beta,i}(inst)$	[(Bq/cm <sup>2</sup> ) /cps]	Calibration factor of the beta monitoring instrument (inst) for one of the radionuclides listed in Table 31	Eq. (48)
$CalF_{\beta,i}(suitable-inst$	t) $[(Bq/cm^2)/cps]$	Calibration factor of a suitable beta monitoring instrument for radionuclide i	Table 31
CalF <sub>R,I-131</sub> (baseline- inst)	- [Bq/ (Sv/s)]	Calibration factor of the baseline monitoring instrument for the activity of I-131 in the thyroid relative to the ambient dose rate measured in front of the thyroid in contact with the skin of an infant	3.5E+13
$C_{\beta,i}$ (baseline-inst)	[cps/ (Bq/cm <sup>2</sup> )]	Baseline beta monitoring instrument coefficient for radionuclide i	Table 31
$C_{\beta,i}(inst)$	[cps/ (Bq/cm <sup>2</sup> )]	Beta monitoring instrument coefficient for radionuclide i	Eq. (49)
$C_{\beta,i}(suitable-inst)$	[cps/ (Bq/cm <sup>2</sup> )]	Instrument coefficient of a suitable beta monitoring instrument for radionuclide i	Table 31
$ConF_{\varphi \to H^*(10), E^j}$	[pSv cm <sup>2</sup> ]	Conversion factor for the ambient dose equivalent $H^*(10)$ [pSv] per fluence [cm <sup>-2</sup> ] of photons with energy $E_j$	Table A.21 of Ref. [78]
$CorF_{distance,\beta}$	[unitless]	Distance correction factor for the beta monitoring instrument that accounts for the reduction due to absorption in air at the recommended distance of 2 cm for skin monitoring	0.5
$CorF_{field,\beta}$	[unitless]	Field beta correction factor for the beta monitoring instrument that accounts for the reduction in the beta count rate for measurements made under field (i.e. non-ideal) conditions	0.25
CorF <sub>grd</sub>	[unitless]	Ground roughness correction factor to account for the dose rate reduction due to ground roughness	0.7
$CorF_{I-131-thyroid-burden}(t_m)$	[unitless]	Correction factor to account for the reduction in the activity in the thyroid due to the radioactive decay and biological removal of I-131	Fig. 8

Notation	Unit	Definition	Value
$CorF_{surface,\beta}$	[unitless]	Surface correction factor for the beta monitoring instrument that accounts for the absorption due to non-ideal surface conditions encountered during an emergency	0.5
DA <sub>OIL1</sub> (t,mix)	[Bq/m <sup>2</sup> ]	Derived gross activity on the ground at time t after reactor shutdown, which, for the 'ground' scenario, results in any of the generic criteria specified for $OIL1_{\gamma}$ (in Table 2) being met, for a specific radionuclide mix	Eq. (51)
DA <sub>OIL2</sub> (t,mix)	[Bq/m <sup>2</sup> ]	Derived gross activity on the ground at time t after reactor shutdown, which, for the 'ground' scenario, results in any of the generic criteria specified for $OIL2_{\gamma}$ (in Table 2) being met, for a specific radionuclide mix	Eq. (55)
DA <sub>OIL3</sub> (t,mix)	[Bq/m <sup>2</sup> ]	Derived gross activity on the ground at time t after reactor shutdown, which, for the 'food pre-analysis' scenario, results in any of the generic criteria specified for $OIL3_{\gamma}$ (in Table 2) being met, for a specific radionuclide mix	Eq. (59)
DA <sub>OIL4</sub> (t,mix)	[Bq/m <sup>2</sup> ]	Derived gross activity on the skin surface at time t after reactor shutdown, which, for the 'skin' scenario, results in any of the generic criteria specified for OIL4 <sub><math>\gamma</math></sub> and OIL4 <sub><math>\beta</math></sub> (in Table 2) being met, for a specific radionuclide mix	Eq. (63)
DA <sub>OIL7</sub> (t,mix)	[Bq/kg]	Activity concentration in food, milk or drinking water at time t after reactor shutdown, which. for the 'food post- analysis' scenario, results in any of the generic criteria specified for OIL7 (in Table 2) being met, for a specific radionuclide mix	Eq. (69)
Δ	[s]	Exposure period	Variable
$\Delta_{\rm available-ing}$	[s]	Period of time that the affected food or forage is available for consumption	365 d = 3.15E+07 s
$\delta_{\beta}(inst)$	[unitless]	Ratio factor of the beta monitoring instrument (inst)	Eq. (76)
$\Delta_{dep \rightarrow consumption}$	[s]	Period of time between deposition and consumption	1 d = 86 400 s
$\Delta_{\rm eff,OIL3,i}$	[s]	Effective availability period for radionuclide i for ingestion in the 'food pre-analysis' scenario	Table 20
$\Delta_{\rm eff,OIL4-Acute,i}$	[s]	Effective availability period for radionuclide i for inadvertent ingestion from the skin in the 'skin' scenario for the acute generic criteria	Table 20
$\Delta_{\rm eff,OIL4-Urgent,i}$	[s]	Effective availability period for radionuclide i for inadvertent ingestion from the skin in the 'skin' scenario for the urgent generic criteria	Table 20
$\Delta_{\rm eff,OIL7,i}$	[s]	Effective availability period for radionuclide i for ingestion in the 'food post-analysis' scenario	Table 20
d <sub>inadv-ing</sub>	[m]	Depth available for inadvertent ingestion	1 mm = 0.001 m
d <sub>monitoring</sub>	[m]	Monitoring distance at which the measurements of the skin are recommend to be taken for the default $OIL4_{\gamma}$ value (as indicated in Section 2.6.2)	0.1
$\Delta_{ m OIL4-Acute}$	[s]	Exposure period considered for the acute generic criteria (given in Table 2)	10 h = 36 000 s

Notation	Unit	Definition	Value
$\Delta_{ m OIL4-Urgent}$	[s]	Exposure period considered for the urgent generic criteria (given in Table 2)	7 d = 604 800 s
e <sub>air-sh,i</sub> (adult)	[(Sv/s)/ (Bq/m <sup>3</sup> )]	External effective dose rate to the adult from air submersion, per unit volume activity of radionuclide i	Table 24
$e_{air-sh(grd-scenario),i}(\Delta)$	[Sv/ (Bq/m <sup>2</sup> )]	External effective dose to the representative person, over an exposure period $\Delta$ , from air shine resulting from resuspended ground deposition, for the 'ground' scenario, per unit ground surface activity of radionuclide i	Eq. (24)
e <sub>ing,food</sub> -pre-analysis- scenario,i	[Sv/ (Bq/m <sup>2</sup> )]	Committed effective dose to the representative person from ingestion over 1 year, for the 'food pre-analysis' scenario, per unit ground surface activity of radionuclide i	Table 26
e <sub>ing,skin-scenario,i</sub>	[Sv/ (Bq/m <sup>2</sup> )]	Committed effective dose to the representative person from inadvertent ingestion, for the 'skin' scenario, per unit skin surface activity of radionuclide i	Table 28
$E_{grd-scenario,i}(\Delta)$	[Sv/ (Bq/m <sup>2</sup> )]	Total effective dose to the representative person over an exposure period $\Delta$ , for the 'ground' scenario, per unit ground surface activity of radionuclide i	Table 23
$e_{grd-sh(grd-scenario),i}(\Delta)$	[Sv/ (Bq/m <sup>2</sup> )]	External effective dose to the representative person, over an exposure period $\Delta$ , from ground shine resulting from ground deposition, for the 'ground' scenario, per unit ground surface activity of radionuclide i	Eq. (23)
$e_{inadv-ing(grd-scenario),i}(\Delta$	)[Sv/ (Bq/m <sup>2</sup> )]	Committed effective dose to the representative person, over an exposure period $\Delta$ , from inadvertent ingestion of ground deposition, for the 'ground' scenario, per unit ground surface activity of radionuclide i	Eq. (26)
e <sup>ing, food-post-analysis-</sup> scenario, i	[Sv/ (Bq/kg)]	Committed effective dose to the representative person from ingestion of food, milk and drinking water over 1 year, for the 'food post-analysis' scenario, per unit activity concentration of radionuclide i in food, milk or drinking water	Table 29
e <sub>ing,i</sub> (adult)	[Sv/Bq]	Committed effective dose to the adult from ingestion, per unit intake of radionuclide i	Table 27
e <sub>ing,i</sub> (infant)	[Sv/Bq]	Committed effective dose to the infant from ingestion, per unit intake of radionuclide i	Table 24 and Table 27
e <sub>inh,i</sub> (adult)	[Sv/Bq]	Committed effective dose to the adult from inhalation, per unit intake of radionuclide i	Table 24
$e_{inh-resusp(grd-scenario),i}(\Delta)$	[Sv/ (Bq/m <sup>2</sup> )]	Committed effective dose to the representative person, over an exposure period $\Delta$ , from inhalation of resuspended ground deposition, for the 'ground' scenario, per unit ground surface activity of radionuclide i	Eq. (25)
e <sub>plane-srf,i</sub> (adult)	[(Sv/s)/ (Bq/m <sup>2</sup> )]	Effective dose rate to the adult from an infinite smooth plane source, per unit surface activity of radionuclide i	Table 24
$\Phi_1$	[m <sup>2</sup> /kg]	Mass interception factor for pasture grass in dry weight, i.e. ratio between the activity concentration on the plant $[Bq/kg]$ and the unit ground surface activity on the terrestrial surface $[Bq/m^2]$ (soil plus vegetation)	3

Notation	Unit	Definition	Value
$\Phi_2$	[m <sup>2</sup> /kg]	Mass interception factor for leafy vegetables (fresh or wet weight), i.e. ratio between the activity concentration on the plant $[Bq/kg]$ and the unit ground surface activity on the terrestrial surface $[Bq/m^2]$ (soil plus vegetation)	0.3
F <sub>consumption,i</sub>	[unitless]	Fraction of the radionuclide i remaining at the time of human consumption	Table 22
F <sub>cow-feed</sub>	[unitless]	Fraction of the cow feed assumed to be affected	0.7
F <sub>diet,OIL7</sub>	[unitless]	Fraction of the total diet (i.e. all food, milk and drinking water consumed) assumed to be affected for the 'food post-analysis' scenario	0.5
F <sub>I-131-retained-in-thyroid</sub>	[unitless]	Fraction of I-131 that is retained in the thyroid following intake	0.3
$F_{ideal-surf-emission}$	[unitless]	Ideal surface emission factor, i.e. ratio between (a) the number of particles that emerge from the front face of a surface and (b) the number of particles emitted from a radionuclide on the surface	0.5
F <sub>lv</sub>	[unitless]	Fraction of the leafy vegetables consumed assumed to be affected before actions are taken to control intake	0.5
F <sub>milk</sub>	[unitless]	Fraction of the milk consumed assumed to be affected before actions are taken to control intake	0.5
F <sub>of</sub>	[unitless]	Fraction of the exposure period that the person is assumed to spend inside a building (occupancy fraction)	0.6
F <sub>prep</sub>	[unitless]	Fraction of the radioactive material remaining for ingestion after preparations	1
F <sub>rf</sub>	[unitless]	Fraction of the airborne radioactive material that is deposited in the pulmonary region of the lung (respirable fraction)	1
F <sub>sf</sub>	[unitless]	Fraction of the dose rate received from ground deposition for a person located inside a building relative to the dose rate for that person located outside the building (shielding factor)	0.4
$F_{soil,inadv-ing}(\tau_0)$	[unitless]	Fraction of the deposited material available for ingestion (e.g. material that is within the $d_{inadv-ing}$ or not adhered to a surface) at time $\tau_0$	1
GC(Acute, AD <sub>skin-ext</sub> ,10h)	[Gy]	Generic criterion used to take response actions under any circumstance based on the RBE weighted absorbed dose to $100 \text{ cm}^2$ of the skin dermis of the representative person from acute external exposure in the first 10 hours	10 Gy to 100 $\text{cm}^2$
GC(Early,E,1a)	[Sv]	Generic criterion used to take early response actions based on the total effective dose to the representative person in the first year	0.1
GC (Early,H <sub>fetus</sub> ,9mo)	[Sv]	Generic criterion used to take early response actions based on the total equivalent dose to the fetus in the full period of in utero development	0.1
GC (Ingestion,e <sub>ing</sub> ,1a)	[Sv]	Generic criterion used to restrict the ingestion of non- essential food, milk and drinking water based on the committed effective dose from ingestion to the representative person over the first year	0.01

Notation	Unit	Definition	Value
GC(Ingestion, h <sub>fetus,ing</sub> ,9mo)	[Sv]	Generic criterion used to restrict the ingestion of non- essential food, milk and drinking water based on the committed equivalent dose to the fetus from ingestion over the full period of in utero development	0.01
GC(Urgent,E,7d)	[Sv]	Generic criterion used to take urgent response actions based on the total effective dose to the representative person in the first 7 days	0.1
GC (Urgent,H <sub>fetus</sub> ,7d)	[Sv]	Generic criterion used to take urgent response actions based on the total equivalent dose to the fetus in the first 7 days	0.1
GC(Urgent, h <sub>thyroid,thy-burden</sub> )	[Sv]	Generic criterion used to take urgent response actions based on the committed equivalent dose to the thyroid from radioiodine in the thyroid (thyroid burden)	0.1
GF <sub>srf</sub>	[unitless]	Geometry factor to account for the measurement being taken at 10 cm from the skin with an instrument with a view of $100 \text{ cm}^2$	Eq. (42)
H* <sub>grd-sh,i</sub>	[(Sv/s)/ (Bq/m <sup>2</sup> )]	Ambient dose equivalent rate at 1 m above ground level, per unit ground surface activity of radionuclide i	Table 30
H* <sub>1-131-thyroid-burden</sub> (baseline-inst)	[(Sv/s) /Bq]	Ambient dose equivalent rate in front of the thyroid as measured with the baseline instrument for monitoring the thyroid (baseline-inst) against the neck of an infant, per unit activity of I-131 (burden) in the thyroid	2.9E-14
H* <sub>skin-100cm</sub> <sup>2</sup> -srf,i	[(Sv/s)/ (Bq/m <sup>2</sup> )]	Ambient dose equivalent rate at 10 cm from activity on $100 \text{ cm}^2$ of the skin surface, per unit surface activity of radionuclide i	Table 30
$\mathbf{h}_{ ext{fetus,air-sh(grd-scenario),i}}(\Delta)$	[Sv/ (Bq/m <sup>2</sup> )]	External equivalent dose to the fetus, over an exposure period $\Delta$ , from air shine resulting from resuspended ground deposition, for the 'ground' scenario, per unit ground surface activity of radionuclide i	Eq. (29)
h <sub>fetus,ing,food-pre-analysis-</sub> scenario,i	[Sv/ (Bq/m <sup>2</sup> )]	Committed equivalent dose to the fetus from ingestion of food, milk and drinking water by the pregnant woman, over the period of in utero development, for the 'food pre- analysis' scenario, per unit ground surface activity of radionuclide i	Table 26
$\mathrm{H}_{\mathrm{fetus,grd-scenario,i}}(\Delta)$	[Sv/ (Bq/m <sup>2</sup> )]	Total equivalent dose to the fetus over an exposure period $\Delta$ , for the 'ground' scenario, per unit ground surface activity of radionuclide i	Table 23
$\mathbf{h}_{ ext{fetus,grd-sh(grd-scenario),i}}(\Delta)$	[Sv/ (Bq/m <sup>2</sup> )]	External equivalent dose to the fetus, over an exposure period $\Delta$ , from ground shine resulting from ground deposition, for the 'ground' scenario, per unit ground surface activity of radionuclide i	Eq. (28)
$\mathbf{h}_{ ext{fetus,inadv-ing(grd-scenario),i}}(\Delta)$	[Sv/ (Bq/m <sup>2</sup> )]	Committed equivalent dose to the fetus, over an exposure period $\Delta$ , from inadvertent ingestion of ground deposition by an adult (i.e. pregnant woman), for the 'ground' scenario, per unit ground surface activity of radionuclide i	Eq. (31)
h <sub>fetus,ing,food-post-analysis-</sub> scenario,i	[Sv/ (Bq/kg)]	Committed equivalent dose to the fetus from ingestion of food, milk and drinking water by the pregnant woman, over 1 year, for the 'food post-analysis' scenario, per unit activity concentration of radionuclide i in food, milk or drinking water	Table 29

Notation	Unit	Definition	Value
$\mathbf{h}_{\text{fetus,ing,i}}$	[Sv/Bq]	Committed equivalent dose to the fetus from ingestion by an adult (i.e. pregnant woman), per unit intake of radionuclide i	Table 25 and Table 27
h <sub>fetus,ing,skin-scenario,i</sub>	[Sv/ (Bq/m <sup>2</sup> )]	Committed equivalent dose to the fetus from inadvertent ingestion of radioactive material on the skin, for the 'skin' scenario, per unit skin surface activity of radionuclide i	Table 28
h <sub>fetus,inh,i</sub>	[Sv/Bq]	Committed equivalent dose to the fetus from inhalation by an adult (i.e. pregnant woman), per unit intake of radionuclide i	Table 25
$\mathbf{h}_{ ext{fetus,inh-resusp(grd-scenario),i}}(\Delta)$	[Sv/ (Bq/m <sup>2</sup> )]	Committed equivalent dose to the fetus, over an exposure period $\Delta$ , from inhalation of resuspended ground deposition by a pregnant woman, for the 'ground' scenario, per unit ground surface activity of radionuclide i	Eq. (30)
h <sub>red marrow,air-sh,i</sub> (adult)	[(Sv/s)/ (Bq/m <sup>3</sup> )]	Equivalent dose rate to the red marrow of the adult from submersion in a semi-infinite cloud of airborne radioactive material, per unit air activity concentration of radionuclide i	Table 25
h <sub>red marrow,</sub> <sub>plane-srf,i</sub> (adult)	[(Sv/s)/ (Bq/m <sup>2</sup> )]	Equivalent dose rate to the red marrow of the adult from an infinite smooth plane source, for radionuclide i	Table 25
h <sub>thyroid,I-131-thyroid-burden</sub>	[Sv/Bq]	Committed equivalent dose to the thyroid of an infant from I-131 in the thyroid (thyroid burden)	1.2E-05
h <sub>thyroid,ing-I-131</sub>	[Sv/Bq]	Committed equivalent dose to the thyroid from ingestion of I-131 by the infant, per unit intake of I-131	3.6E-06
$I_{\text{fuel-type},i}(t)$	[Bq]	Inventory of radionuclide i in the fuel at time t after reactor shutdown	Eq. (3)
$I_{\text{fuel-type,i}}(t_0)$	[Bq]	Inventory of radionuclide i in the fuel at time $t_0$	Table 4
λ <sub>bio-I-131</sub>	[s <sup>-1</sup> ]	Biological removal constant of I-131 in the thyroid	1.0E-07
λ <sub>i</sub>	[s <sup>-1</sup> ]	Decay constant for radionuclide i	Table 6
L <sub>hand</sub>	[kg/m <sup>2</sup> ]	Amount of soil loaded on the hand of the maximally exposed individual (e.g. children playing outdoors)	$1.8 \text{ mg/cm}^2 = 0.018 \text{ kg/m}^2$
$\lambda_{w-skin}$	[s <sup>-1</sup> ]	Removal constant from skin due to weathering	Eq. (17)
$\lambda_{w-vegetation}$	[s <sup>-1</sup> ]	Removal constant from vegetation due to weathering	Eq. (13)
OIL1 <sub>γ</sub>	[µSv/h]	Default $OIL1_{\gamma}$ value, i.e. the ambient dose equivalent rate value measured at 1 m above ground level at which to implement the response actions specified in the OIL chart in Section 2.6.1	1000
OIL1 <sub>γ</sub> (t,mix)	[µSv/h]	Ambient dose equivalent rate measured at 1 m above ground level at time t after reactor shutdown, which, for the 'ground' scenario, results in any of the generic criteria specified for $OIL1_{\gamma}$ (in Table 2) being met, for a specific radionuclide mix	Fig. 9
OIL2 <sub>γ</sub>	[µSv/h]	Default OIL2 $_{\gamma}$ value, i.e. the ambient dose equivalent rate value measured at 1 m above ground level at which to implement the response actions specified in the OIL chart in Section 2.6.1	$ \frac{100 (t \le 10 d)}{25 (t > 10 d)} $

Notation	Unit	Definition	Value
OIL2 <sub>γ</sub> (t,mix)	[µSv/h]	Ambient dose equivalent rate measured at 1 m above ground level at time t after reactor shutdown, which, for the 'ground' scenario, results in any of the generic criteria specified for $OIL2_{\gamma}$ (in Table 2) being met, for a specific radionuclide mix	Fig. 10
OIL3 <sub>γ</sub>	[µSv/h]	Default $OIL3_{\gamma}$ value, i.e. the ambient dose equivalent rate value measured at 1 m above ground level at which to implement the response actions specified in the OIL chart in Section 2.6.1	1
OIL3 <sub>γ</sub> (t,mix)	[µSv/h]	Ambient dose equivalent rate measured at 1 m above ground level at time t after reactor shutdown, which, for the 'food pre-analysis' scenario, results in any of the generic criteria specified for OIL3 <sub><math>\gamma</math></sub> (in Table 2) being met, for a specific radionuclide mix	Fig. 11
OIL4 <sub>β</sub>	[cps]	Default $OIL4_{\beta}$ value, i.e. the beta count rate value measured with the baseline beta monitoring instrument at 2 cm from bare skin, at which to implement the response actions specified in the OIL chart in Section 2.6.2	1000
OIL4 <sub>β</sub> (inst)	[cps]	Default $OIL4_{\beta}$ value for a specific instrument, i.e. the beta count rate value measured with the instrument (inst) at 2 cm from bare skin, at which to implement the response actions specified in the OIL chart in Section 2.6.2	Eq. (77)
OIL4 <sub>β</sub> (t,mix)	[cps]	Beta count rate measured with the baseline beta monitoring instrument at 2 cm from bare skin at time t after reactor shutdown, which, for the 'skin' scenario, results in any of the generic criteria specified for $OIL4_{\beta}$ (in Table 2) being met, for a specific radionuclide mix	Fig. 13
OIL4 <sub>γ</sub>	[µSv/h]	Default OIL4 $_{\gamma}$ value, i.e. the ambient dose equivalent rate value measured at 10 cm from the bare skin, at which to implement the response actions specified in the OIL chart in Section 2.6.2	1
OIL4 <sub>γ</sub> (t,mix)	[µSv/h]	Ambient dose equivalent rate measured at 10 cm from bare skin at time t after reactor shutdown, which, for the 'skin' scenario results in any of the generic criteria specified for OIL4 $_{\gamma}$ (in Table 2) being met, for a specific radionuclide mix	Fig. 12
OIL5	[Bq/kg]	Gross beta or alpha concentrations in food, milk or drinking water at which to implement the response actions specified in para. II.21 of Ref. [2] (i.e. stop consumption)	Table 9 of Ref. [2]
OIL6	[Bq/kg]	Activity concentration of a specific radionuclide in food, milk or drinking water at which to implement the response actions specified in para. II.21 of Ref. [2] (i.e. stop consumption)	Para. II.24 of Ref. [2]
OIL7	[Bq/kg]	Combination of the default $OIL7_{I-131}$ and $OIL7_{Cs-137}$ values	1000 for I-131 200 for Cs-137
OIL7 <sub>Cs-137</sub>	[Bq/kg]	Default OIL7 value for the activity concentration of the marker radionuclide Cs-137 in food, milk or drinking water, i.e. the activity concentration value of Cs-137 at which to implement the response actions specified in the OIL chart in Section 2.6.3 (needs to be checked in combination with OIL7 <sub>I-131</sub> )	200

Notation	Unit	Definition	Value
OIL7 <sub>Cs-137</sub> (t,mix)	[Bq/kg]	Activity concentration of the marker radionuclide Cs-137 in food, milk or drinking water at time t after reactor shutdown, which, for the 'food post-analysis' scenario, results in any of the generic criteria specified for OIL7 (in Table 2) being met, for a specific radionuclide mix (needs to be used together with $OIL7_{I-131}$ )	Fig. 15
OIL7 <sub>I-131</sub>	[Bq/kg]	Default OIL7 value for the activity concentration of the marker radionuclide I-131 in food, milk or drinking water, i.e. the activity concentration value of I-131 at which to implement the response actions specified in the OIL chart in Section 2.6.3 (needs to be checked in combination of $OIL7_{Cs-137}$ )	1000
OIL7 <sub>I-131</sub> (t,mix)	[Bq/kg]	Activity concentration of the marker radionuclide I-131 in food, milk or drinking water at time t after reactor shutdown, which, for the 'food post-analysis' scenario, results in any of the generic criteria specified for OIL7 (in Table 2) being met, for a specific radionuclide mix (needs to be used together with $OIL7_{Cs-137}$ )	Fig. 14
OIL8 <sub>7</sub>	[µSv/h]	Default OIL8 $_{\gamma}$ value, i.e. the ambient dose equivalent rate value measured in front of the thyroid in contact with the skin at which to implement the response actions specified in the OIL chart in Section 2.6.4	0.5
OIL8 <sub>γ</sub> (t <sub>m</sub> , baseline-inst)	[µSv/h]	Ambient dose equivalent rate measured with the baseline monitoring instrument (baseline-inst) in front of the thyroid in contact with the skin of an infant, at time $t_m$ after intake, for which the activity concentration of I-131 in the thyroid results in the generic criterion specified for OIL8 <sub><math>\gamma</math></sub> (in Table 2) being met	Fig. 17
Q <sub>air</sub>	[m <sup>3</sup> /s]	Breathing rate of the adult performing light activity	$1.2 \text{ m}^3/\text{h} =$ 3.3E-04 m <sup>3</sup> /s
Q <sub>diet</sub> (g)	[kg/s]	Consumption rate of food, milk and drinking water (total diet) for an individual in age group g	415 kg/a= 1.6E-05 kg/s for infants 1040 kg/a = 3.2E-05 kg/s for adults
Q <sub>lv</sub> (g)	[kg/s]	Consumption rate of leafy vegetables in fresh weight by an individual in age group g	20 kg/a = 6.3E-07 kg/s for infants 60 kg/a = 1.9E-06 kg/s for adults
Q <sub>milk</sub> (g)	[L/s]	Consumption rate of milk by an individual in age group g	120 L/a = 3.8E-06 L/s for infants 105 L/a = 3.3E-06 L/s for adults

Notation	Unit	Definition	Value
Q <sub>soil</sub> (g)	[kg/s]	Inadvertent ingestion rate of ground deposition for an individual in age group g living under normal conditions	100 mg/d = 1.2E-09 kg/s for infants 50 mg/d = 5.8E-10 kg/s for adults
$R_{4\pi,\beta,i}(baseline-inst)$	[cps/Bq]	Ideal response factor of the baseline beta monitoring instrument, which reflects the response [cps] for each nuclear transformation per second [Bq] of radionuclide i on the surface, for $4\pi$ geometry, assuming ideal conditions (e.g. no self-absorption or air absorption) and the instrument window being very close to the surface	Table 32
$R_{4\pi,\beta,i}(inst)$	[cps/Bq]	Response factor of the beta monitoring instrument, which reflects the response [cps] for each nuclear transformation per second [Bq] of radionuclide i on the surface, for $4\pi$ geometry, assuming ideal conditions (e.g. no self-absorption or air absorption) and the instrument window being very close to the surface.	External input
$R_{\beta}$ (baseline-inst)	[counts]	Ratio between (a) the number of counts by the baseline instrument and (b) the number of particles of interest with an energy above 75 keV (and thus assumed to be detected) that reach the outside face of the detector	0.3
ρ <sub>dep</sub>	[kg/m <sup>3</sup> ]	Density of the deposition	$1600 \text{ mg/cm}^3 = 1600 \text{ kg/m}^3$
RA <sub>i</sub> (t,mix)	[unitless]	Relative activity of radionuclide i at time t after reactor shutdown for a specific radionuclide mix	Eq. (1)
RF <sub>i</sub> (mix)	[unitless]	Release fraction of radionuclide i from the fuel for a specific radionuclide mix	Table 7, Table 8 and Table 9
ROIL7(t,mix)	[unitless]	Maximum value of $OIL7_{I-131}(t,mix)$ and $OIL7_{Cs-137}(t,mix)$ relative to their corresponding default OIL value	Fig. 16
R <sub>view-area</sub>	[m]	Radius of a circle with an area of 100 cm <sup>2</sup>	0.0564
SF <sub>e→H*</sub>	[unitless]	Scaling factor to convert the effective dose into ambient dose equivalent	1.4
$SF_{eext(adult \rightarrow infant)}$	[unitless]	Scaling factor to convert the effective dose to the adult to the effective dose to the infant for external exposure	1.4
${\rm SF}_{({\rm hredmarrow}  ightarrow { ightarrow} { hoffetus}), { m air-sh}}$	[unitless]	Scaling factor to convert the dose conversion factor for the equivalent dose rate to the red marrow into equivalent dose rate to the fetus, for air shine	0.9
$SF_{(hredmarrow \rightarrow hfetus),grd-sh}$	[unitless]	Scaling factor to convert the dose conversion factor for the equivalent dose rate to the red marrow into equivalent dose rate to the fetus, for ground shine	0.8
τ	[s]	Time after deposition	Variable
t	[s]	Time after shutdown of the reactor (i.e. after the fuel was last irradiated in a reactor)	Variable
τ <sub>0</sub>	[s]	One day	1 d = 86 400 s

Notation	Unit	Definition	Value
t <sub>o</sub>	[s]	Time after reactor shutdown before which no exposure of the public is expected to occur	30 min = 1800 s
T <sub>1/2,bio-I-131</sub>	[s]	Biological half-life of I-131 in the thyroid	80 d = 6.9E+06 s
T <sub>1/2,w-skin</sub>	[s]	Half-life for removal of surface deposition from the skin due to weathering	14.7 h = 5.2E+04 s
T <sub>1/2,w-vegetation</sub>	[s]	Half-life for removal of surface deposition from vegetation due to weathering	14 d = 1.2E+06 s
t <sub>m</sub>	[s]	Time between intake and measurement of the thyroid	Variable
$TI_{grd \rightarrow air,i}(\Delta)$	[s/m]	Time integrated transfer factor of radionuclide i from the ground to the air (resuspended into the air by wind or other natural processes), integrated over the exposure period $\Delta$	Table 17
$TI_{grd \rightarrow GI,i}(\Delta,g)$	[m <sup>2</sup> ]	Time integrated transfer factor of radionuclide i from the ground to the GI tract by inadvertent ingestion for an individual in age group g, integrated over the exposure period $\Delta$	Table 19
$T_{feed \rightarrow cow-milk,i}$	[s/L]	Transfer factor of radionuclide i from feed to the milk of the cow	Table 21
$T_{grd \rightarrow air}(\tau)$	[m <sup>-1</sup> ]	Transfer factor from the ground to the air (by wind or other natural processes) at time $\tau$ after deposition, also called resuspension factor	Fig. 6
$T_{grd \rightarrow air}(\tau_0)$	[m <sup>-1</sup> ]	Transfer factor from the ground to the air (by wind or other natural processes), at time $\tau_0$	10E-05
$T_{grd \rightarrow GI}(\tau,g)$	[m <sup>2</sup> /s]	Transfer factor from the ground to the GI tract by inadvertent ingestion, for an individual in age group g (i.e. the infant for the representative person and the adult for the fetus), at time $\tau$ after deposition	Fig. 7
$T \to \sigma(\tau, \sigma)$	[m <sup>2</sup> /s]	Transfer factor from the ground to the GI tract by inadvertent ingestion, for an individual in age group g (i.e. the infant for the representative person and the adult for the fetus), at time $\tau_0$	3.6E-10 m <sup>2</sup> /s for adults
			7.2E-10 m <sup>2</sup> /s for infant
- ()	r 2/ r	Transfer factor from the skin to the GI tract due to inadvertent ingestion for an individual in age group g (i.e. the infant for the representative person and the adult for the fetus)	6.4E-08 m <sup>2</sup> /s for infants
$T_{skin \rightarrow GI}(g)$	[m <sup>-</sup> /s]		3.2E-08 m <sup>2</sup> /s for adults
UC	as applicable	Unit conversion factor	as applicable
U <sub>cow</sub>	[kg/s]	Cow consumption rate of feed in dry weight	16 kg/d = 1.9E-04 kg/s
$WI_{G,i}(\Delta)$	[s]	Time integrated external ground dose rate weathering factor, for radionuclide i, integrated over the exposure period $\Delta$	Table 17
WF <sub>OIL1γ</sub>	[unitless]	Weighting factor used to avoid the implementation of unwarranted response actions based on the default $OIL1_{\gamma}$ value	3

Notation	Unit	Definition	Value
WF <sub>OIL2γ</sub>	[unitless]	Weighting factor used to avoid the implementation of unwarranted response actions based on the default $\text{OIL2}_\gamma$ value	1
WF <sub>OIL3γ</sub>	[unitless]	Weighting factor used to avoid the implementation of unwarranted response actions based on the default $\text{OIL3}_\gamma$ value	5
$WF_{OIL4\gamma}$	[unitless]	Weighting factor used to avoid the implementation of unwarranted response actions based on the default $OIL4_{\gamma}$ value	0.5
$WF_{OIL4\beta}$	[unitless]	Weighting factor used to avoid the implementation of unwarranted response actions based on the default $OIL4_\beta$ value	0.5
WF <sub>OIL7</sub>	[unitless]	Weighting factor used to avoid the implementation of unwarranted response actions based on the default OIL7 value	5
WF <sub>OIL8γ</sub>	[unitless]	Weighting factor used to avoid the implementation of unwarranted response actions based on the default $\text{OIL8}_\gamma$ value	1
$W_G(\tau)$	[unitless]	External ground dose rate weathering factor at time $\tau$ after deposition	Fig. 5
$Y_{\beta,E,i}$	[Bq <sup>-1</sup> s <sup>-1</sup> ] [unitless]	Number of particles of interest emitted with energies equal to or above 75 keV (i.e. detectable by the baseline monitor) per nuclear transformation of radionuclide i (yield); the particles of interest include beta +, beta – and internal conversion electrons.	Table 32
$Y_{E_{j,i}}^{\gamma}$	[Bq <sup>-1</sup> s <sup>-1</sup> ] [unitless]	Absolute yield of discrete photons of energy $E_j$ emitted in nuclear transformation of the radionuclide i	'Spectra' spreadsheet on the attached CD.

# ABBREVIATIONS

BWR	boiling water reactor
Е	effective dose
EAL	emergency action level
EPD	extended planning distance
EPR	emergency preparedness and response
GC	generic criteria
GI	gastrointestinal
HAZMAT	hazardous materials
ICPD	ingestion and commodities planning distance
ICRP	International Commission on Radiological Protection
ITB	iodine thyroid blocking
LWR	light water reactor
MW(e)	megawatt electric
MW(th)	megawatt thermal
NPP	nuclear power plant
OIL	operational intervention level
PAZ	precautionary action zone
PWR	pressurized water reactor
RBE	relative biological effectiveness
WHO	World Health Organization

# CONTRIBUTORS TO DRAFTING AND REVIEW

Bučar, T.	Jožef Stefan Institute, Slovenia
Dodd, B.	BDConsulting, United States of America
Hoffman, F. O.	Oak Ridge Center for Risk Analysis, United States of America
Kouts, K.	Scientific Practical Centre of Hygiene, Belarus
Kutkov, V.	National Research Centre Kurchatov Institute, Russian Federation
Mc Kenna, T.	International Atomic Energy Agency
Nadeem, M.	Pakistan Nuclear Regulatory Authority, Pakistan
Nestoroska Madjunarova, S.	International Atomic Energy Agency
Pucelj, B.	Jožef Stefan Institute, Slovenia
Sadañiowski, I.	Nuclear Regulatory Authority of Argentina, Argentina
Stepišnik, M.	Jožef Stefan Institute, Slovenia
Vilar Welter, P.	International Atomic Energy Agency
Winters, G.	International Atomic Energy Agency



# **ORDERING LOCALLY**

In the following countries, IAEA priced publications may be purchased from the sources listed below or from major local booksellers.

Orders for unpriced publications should be made directly to the IAEA. The contact details are given at the end of this list.

# BELGIUM

Jean de Lannoy Avenue du Roi 202, 1190 Brussels, BELGIUM Telephone: +32 2 5384 308 • Fax: +32 2 5380 841 Email: jean.de.lannoy@euronet.be • Web site: http://www.jean-de-lannoy.be

## CANADA

Renouf Publishing Co. Ltd. 22-1010 Polytek Street, Ottawa, ON K1J 9J1, CANADA Telephone: +1 613 745 2665 • Fax: +1 643 745 7660 Email: order@renoufbooks.com • Web site: http://www.renoufbooks.com

#### Bernan Associates

4501 Forbes Blvd., Suite 200, Lanham, MD 20706-4391, USA Telephone: +1 800 865 3457 • Fax: +1 800 865 3450 Email: orders@bernan.com • Web site: http://www.bernan.com

#### **CZECH REPUBLIC**

Suweco CZ, s.r.o. SESTUPNÁ 153/11, 162 00 Prague 6, CZECH REPUBLIC Telephone: +420 242 459 205 • Fax: +420 284 821 646 Email: nakup@suweco.cz • Web site: http://www.suweco.cz

# FRANCE

Form-Edit

5 rue Janssen, PO Box 25, 75921 Paris CEDEX, FRANCE Telephone: +33 1 42 01 49 49 • Fax: +33 1 42 01 90 90 Email: fabien.boucard@formedit.fr • Web site: http://www.formedit.fr

#### Lavoisier SAS

14 rue de Provigny, 94236 Cachan CEDEX, FRANCE Telephone: +33 1 47 40 67 00 • Fax: +33 1 47 40 67 02 Email: livres@lavoisier.fr • Web site: http://www.lavoisier.fr

#### L'Appel du livre

99 rue de Charonne, 75011 Paris, FRANCE Telephone: +33 1 43 07 43 43 • Fax: +33 1 43 07 50 80 Email: livres@appeldulivre.fr • Web site: http://www.appeldulivre.fr

#### GERMANY

Goethe Buchhandlung Teubig GmbH Schweitzer Fachinformationen Willstätterstrasse 15, 40549 Düsseldorf, GERMANY

Telephone: +49 (0) 211 49 874 015 • Fax: +49 (0) 211 49 874 28 Email: kundenbetreuung.goethe@schweitzer-online.de • Web site: http://www.goethebuch.de

#### HUNGARY

*Librotrade Ltd., Book Import* Pesti ut 237. 1173 Budapest, HUNGARY Telephone: +36 1 254-0-269 • Fax: +36 1 254-0-274 Email: books@librotrade.hu • Web site: http://www.librotrade.hu

## INDIA

Allied Publishers 1<sup>st</sup> Floor, Dubash House, 15, J.N. Heredi Marg, Ballard Estate, Mumbai 400001, INDIA Telephone: +91 22 4212 6930/31/69 • Fax: +91 22 2261 7928 Email: alliedpl@vsnl.com • Web site: http://www.alliedpublishers.com

# Bookwell

3/79 Nirankari, Delhi 110009, INDIA Telephone: +91 11 2760 1283/4536 Email: bkwell@nde.vsnl.net.in • Web site: http://www.bookwellindia.com

#### ITALY

## Libreria Scientifica "AEIOU"

Via Vincenzo Maria Coronelli 6, 20146 Milan, ITALY Telephone: +39 02 48 95 45 52 • Fax: +39 02 48 95 45 48 Email: info@libreriaaeiou.eu • Web site: http://www.libreriaaeiou.eu

#### JAPAN

*Maruzen-Yushodo Co., Ltd.* 10-10, Yotsuyasakamachi, Shinjuku-ku, Tokyo 160-0002, JAPAN Telephone: +81 3 4335 9312 • Fax: +81 3 4335 9364 Email: bookimport@maruzen.co.jp • Web site: http://maruzen.co.jp

# **RUSSIAN FEDERATION**

#### Scientific and Engineering Centre for Nuclear and Radiation Safety 107140, Moscow, Malaya Krasnoselskaya st. 2/8, bld. 5, RUSSIAN FEDERATION Telephone: +7 499 264 00 03 • Fax: +7 499 264 28 59 Email: secnrs@secnrs.ru • Web site: http://www.secnrs.ru

## UNITED STATES OF AMERICA

#### Bernan Associates

4501 Forbes Blvd., Suite 200, Lanham, MD 20706-4391, USA Telephone: +1 800 865 3457 • Fax: +1 800 865 3450 Email: orders@bernan.com • Web site: http://www.bernan.com

#### Renouf Publishing Co. Ltd.

812 Proctor Avenue, Ogdensburg, NY 13669-2205, USA Telephone: +1 888 551 7470 • Fax: +1 888 551 7471 Email: orders@renoufbooks.com • Web site: http://www.renoufbooks.com

Orders for both priced and unpriced publications may be addressed directly to:

IAEA Publishing Section, Marketing and Sales Unit International Atomic Energy Agency Vienna International Centre, PO Box 100, 1400 Vienna, Austria Telephone: +43 1 2600 22529 or 22530 • Fax: +43 1 2600 29302 Email: sales.publications@iaea.org • Web site: http://www.iaea.org/books

EPR-NPP-OILS 2017

INTERNATIONAL ATOMIC ENERGY AGENCY VIENNA 2518-685X